

https://theses.gla.ac.uk/

Theses Digitisation:

https://www.gla.ac.uk/myglasgow/research/enlighten/theses/digitisation/

This is a digitised version of the original print thesis.

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk



135

The effect of iron and ironbinding proteins on murine and human lymphocytes

by

Abdelhakim Djeha

being a thesis submitted for the degree of Doctor of Philosophy in the Faculty of Science

> Department of Bacteriology and Immunology Western Infirmary

> > SCOTLADD 1990

© 1990 by Abdelhakim Djeha



ł

and the

「「「「「「「「」」」の「「」」」「「」」」の「「」」」の「「」」」」の「「」」」の「」」」の「」」」」の「」」」」の「」」」の「」」」の「」」の「」」の「」」の「」」の「」」の「」」の「」」の「」」の



ACKNOWLEDGMENTS

It is with immense pleasure that I take this opportunity to thank my supervisor Dr Jeremy H. Brock for showing patience and kindness towards me during the course of this thesis, and also for his constant encouragement, advice and invaluable guidance without which this work and the development of my own knowledge as a scientist would not have been possible.

My sincere thanks are also extended to Professor Delphine M.V. Parrott who in the first instance gave me the opportunity to complete my study in the Department of Bacteriology and Immunology, and who's near departure will be felt with great regret.

I am also indebted to all my colleagues I worked with in Dr Brock's laboratory during the time spent in this thesis for their kindness, assistance, patience and understanding of whom I name Dr Sheila McGregor, Mrs Joan Licéaga, Mr Edson Rocha and Dr Rosa Oria. Special mention should made to Dr Xavier Alvarez-Hernàndez who has instructed me in invaluable ways and to whom I am extremely grateful for his interest in the work and for his practical help. I would like also to express my deep gratitude to Dr. Charles McSharry for the way that he kindly gave me constant personal encouragement and words of support.

My gratitude is also due to many people who have helped to make this thesis possible; staff of the Biochemistry Department, Gartnaval General Hospital for performing the human ferritin assay, the nursing staff of Yorkhill Hospital who provided me with colostrum samples, and Mr Frazer Darling and his staff for care of the animals. The work presented in this thesis has only been made possible by the financial support from the Algerian government which is sincerely acknowledged.

Finally this thesis is dedicated to my parents, brothers and sisters in Algeria.

TABLE OF CONTENTS

TITLE PAGE	i
ACKNOWLEDGMENTS	ii
TABLE OF CONTENTS	iv
LIST OF TABLES	x
LIST OF FIGURES	xi
ABBREVIATIONS	xiv
SUMMARY	xvii

CHAPTER ONE

æ

PART ONE: Literature Review of Iron Metabolism
1.1.1 Role of iron in living organisms1
1.1.2 The transferrins: general properties,
synthesis and function4
1.1.2.1 Transferrin 5
1.12.2 Lactoferrin 10
1.1.3 Structure and iron binding properties
of the transferrins13
1.1.4 Iron storage proteins
1.1.4.1 Ferritin 19
1.1.4.2 Haemosiderin 24
1.1.5 Cellular iron uptake:
the transferrin cycle
1.1.5.1 Structure and regulation of
the transferrin receptor
1.1.5.2 Cellular iron uptake
from transferrin
1.1.6 Metabolism and distribution
of intracellular iron
FARM NWO: Literature Review of fron and Lymphocyte
Function
1.2.1 The lymphocyte
1.2.2 T-lymphocyte interaction with
antigen or mitogen: activation proliferation

1.2.3	The role of transferrin in proliferation	
	and iron metabolism of lymphocytes4	12
1.2.4	Effect of iron deficiency and iron	
	overload on lymphocyte function4	17
AIMS	OF THESIS	50

CHAPTER TWO

The Effect of Different Levels and Forms of Iron on Murine Lymphocytes

2.1 INTRODU	CTION	
2.2 MATERIA	L S	
2.2.1 Anima	uls	
2.2.2 Reage	nts	
2.2.2.1	Phosphate .	Buffered
	Saline (1	PBS) 56
2.2.2.2	Human Ser	um Albumin
	solution	(HSA) 56
2.2.2.3	Concanava	lin A solution (Con A) 57
2.2.2.4	Ferric nite	rilotriacetate(FeNTA) 57
2.2.2.5	Ferric pyria	doxal isonicotinoyl
	hydrazone	(<i>FePIH</i>) 57
2.2.2.6	Standard	culture medium57
2.2.2.7	Reagents fo	r anaphthyl
	acetate es	terase staining
	2.2.2.7.1	<i>Fixative</i>
	2.2.2.7.2	Phosphate Buffer 58
	2.2.2.7.3	Hexazotized pararosaniline 58
	2.2.2.7.4	α-naphthyl acetate
		esterase stain 59
2.2.2.8	Reagents fo	r Poly Acrylamide Gel
	Electropho	resis for ferritin 59
	2.2.2.8.1	Acrylamide solution 59
	2.2.2.8.2	Sample buffer 59
	2.2.2.8.3	<i>Tank buffer</i> 59
	2.2.2.8.4	<i>Fixer</i>
	2.2.2.8.5	Protein stain 60

			2.2.2.8.6	Perl's Prussian	
				blue reagent	60
			2.2.2.8.7	Destaining solution	60
	2	2.2.2.9	Reagent for	immunoprecipitation	
			of transfe	errin	60
			2.2.2.9.1	3D Buffer	60
	2	2.2.2.10	Reagents fo	or SDS-PAGE for	
			transferri	n	.61
			2.2.2.10.1	Urea buffer	61
			2.2.2.10.2	Sample buffer	61
			2.2.2.10.3	Tank buffer	61
2.3	МЕТ	HOD	S		.62
	2.3.1	Prepara	tion of cell s	uspension	
		and cu	ulture con	ditions	62
	2.3.2	Measur	ement of	proliferation	62
	2.3.3	Uptake	of iron by p	roliferating mouse	
		lympho	cytes culture	d with different	
		transfe	rrin satur	ations	63
	2.3.4	Isolatio	on of fer	ritin	64
		2.3.4.1	Agarose po	olyacrylamide gel	
			electroph	presis for ferritin	65
	2.3.5	Prepara	tion of anti	bodies against ferritin	66
	2.3.6	Affinit	y chromat	ography	67
	2.3.7	Intracel	lular distribu	ation of iron in	
		prolifera	ating mouse	lymphocytes	
		culture	d with diff	ferent carriers	68
	2.3.8	α-napht	hyl acetate	esterase	
		stainin	g (ANAE)	70
	2.3.9	Immuno	oprecipitation	nmethod	
		for tr	ansferrin		72
		2.3.9.1	SDS Polya	crylamide Gel	
			Electropho	presis for	
			transferi	-in	75
	2.3.10	Labelli	ng of rabbit	antiferritin with ¹²³ I	77
	2.3.11	Immun	oradiometri	c assay for ferritin (IRMA)	. 79
		2.3.11	.1 Titration	n of first and	
			second	antibodies	79

vi

	2.3.12	Expression of results and statistical analysis	. 80
2.4	RES	ULTS	81
	2.4.1	Titration of concanavalin A	
		concentrations	. 81
	2.4.2	Uptake of iron and thymidine by	
		proliferating mouse lymphocytes	
		cultured with transferrin at	
		different iron saturations	. 83
	2.4.3	Proliferation of lymphocytes cultured	
		with different iron chelates	.88
	2.4.4	Iron uptake and its intracellular	
		distribution in proliferating	
		mouse lymphocytes cultured	
		with different iron carriers	91
	2.4.5	Detection of transferrin synthesis by	
		stimulated mouse lymph node cells	. 96
	2.4.6	Intracellular ferritin levels in	
		proliferating mouse lymphocytes	
		cultured with different iron carriers	102
2.5	DISC	CUSSION	108
	2.5.1	Uptake of iron and thymidine by	
		proliferating mouse lymphocytes	
		in the presence of different	
		transferrin saturations	108
	2.5.2	The effect of different iron	
		chelates on mouse lymphocyte	
		proliferation	113
	2.5.3	Uptake and intracellular handling	
		of iron from different carriers by	
		mouse lymphocytes	116
	2.5.4	Transferrin synthesis by mouse	
		lymph node cells	131

CHAPTER THREE

,

The Effect of Different Levels and Forms of Iron on Human Lymphocytes

3.1	INTR	ODUCTION	5
3.2	MET	HODS 140)
	3.2.1	Cell culture conditions of human	
	1	mononuclear cells140)
	3.2.2	Indirect immunofluorescence140)
	3.2.3	Preparation of lactoferrin142	2
	3.2.4	Cell line144	4
	3.2.5	Detection of transferrin synthesis	
	1	by CCRF-CEM cell line144	1
2 2	RESI	14e	5
5.5	331	Titration of phytohaemagglutinin 146	5
	332	The effect of iron saturation of	,
	J.J.H	transferrin on human lymphocyte	
		proliferation	5
	3.3.3	The effect of FeNTA and FePIH on	-
		human lymphocyte proliferation)
	3.3.4	The effect of iron saturation of	
		transferrin on intracellular ferritin	
		of proliferating human lymphocytes	9
	3.3.5	The effect of lactoferrin on human	
		lymphocyte proliferation	1
	3.3.6	The effect of lactoferrin on intracellular	
		ferritin levels of proliferating	
		human lymphocytes	3
	3.3.7	The effect of iron saturation of	
		transferrin on T-cell subsets150	5
	3.3.8	The effect of FeNTA and	
		FePIH on T-cell Subsets150	5
	3.3.9	Proliferation of CEM cells cultured	
		in the presence of different iron	
		saturations of transferrin	8
	3.3.10	Proliferation of CEM cells in the	
		presence of FeNTA and FePIH162	1

3.3.11 The effect of transferrin saturation
on intracellular ferritin levels
of CCRF-CEM cells161
3.3.12 Proliferation of CEM cells in the
presence of lactoferrin
3.3.13 Detection of transferrin synthesis
by CEM cells166
3.4 DISCUSSION
3.4.1 The effect of iron saturation of
transferrin, FeNTA and FePIH
on proliferation of human
lymphocytes168
3.4.2 The effect of iron saturation of
transferrin on intracellular
ferritin levels of proliferating
human lymphocytes170
3.4.3 The effect of lactoferrin on
proliferation and ferritin levels
of human lymphocytes173
3.4.4 The effect of iron saturation of transferrin,
FeNTA and FePIH on the expression
of T-cell surface markers178
3.4.5 Proliferation of CCRF-CEM cells
in the presence of transferrin,
FeNTA, FEPIH and lactoferrin
GENERAL DISCUSSION
REFERENCES

LIST OF TABLES

Table 1	Binding of ⁵⁹ Fe-mouse transferrin to anti-
	human transferrin affinity column
Table 2	Intracellular ferritin levels in Con A- stimulated mouse lymphocytes cultured in the presence of different iron carriers 107
Table 3	The effect of transferrin saturation on intracellular ferritin in proliferating human
	lymphocytes 152
Table 4	The effect of lactoferrin on intracellular
	ferritin in proliferating human lymphocytes 155
Table 5	The effect of transferrin saturation on expression of CD3 CD4 and CD8 by
	proliferating human lymphocytes 157
Table 6	The effect of Fe-transferrin, FeNTA and
	FePIH on expression of CD3, CD4 and
	CD8 by proliferating human lymphocytes 159
Table 7	The effect of transferrin saturation on
	intracellular ferritin in CCRF-CEM cells

LIST OF FIGURES

.

Figure 1	Lymphocyte triggering transmembrane
	signals40
Figure 2	Diagram of an affinity column used in
	fractionating lysate supernatants for
	intracellular distribution of iron71
Figure 3	Titration of Concanavalin A concentrations
	in stimulating mouse lymphocytes
Figure 4	Iron uptake by Con A-stimulated mouse
	lymphocytes from transferrin of different
	iron saturations84
Figure 5	Iron and thymidine uptake by Con A-
	stimulated mouse lymphocytes cultured with
	different saturations of transferrin with iron85
Figure 6	Thymidine uptake by Con A-stimulated
	mouse lymphocytes cultured with different
	saturations of transferrin with iron
Figure 7	The effect of FeNTA on proliferation of
	Con A-stimulated mouse lymphocytes
Figure 8	The effect of FePIH on proliferation of Con
	A-stimulated mouse lymphocytes
Figure 9	Iron uptake by Con A-stimulated mouse
	lymphocytes cultured with different iron
	carriers
Figure 10	Intracellular distribution of iron in Con A-
	stimulated mouse lymphocytes cultured with
	different iron carriers at 10 ng/ml iron

Figure 11	Intracellular distribution of iron in Con A- stimulated mouse lymphocytes cultured with different iron carriers at 50 ng/ml iron
Figure 12	Autoradiography of ^{3 5} S-cysteine incorporated into transferrin by unstimulated, Con A-stimulated and <i>in</i> <i>vivo</i> -stimulated mouse lymph node cells
Figure 13	In vivo-stimulated lymph node cells stained with alpha-naphthyl acetate esterase stain 100
Figure 14	Autoradiography of ^{3 5} S-cysteine incorporated into transferrin by adherent and non-adherent <i>in vivo</i> -stimulated mouse lymph node cells101
Figure 15	Autoradiography of ^{3 5} S-cysteine incorporated into transferrin by B- and T- lymphocyte depleted and non-depleted <i>in</i> <i>vivo</i> -stimulated mouse lymph node cells
Figure 16	Immunoradiometric assay standard curves with spleen ferritin-α-spleen Ft
Figure 17	Titration of the first antibody (α-mouse liver ferritin) with mouse liver ferritin
Figure 18	Optical densities of lactoferrin fractions at 470 and 280 nm143
Figure 19	Titration of response to phytohaemagglutinin of human lymphocytes 147
Figure 20	The effect of the degree of transferrin saturation with iron on proliferation of PHA-stimulated human lymphocytes
Figure 21	The effect of different iron carriers on proliferation of PHA-stimulated human lymphocytes

xii

Figure	22	The effect of lactoferrin on proliferation of	
		PHA-stimulated human lymphocytes	
		cultured with or without transferrin	154
Figure	23	The effect of transferrin saturation on	
		CCRF-CEM cell growth	160
Figure	24	The effect of different iron carriers on	
		CCRF-CEM cell growth	162
Figure	25	The effect of lactoferrin on growth of	
		CCRF-CEM cells cultured with or without	
		transferrin	165
Figure	26	Autoradiography of ^{3 5} S-cysteine	
		incorporated into transferrin by CCRF-CEM	
		cells	167

;

,

ABBREVIATIONS

α-	anti-
ANAE	alpha-naphtyl acetate esterase
APC	antigen presenting cells
apoFt	apoferritin (free of iron)
apoLf	apolactoferrin (free of iron)
apoTf	apotransferrin (free of iron)
Arg	arginine
Asp	asparagine
ATP	adenosine triphosphate
BD	below detection
BSA	bovine serum albumin
C-	carboxy-
[Ca ²⁺]i	intracellular calcium concentration
CD	cluster of differentiation
cDNA	complimentary deoxyribonucleic acid
Ci	Curie
Con A	concanavalin A
cpm	counts per minute
CSA	colony-stimulating activity
CTL	cytotoxic T lymphocyte
Cys	cysteine
d	dalton
DFO	desferrioxamine
DG	diacyl glycerol
DNA	deoxyribonucleic acid
Fab	fragment obtained by papain hydrolysis of
	immunoglobulins (a pair of heavy and light chains)
FCS	fetal calf serum
FeNTA	ferric nitrilotriacetate
FePIH	ferric pyridoxal isonicotinoyl hydrazone
FeTf	ferric transferrin
FITC	fluorescein isothiocyanate
FSH	follicle stimulating hormone
Ft	ferritin
GM-CSF	granulocyte/monocyte-colony stimulating factor

GTP	guanosine triphosphate
His	histidine
Hs	haemosiderin
HSA	human serum albumin
HTf	human transferrin
IEF	isoelectric focusing
IFN	interferon
Ig	immunoglobulin
IgG	immunoglobulin G
IL	interleukin
IL-2R	interleukin-2 receptor
IP	inositol phosphate
IRE	iron regulatory element .
IRMA	immunoradiometric assay
IU	international unit
Kd	kilo dalton
Lf	lactoferrin
LfR	lactoferrin receptor
mA	milli-Ampere
Me	mercaptoethanol
MHC	major histocompatibility complex
MLC	mixed lymphocyte culture
MLFt	mouse liver ferritin
MSFt	mouse spleen ferritin
MTf	mouse transferrin
mRNA	messenger ribonucleic acid
N-	amino-
NAD	nicotinamide adenine dinucleotide
NADH	nicotinamide adenine dinucleotide (reduced form)
NK	natural killer
NRIg	normal rabbit immunoglobulins
NSIg	normal sheep immunoglobulins
NTA	nitrilotriacetate
Р	probability of significance
p97	melano-transferrin
PBS	phosphate buffered saline
PDB	phorbol dibutyrate

xv

-

.

4

PG	prostaglandin
PHA	phytohaemagglutinin
pI	isoelectric point
PIH	pyridoxal isonicotinoyl hydrazone
PIP-PDE	polyphosphoinositide phosphodiesterase
РКС	protein kinase C
PMA	phorbol myristate acetate
PMSF	phenylmethyl sulphonyl fluoride
Ra-MLFt	rabbit antimouse liver ferritin
Ra-MSFt	rabbit antimouse spleen ferritin
rpm	revolutions per minute
RPMI	Roswell Park Memorial institute (medium number
	1640)
S-	Sepharose 4B coupled to-
Sa-HTf	sheep antihuman transferrin
sat.	saturation
SD	standard deviation from the mean
SDS	sodium dodecyl sulphate
T X-100	Triton X 100
TCA	trichloroacetic acid
TEMED	N,N,N',N'-tetramethylethylenediamine
Tf	transferrin
TfR	transferrin receptor
Ti	antigen receptor
Tyr	tyrosine
UTR	untranslated region
v/v	volume per volume
w/o	without
w/v	weight per volume
Å	Ångstrom

•

SUMMARY

It is well documented that iron deficiency affects the function of lymphocytes because this essential element is required by these cells. However, the way that lymphocytes respond to high levels of iron is not fully understood. Reports of an increased incidence of infection and neoplasia among iron-overloaded patients might indicate that excess iron causes depression of specific immune responses in these patients. The broad objective of these study was therefore to investigate how lymphocytes react to different degrees of extracellular iron availability.

Iron uptake by proliferating mouse lymphocytes from transferrin (Tf) saturated to different degrees with iron showed a gradual increase at saturations below complete saturation of the protein. The uptake rose sharply when non-Tf bound iron was present in the medium and the ratio of iron uptake to iron available increased. The proliferative capacity of these cells assayed alongside iron uptake was low at low Tf saturations and an increased rate of transformation was associated with increased percentage of saturation of Tf with iron. When saturation exceeded the binding capacity of the protein, proliferation decreased and at high levels of iron it was reduced below control level. Ferric nitrilotriacetate (FeNTA) was found to donate very large amounts of iron to cells compared to Tf but did not promote proliferation and when present in high amounts caused inhibition. In contrast ferric pyridoxal isonicotinoyl hydrazone (FePIH) which donated iron to cells at a slightly higher rate than Tf was found to support proliferation as efficiently as Tf.

A study has been made of intracellular events in the iron

metabolism of proliferating mouse lymphocytes to clarify the relationship between iron uptake and intracellular iron metabolism in these cells cultured with different iron carriers and to relate this to their ability to promote proliferation. This involved iron chelation, immunoprecipitation and ultrafiltration. In cells cultured with FeNTA, iron was found predominantly in an insoluble non-Ferritin (Ft) macromolecular form, while in the cells cultured with FeTf or FePIH the largest proportion of iron was found in the intermediate molecular weight fraction, which probably represents iron being used to form enzymes. The cells showed no marked increase in synthesis of Ft irrespective of the form of iron Tf. Synthesis of Tf was found in *in vivo*-stimulated lymphocytes, and macrophages were found to be the most active cells in synthesising this protein.

Comparable studies of the effect of different iron carriers on cellular proliferation were performed with human lymphocytes and a related cell line, the T-lymphoblastoid CCRF-CEM line. Generally human lymphocytes gave similar results to mouse cells except with FePIH which was found to be less effective than with mouse cells. In addition, the presence of non-Tf bound iron in the form of FeNTA, but not FePIH, caused a decrease in CD4/CD8 ratio, due mainly to depression of the proportion of CD4⁺ cells. However unlike normal cells CCRF-CEM cells did not show any difference in their proliferative activity at different saturations of Tf and were able to achieve good proliferation when high levels of non-Tf bound iron in the form of FeNTA or FePIH was present in the culture medium. These cells were also found to have the ability to make their own Tf.

Unlike Tf, Lf, the other member of the Tf family did not have any

effect on proliferation of stimulated human lymphocytes whether added in the (apo) or loaded from. However, in the presence of excess non-Tf bound iron, apoLf increased the ability of these cells to proliferate.

Therefore the main conclusion of this study is that the presence of concentrations of iron above the level that saturates all transferrin present in the medium inhibits proliferation of lymphocytes, and preferentially affects the helper subsets. It seems likely that some mechanism for uptake of unbound iron into these cells may exist resulting in toxic consequences.



PART ONE

J

LITERATURE REVIEW OF IRON METABOLISM

1.1.1 Role of iron in living organisms

Iron, is the second most common metal in the earth's crust and an invariable essential constituent of all living organisms with possibly the exception of the lactobacilli (Archibald, 1983).

Like other transition metals, iron is an important biological catalyst. Because it possesses unfilled d atomic orbitals, iron has oxidation states varying from -2 to +6, but normally it exists in the equilibrium between the +2 and +3 oxidation states, an equilibrium which is very sensitive to both ligation and pH (Hider, 1984). These changes in oxidation states involving one electron allow iron to coordinate electron donors and play a crucial role in many biological processes requiring oxidation and reduction.

The thermodynamically stable state of iron is Fe^{3+} , but at concentrations greater than 10^{-18} M, which is the equilibrium concentration of the ion at neutral pH, uncomplexed Fe^{3+} ions are very readily hydrolysed and polymerised to form essentially insoluble rust-like, ferric hydroxides and oxyhydroxide polymers (May and Williams, 1980).

The evolution of organisms shows a long dependence on iron, and nature has developed rather sophisticated iron chelating and transport systems to utilize the mineral as well as storage systems, though many organisms are auxotrophic for Fe^{3+} . Microorganisms tend to utilize small non-proteinaceous molecules to ensure an adequate supply of the metal. Bacteria have developed a very efficient apparatus for gaining access to this element, involving a group of low molecular weight high affinity chelating agents named siderophores for sequestering and transporting iron from the surrounding environment (Neilands, 1981). These are virtually ferric ion-specific ligands, and usually catechols or hydroxamates. Plants also synthesise and secrete substances that augment iron absorption from the external environment (Finch and Huebers, 1982).

the iron binding functions are met by more In mammals. complicated molecules that form reversible complexes with the metal. These include a group of very similar iron-transport proteins called transferrins (Tfs) and two other proteins, ferritin (Ft) and its lysosomal degradation product haemosiderin (Hs), which fulfill the role of maintaining iron in an available form, but which are essentially iron storage proteins. Beside the group of iron transport and storage proteins, there are two other groups of iron-containing proteins i.e. those which complex oxygen and transport it throughout the body, and iron There are data showing the existence of containing enzymes. siderophore-like low molecular weight iron binding ligands with high binding specificity for Fe³⁺ at neutral pH in mammalian cells, which are preferentially produced under conditions of iron deprivation such as iron deficiency, anaemia, or in pregnant women (Apte and Brown, 1969). Fernandez-Pol (1978) has identified and partially purified a highly specific iron-binding ligand from a mutant of SV-40 transformed mouse cells adapted to grow in picolinic acid, which consists of a peptide of approximately 1.6 Kd.

In all biological systems, iron is always found bound to

complexing agents. This is not only to permit the use of the element in a wide variety of metabolic processes, but also to prevent its potentially deleterious effect which is expressed in the form of free radical formation. It is well known that iron enhances the toxicity of oxygen free radicals (McCord and Day, 1978; Halliwell, 1978). These free radicals are highly reactive and could be very harmful to the host.

There is no doubt that almost all living cells, prokaryotes, eukaryotes, aerobic, anaerobic, photosynthetic, and nitrogen-fixing, cannot survive without iron. Iron plays a crucial role in modulating oxygen supply and transport. Most of the proteins responsible for oxygen binding (haemoglobin and myoglobin) use iron to enable them to bind oxygen reversibly, the few exceptions being found in certain molluscs and arthropods, which use copper in haemocyanin as an oxygen carrier.

Iron enables other haem-containing proteins to participate in several essentially biochemical reactions with molecular oxygen, such as transfer of electrons in cellular respiration, the reduction of peroxides, oxygenation and hydrogenation of organic substances, as well as the transfer of electrons between dehydrogenases and various electron acceptors (Griffiths, 1987).

Iron is also present in a number of enzymes known as iron-sulphur proteins, and through them participates in a number of redox reactions involved in the biosynthesis of steroid hormones, oxidative phosphorylation reactions, and in non-redox reactions involved in the Krebs cycle and purine synthesis (Griffiths, 1987). The process of DNA replication depends on iron through a nonhaem iron-containing enzyme, ribonucleotide reductase which catalyses the conversion of ribonucleotides to deoxyribonucleotides, an essential step in DNA synthesis (Reichard and Ehrenberg, 1983). Iron seems not to participate as an electron carrier in the process of DNA synthesis, but helps to generate and stabilize the organic radical intermediate during catalysis (Atkin *et al.*, 1973), which is necessary for the activity of the enzyme.

The total body iron content of a man of 70 Kg weight is about 4.2 g. Haemoglobin normally accounts for 74.3%, myoglobin 3.3%, haptoglobin and haemoglobin 0.2%, Tf iron accounts for only 0.07%, catalase 0.11%, cytochrome C 0.08%, and the remaining 16.4% is distributed between the proteins of storage Ft and Hs (Beinert, 1973). This distribution of the body iron rapidly changes in pathological disturbances such as iron deficiency and iron overload.

1.1.2 The transferrins: general properties, synthesis, and function

The transferrin protein family consists of a group of evolutionarily and therefore structurally-related metalloproteins, which are found only in the phylum of chordata. These are transferrin (Tf), lactoferrin (Lf), ovotransferrin (ovo-Tf), and another one recently discovered; melanotransferrin (melano-Tf). They are present and widely distributed in numerous vertebrate biological fluids (Feeney and Komatsu, 1966; Aisen and Listowsky, 1980). They are monomeric glycoproteins which can reversibly bind two atoms of Fe^{3+} per molecule of protein, so they may exist in the apo-, mono-, and diferric forms.

Plasma transferrin carries iron and transports it from the intestinal mucosa (site of absorption), the reticuloendothelial system (site of erythrocyte catabolism), and the liver parenchymal cells (site of storage) to supply most body tissues with iron. The function of the other related protein lactoferrin remains unclear, and there is little evidence of an iron-transport role, but it may, like Tf, contribute to antimicrobial defences. Ovotransferrin was the first of the group to be found in 1889 (cited by Weinberg, 1978) in eggwhite from which its name was derived.

1.1.2.1 Transferrin

The major circulating iron binding protein was isolated and characterised in 1946 by Schade and Caroline. The following year, the protein was named transferrin and its physiological importance in the transport of iron in the circulation and also into the cell was confirmed (Laurell and Ingelman, 1947).

The total amount of Tf in man has been estimated to be about 240 mg/Kg body weight. The normal blood concentration of Tf is about 30 μ M, at an iron saturation of about 30%, with half life of about 8-10 days. The concentration of Tf in the serum was found to increase progressively throughout fetal life from about 0.4 mg/ml at 10-15 weeks gestation to levels greater than that of the mother near term (Morgan, 1964; Baker and Morgan, 1969; Wong and Morgan, 1973). Tf is found throughout the extracellular fluids, being equally divided between the intra- and extravascular compartment. It has also been identified in lymph (von Ehrenstein, 1956), pleural, ascitic, oedema (Bogdanikowa and Grabowski, 1972), and cerebrospinal fluid (Parker *et al.*, 1963). Tf is present in the milk of certain animals such as the rat (Morgan, 1981), rabbit (Suard *et al.*, 1983), and mouse (Lee *et al.*, 1987), cow (Sanchez *et al.*, 1988). It has been reported that in milk of some of these animals Tf predominates, rather than Lf (see below).

Tf is synthesised mainly in the liver (Morgan and Peters, 1971; Morton and Tavill, 1977; McKnight et al., 1980). Tf cDNA from a human liver library has been identified, characterised, and the Tf gene mapped on human chromosome 3 (Yang et al., 1984). In the fetus Tf synthesis may occur in the yolk sac (Yeoh and Morgan, 1974) and the lung (Gitlin and Biasucci, 1969) as well as in the liver. The newly synthesised Tf in the hepatocyte contains an additional leader sequence of some 19-20 amino acids, which is split by proteolysis prior to secretion of the protein (Thibodeau et al., 1978; Jeltsch and Chambon, 1982). It seems that this latter step is crucial in the synthetic process, since inhibition of proteolysis prevents secretion (Schreiber et al., 1979). The final step in Tf synthesis is glycosylation, which takes place in the rough endoplasmic reticulum and the Golgi system while en route to be secreted before it enters circulation. Omission of this step seems to have little effect on the secretion process (Schreiber et al., 1979). The rate of Tf synthesis by human liver was determined to be 10.5% of the plasma pool per day (Wochner et al., 1968; Kernof and Baker, 1980).

Other nonhepatic cell types have the ability to make Tf. Synthesis of the protein in extrahepatic sites probably contributes to the requirements of specific tissues, which are separated from plasma by blood barriers that may prevent adequate Tf from being supplied by the circulation, in inflammation, and in iron overload (Bowman *et al.*, 1988).

Ovary cells (Thorbecke et al., 1973), as well as testis Sertoli cells (Skinner and Griswold, 1980; 1982; Skinner et al., 1984) which are responsible for secretion of most of the fluid components of the seminiferous tubules, synthesise a Tf-like protein in vitro differing from serum Tf only in the glycan composition. However it has been suggested that in vivo, Sertoli cells are not actively engaged in the synthesis of testicular Tf and that upon in vitro culturing Sertoli cells activate the Tf gene (Lee et al., 1986). In chicken, Tf is synthesised by the oviduct and its transcription is induced by oestrogens (Lee et al., 1978). Fibroblasts make Tf (Stecher and Thorbecke, 1967b). as also do macrophages (Stecher and Thorbecke, 1967a; Haurani et al., 1973), and the muscle (Levin et al., 1984). Tf mRNA has been found at low levels in rat placenta, spleen, kidney, muscle, and heart (Aldred et al., 1987). It has also been found in newborn, but not adult intestine (Levin et al., 1984). Tf has been described as a fetal growth factor in embryonic induction (Ekblom et al., 1983). It has also been identified as a major protein synthesised and secreted by the mammary epithelium of certain animals, which is identical to the corresponding plasma homologue except for a lower content of sialic acid (Baker et al., 1968; Lee et al., 1987). It has been reported that in late lactation rat milk contains as much as 4-5 mg/ml Tf. The regulation of its synthesis and secretion is distinct from that of other milk proteins (Lee et al., 1987), and exercised at the mRNA level. The central nervous system has been found to make Tf (Bloch et al., 1985; Dickson et al., 1985). It has

been found not only in oligoendrocytes and glial cells, but also in neurons and endothelial cells (Møllgård *et al.*, 1987), and has been described as acting as a neurotrophic factor (Beach *et al.*, 1983). Nevertheless, the amount of Tf synthesised by extrahepatic sites is far less than that produced by the liver. Idzerda *et al* (1986) measured Tf mRNA levels in various tissues of normal rats and found 6500 molecules/cell in liver, compared with 114 in testis, 83 in brain, 11 in spleen, and 5 in kidney.

As far as lymphocytes are concerned, early studies have shown that human lymph nodes (Prunier et al., 1964) and lymphocytic cell lines (Stecher and Thorbecke, 1967b) make Tf. It has been shown that there is an association of radiolabelled immunoreactive Tf with cell extracts of peripheral mononuclear cells at periods of both pre- and postactivation, although the molecular weight was not determined (Soltys and Brody, 1970). More precisely the release of Tf has been demonstrated to be from OKT8+ subpopulations of T-lymphocytes (Broxmeyer et al., 1983). In contrast to Broxmeyer's findings, another group has demonstrated the transcription and synthesis of Tf by T-lymphocytes and localised Tf mRNA specifically in T4+ helper-inducer human lymphocyte subsets, whereas no Tf mRNA was detected in $T8^+$ or in B-cells (Lum et al., 1985: 1986). In peripheral blood mononuclear cells, approximately 2-5% were seen having silver grains denoting Tf mRNA hybridization by using in situ hybridization, and after T-cells were isolated by rosette formation with sheep red blood cells, 10-20% of the T-cells transcribed Tf mRNA (Bowman and Yang, 1987). Cloned Tfindependent lymphoma cell lines have been isolated and reported also to

produce a Tf-like activity (Morrone et al., 1988).

Childhood, pregnancy, iron deficiency anaemia, oestrogen, and administration of some steroid hormones (Bowman et al., 1988) are factors which increase plasma Tf levels. In the liver, oestrogen stimulates Tf synthesis by increasing mRNA transcription (McKnight et al., 1980). Tf synthesis by the testis appears to be hormone controlindependent (Perez-Infante et al., 1986). However other workers have shown that there is a regulation by various hormones, such as FSH, testosterone, and insulin (Skinner and Griswold, 1982), and that this modulation of the Tf expression is exerted on a transcriptional level (Huggenvik et al., 1987). Iron deficiency is an other factor that increases plasma Tf concentration. An iron deficient diet induces an increase of 2-4-fold in serum Tf as well as in Tf mRNA synthesis in the liver (McKnight et al., 1980). In isolated liver nuclei of rats raised on low iron diet, a 2.4-fold increase in Tf transcription activity was found but without physiologic effect since serum Tf iron binding capacity was unchanged (Idzerda et al., 1986). However, the Tf mRNA content of other Tf synthesising tissues such as brain, testis, spleen, and kidney remained unchanged (Idzerda et al., 1986). It has been reported that iron overload had no detectable effect on Tf gene expression (Tuil et al., 1985). However, recently Lescoat et al (1989) have demonstrated that iron overload decreases Tf secretion and postulated that this is exerted through a translational control of the protein.

Very little is known about the precise mechanism or site of catabolism under normal conditions. A proportion (10%) of breakdown has been found to be hepatic as suggested by isolated perfused liver (Hoffenberg *et al.*, 1970). The catabolic rate of plasma Tf is 6-14%/day of plasma pool (Morgan, 1983) which corresponds closely to determinations of the rate of Tf synthesis.

The main function of Tf is to transport iron from sites of storage, absorption, and erythroid catabolism to specialized iron requiring cells, mainly erythroid precursors for haemoglobin synthesis, and specialized iron storage cells (hepatocytes), which incorporate iron into ferritin. The role of Tf in cellular uptake is described in detail in section 1.1.5.2.

Tf also donates iron to actively proliferating cells from different origins. Activated lymphocytes are an example of these cells. Although the amount of iron taken up by those cells is relatively small, this process plays an important role in cell transformation and division. One of the most crucial roles of iron as far as those cells are concerned is its association with ribonucleotide reductase (Brown *et al.*, 1969). However, others have suggested that the Tf growth factor effect is exerted through acting as an electron acceptor for a transmembrane redox system involving NAD (Crane *et al.*, 1985; Navas *et al.*, 1986). Sun *et al* (1987) have reported that the NADH reductase functions in association with the TfR, where Tf attains its effects through modulation of transmembrane potential differences. The role of Tf in lymphocyte proliferation is described in detail in section 1.2.3.

1.1.2.2 Lactoferrin

Lf, which appears to be restricted to mammals, is another member of the Tf family. It is also called lactotransferrin by analogy to its plasma homologue. Lf was first found in its most abundant source, milk

(Montreuil et al., 1960). Subsequently, Lf has been found in most mammalian secretions, including bronchial, nasal, salivary, lachrymal, genital, and gastro intestinal secretions, as well as in sweat, tears, and pancreatic juice (Figarella and Sarles, 1975; Aisen and Listowsky so it is constantly bathing mucous membranes which are 1980), vulnerable to infection. Lf is also found in neutrophils (Baggiolini et al., 1970), more precisely in the specific (secondary) granules (Cramer et al., 1985; Esaguy et al., 1989). During degranulation of neutrophils, Lf is released not only to the phagocyte vacuole but also to extracellular medium. In human plasma, it is present in minute amounts (Bennett and Mohla, 1976). Whether Lf plays any significant role in the transport of iron is not known. If it does, the pathway is probably from plasma to reticuloendothelial cells (Morgan, 1980). Lf is a major constituent of milk and human colostrum contains high concentrations, up to 15 mg/ml. Lf in human milk is less than 5% saturated with iron (Lonnerdal, 1985). Lf was also found in amniotic fluid (Niemelä et al., 1989). Plasma concentration of Lf is only about 1 μ g/ml (Bennett and Mohla, 1976). Sykes et al. (1982) found that the serum Lf level is higher in pregnancy than in normal adult women. Lf escapes digestion and any detrimental effect of the low pH, as judged by the large amount of the so-called coprolactoferrin daily excretion in faeces (5-35 mg) (Spik and Montreuil, 1966; Spik et al., 1982).

Like Tf, genes coding for human Lf are located on the long arm of chromosome 3 (Teng *et al.*, 1987), while mouse Lf genes lie on band q21-25 of chromosome 9 (Yang *et al.*, 1984). Human Lf shows 59 and 49% sequence homology with human Tf and hen ovo-Tf respectively (Metz-Boutigue, 1984). On the other hand, Lf shows marked differences from Tf in antigenic determinants, tryptic peptide patterns, isoelectric point, the disposition and composition of the carbohydrate moieties (see section 1.1.3), and finally Lf has a much higher affinity for Fe³⁺ than that of Tf (Aisen and Leibman, 1972).

Lf is found totally or partially complexed to 80% of lysozyme present in milk probably with a ratio of 2 molecules of lysozyme to one of Lf and it complexes also with some sialyted glycopeptides of unknown role (Montreuil *et al.*, 1985). It is also found to be associated with secretory IgA (Watanabe *et al.*, 1984) by ionic interaction, and to an antitryptic factor (Montreuil *et al.*, 1985). It has also been observed that human Lf binds to murine macrophages and the presence of specific receptors has been reported (van Snick and Masson, 1976; Birgens *et al.*, 1983). The binding of Lf to adenocarcinoma cells (Amouric *et al.*, 1984), to lymphocytic cell lines (Hashizume *et al.*, 1983), and to human T-lymphocytes (Mazurier et al., 1989) has also been reported.

The major function of Lf is thought to involve bacteriostatic activity (Oram and Reiter, 1968) by a mechanism of iron deprivation, and perhaps in concert with other compounds such as sIgA (Oram and Reiter, 1968; Bullen *et al.*, 1972; Rogers and Synge, 1978), or lysozyme (Montreuil *et al.*, 1985), which represent a powerful system for defence of mucosae. It has been suggested that in the small intestine Lf would facilitate iron absorption by donating iron to specific receptors at the brush border membrane (Cox *et al.*, 1979; Mazurier *et al.*, 1985; Lonnerdal, 1985), although others propose an inhibitory effect (Brock, 1980). Nevertheless, a specific LfR was identified in the mouse small intestinal brush-border membrane (Hu *et al.*, 1988). In neutrophils Lf

may function as a regulator of granulopoisis (Broxmeyer, 1989; Fletcher, 1989). It has been reported that Lf regulates the adhesion of neutrophils during the inflammatory response, by promoting aggregation and adherence of those cells to endothelial cells (Oseas et al., 1981). Lf also enhances adherent natural killer cell cytotoxicity, while inhibiting antibody dependent cellular cytotoxicity (Nishiya and Horwitz, 1982). Lf also suppresses antibody production (Duncan and McArthur, 1981) as well as inhibiting colony stimulating factor (CSF) production (Broxmeyer et al., 1978) All these data suggest that Lf may play a role in regulation of the immune response. It has also been reported that Lf may enhance cell proliferation of a human colon adenocarcinoma cell line in the presence of low concentrations of iron (Amouric et al., 1984), and stimulates proliferation of rat crypt cells (Nichols et al., 1987). Recently Mazurier et al., (1989) have reported that PHA-stimulated human lymphocytes express specific surface receptors for Lf, and that the addition of Lf increased the proliferative activity of these cells in a similar way to Tf. However, others claim that it suppresses T-cell proliferation induced by mixed lymphocyte culture (MLC) or (PHA) (Slater and Fletcher, 1987), and that the mechanisms of suppression involve the chelating property of Lf (Richie et al., 1987).

1.1.3 Structure and iron binding properties of the transferrins

Transferrins evolved from a smaller ancestor some 200-500 million years ago, which was probably a polypeptide chain one half the size of the contemporary vertebrate protein (Williams, 1982). This notion is supported by the homology found in the amino acid sequence of the carboxyl halves of vertebrate Tfs (Metz-Boutigue *et al.*, 1984). Following an intragenic duplication that produced the amplified Tf gene, the Tf molecule gained an additional half and another iron binding site, and gave one chain with potential homology between the two halves (MacGillivray *et al.*, 1983; Park *et al.*, 1985). Amino acid sequencing and fragmentation, and X-ray diffraction studies, have concluded that Tfs are folded to form two more or less independent globular lobes, each containing one of the iron-binding sites at equivalent locations (MacGillivray *et al.*, 1983; Anderson *et al.*, 1987; 1989; Bailey *et al.*, 1988). They are connected by a short α -helix consisting of 7 residues in Tf and 9 in ovoTf. Each lobe contains two domains, with the iron binding site at the interface in highly hydrophilic environment.

Full length cDNAs have been cloned for human and rat serum Tf (Yang *et al.*, 1984; Uzan *et al.*, 1984; Levin *et al.*, 1984), chicken ovo-Tf (Jeltsch and Chambon, 1982), and melano-Tf (Rose *et al.*, 1986). Human Tf cDNA has an open frame encoding a protein of 679 amino acids plus the leading sequence of 19 amino acids which probably constitutes a signal for protein secretion since it is not found in the mature protein (Yang *et al.*, 1984; Uzan *et al.*, 1984). A homology of 35-40% was found between amino acids 1-336 and 337-679 in human Tf. The presence of two lobes in the Tf molecule with this high degree of homology has been demonstrated by several techniques (Williams, 1975; Brock and Arzabe, 1976; Lineback-Zins and Brew, 1980; MacGillivray *et al.*, 1983; Anderson *et al.*, 1987). The corresponding lobes in Lf comprise residues 1-332 and 342-686. Melano-Tf has in addition an extra
25 amino acid residues at the C-terminus. This peptide has highly hydrophobic characteristics and is thought to act as a membrane anchor. Greater homology has been found (46%) between the two domains of melano-Tf (Rose *et al.*, 1986).

The N-terminal lobe of all the Tfs has been found to contain 6 disulphide bridges, whereas the C-terminal lobe of Tf contains 13 against 9 for Lf and 10 for ovo-Tf (Montreuil *et al.*, 1985). Beside this difference in the number, there are differences in the localization on each lobe i.e. whether they are local or widely spaced. These bridges serve to stabilize the conformation of the molecule and make its structure compact.

Tfs contain approximately 6% carbohydrate. The attachment sites are widely distributed over the molecule surface (Baker *et al.*, 1987). Human Tf and Lf contain two glycans of the N-acetyllactosaminic type (Spik 1982; Spik *et al.*, 1985), while ovo-Tf contains only one glycan chain (Dorland *et al.*, 1979; Montreuil *et al.*, 1985). The two glycosylated sites of Tf are located in the C-terminal half at residues 413 and 611 (MacGillivray *et al.*, 1983), while in Lf each site is present in each lobe of the protein (residues 137 and 478) (Baker *et al.*, 1987). The only glycan site of ovo-Tf is located in the C-terminal part (residue 473) (Williams *et al.*, 1982; Metz Boutigue *et al.*, 1984). Little is known about the role of these glycan structures in the physiological function of the protein.

The summation of the polypeptide chain of 679 amino acid residues and the two N-linked complex type glycan chains results in a calculated molecular weight of 79.570 D. Human serum-Tf has an isoelectric point (pI) ranging between 5.6-5.9 (Hovanessian and Awdeh, 1976). Lf has a much higher pI ranging from 8.2-9.6 (Kinkade *et al.*, 1976). This is due to the presence only on Lf of a large number of basic residues at the surface of the molecule (Anderson *et al.*, 1989). OvoTf has a pI similar to that of serum Tf (Bain and Deutsch, 1948).

Tfs have two specific sites for iron. Therefore, Tfs may exist in the apo-, monoferric (with either the N- or C-terminal binding site occupied), and diferric form. Earlier studies have shown that the sites are not equally populated. It has been shown that at physiological pH, there is a preferential occupation of the N-terminal domain by iron (Leibman and Aisen, 1979). Zak and Aisen (1986) have found a considerable range in the ratio of occupancies of N-terminal and Cterminal sites between 21 normal human serum specimens with the Nterminal site predominantly occupied in most subjects. However, van Eijk and van Noort (1986) demonstrated that there is a shift of iron from the N- to the C-terminal binding site in Tf of stored human serum, but not in comparable preparations of purified Tf in PBS. On the other hand, it has been reported that the two sites are equally populated in the 1980) by using urea gel circulation (Williams and Moreton, electrophoresis techniques. Another report based on isoelectric focusing (IEF), has suggested that iron is randomly distributed, implying that the two sites are equally occupied whatever the level of saturation (Huebers et al., 1984). The distribution of iron donated to apo-Tf is dependent on the nature of the presenting chelate. When iron was added to rabbit Tf as a complex with nitrilotriacetic acid (FeNTA), the C-terminal binding site was occupied, whereas iron added as ferric citrate occupied the N-

terminal site (Heaphy and Williams, 1982).

Apo-Tfs are colourless, but develop a red brown colour with maximum absorbance at 465-470 nm as the iron is bound (Schlabach and Bates, 1975). Apart from iron, other cations can bind to Tfs, including copper, vanadium, chromium, aluminum, cobalt, gallium, manganese, magnesium, plutonium, indium, zinc, terbium, europium, platinum, americium, and curium (reviewed by Huebers and Finch, 1987). The binding of iron to apo-Tf leads to conformational changes in the molecule, which becomes less susceptible to proteolytic degradation and denaturation (Makey and Seal, 1976; Hovanessian and Awdeh, 1976; Esparza and Brock, 1980). It has also been noticed that some antigenic sites of apo-Tf become hidden when iron is bound to the molecule (Tengerdy *et al.*, 1966).

The affinity of iron binding by Tf is maximal under physiological conditions (Morgan, 1981), the association constant for human Tf being about 10^{20} M⁻¹ (Aisen *et al.*, 1978), and that for Lf some 26 times greater (Aisen and Leibman, 1972). However, the affinity decreases as the pH is reduced. It is frequently stated that iron starts to dissociate from the molecule of Tf at pH 5-6, and is completely released at pH 4.5 (Morgan, 1981). Lactoferrin does not release iron unless the pH is reduced to about 2 (Masson and Heremans, 1968). Iron is also lost at high pH, about 9-10 (Zapolski and Princiotto, 1980). However in the presence of competing chelators this might vary considerably. The iron binding sites of Tfs may differ in their binding affinities. At pH 6.7, the affinity of the N-terminal site of the human Tf for iron is less than 1/20 that of the C-terminal site (Evans and Williams, 1978), but at pH 7.4 the

affinity differs by a factor of only 5 or 6.

The requirement for synergistic binding of an anion for the binding of each iron to the molecule is now well established. The recent crystallographic structure analysis of Anderson *et al* (1989) favoured a carbonate to be the most likely anion involved in the binding of iron.

The high resolution of the Lf molecule (3.2 Å) of Anderson et al (1987) and the more recent one (2.8 Å) (Anderson et al., 1989), and rabbit serum Tf (3.3 Å) of Bailey et al (1988) have identified the amino acid residues involved in the binding of iron and the anion. Each iron is coordinated by four protein ligands, which is in accordance with the earlier finding of Chasteen (1983). According to Bailey et al (1988) these consist of phenolate oxygens of two tyrosines 93 and 191 (Tyr 447 and 540 in C-lobe), which is in line with what Pecoraro et al. (1981) have reported; the imidazole nitrogen of one histidine 252 (His 609 in Clobe), which was also reported earlier (Krysteva et al., 1975; Rogers et al., 1977; Zweier and Aisen, 1977) and the carboxylate oxygen of asparatate 61 (Asp 407 in C-lobe) plus a probable CO₁²⁻ or HCO₁⁻ anion bound to iron and to an adjacent arginine side chain 121 (Arg 477) which is highly conserved among Tfs. This is in contrast with earlier work of Chasteen (1983) who reported that two histidines play a role in the iron binding site. The remaining coordination site of the iron atom is linked to a water molecule.

1.1.4 Iron-storage proteins

1.1.4.1 Ferritin

As was mentioned above (see section 1.1.1), ionic iron is capable of mediating reduction of oxygen, giving rise to unstable intermediates. These potentially noxious compounds are known as free radicals which are highly reactive and could be very harmful. Therefore, an elaborate system of segregation and storage of iron is necessary to prevent iron toxicity and allow reutilization of stored iron when needed. This essential task is accomplished by the storage iron protein ferritin. Haemosiderin, its lysosomal degradation product, is also thought to play this role (see next section).

Ft is an iron storage protein used to maintain iron in an available non-toxic form within the cell. It is found in most cell types of vertebrates, in higher plants, fungi, and bacteria (Theil, 1987). The iron-free protein apo-Ft, which has a molecular weight of 450 Kd (Harrison *et al.*, 1967) is a roughly spherical coat with an outer diameter of about 125 Å, surrounding a core of about 70-80 Å across which houses iron atoms as microcrystalline polymers of ferric hydroxyphosphate (Ford *et al.*, 1984). The molecule of Ft consists of 24 subunits arranged in a form that leaves 6 channels 3-4 Å wide (Harrison *et al.*, 1980), through which iron can pass into and out of the molecule. These channels play a crucial role in the the iron binding process which is helped by side chains of amino acid residues at the wall of these channels. The central cavity can store up to 4500 iron atoms together with variable amounts of phosphate (Mann *et al.*, 1986).

The Ft molecule is made of different sorts of subunits. The isoferritins are hybrid molecules composed of varying proportions of these different subunits. Three subunits have been described (reviewed by Arosio, 1989). The light (L) subunit, is made of 174 amino acids, has a molecular weight of 19 Kd and a high pI, and is predominant in iron loaded tissues eg. the liver and spleen. The heavy (H) subunit, is more acidic than the L one, is made up of 182 amino acids and has a molecular weight of 21 Kd and a lower pI. It is predominant in iron poor tissues eg. heart, lymphoid cells, and malignant cells. Finally the G subunit (for glycosylated), with a molecular weight of 24 Kd, has been isolated from human serum (Cragg et al., 1981), but small amounts may also be present in tissues. L-chains from horse, rat, and human, share about 85% of sequence homology, while H- and L-chains differ by about 45% (Costanzo et al., 1984; Leibold et al., 1984; Boyd et al., 1985).

Ft is present in serum and can be used as a clinical indication of iron status (Worwood, 1986). It is also found in small amounts in milk (Arosio *et al.*, 1984). Adult human heart, kidney, pancreas, and placental Fts as well as Ft derived from neoplasms, contain more acidic iso-Fts, which are rich in H-chains, than liver and spleen Fts, which are more basic and rich in L-chains (Drysdale, 1970).

In humans, the H- and L-chains of Ft are derived from multigene families. Southern blot analyses and chromosomal *in situ* hybridization, indicate that there are about 15 H-chain Ft sequences and 5 L-chain Ft sequences in human DNA (Cragg *et al.*, 1985; Jain *et al.*, 1985). These are found on at least 7 different chromosomes for the Ft H genes

and on 3 chromosomes for the Ft L genes (McGill et al., 1987).

Both the channels and the core fulfill the role of binding iron to the molecule. When Fe^{2+} atoms are added to apo-Ft they migrate to the interior by passing through the channels, probably aided by amino acid side-chains at the surface of these channels. Once in, Fe^{2+} is oxidized to Fe^{3+} and polymerises to form iron-core nucleation clusters (Bakker and Boyer, 1986). The loss of iron from the core via the channels requires reducing agents to form ferrous iron such as dithionite or flavines (Funk *et al.*, 1985) or free radicals (Thomas and Aust, 1986). Subsequently, the ferrous iron leaves the Ft core and must be complexed by a suitable chelator such as EDTA (Biemond *et al.*, 1988). However the nature of the reducing substance(s) or the chelator(s) is unknown under physiological conditions. Release of iron is also enhanced at low pH (Watt *et al.*, 1985). On the other hand, others have suggested that the mechanism by which iron is released from Ft is through the constitutive degradation of the protein by lysosomes (Roberts and Bomford, 1988).

The administration of a variety of iron compounds to animals, results in an increase in the Ft content of many tissues (Harrison *et al.*, 1980). Other factors related to cell differentiation/proliferation regulate the transcription of Ft. Little is known about the molecular basis underlying the relationship between cellular transformation and Ft. Lymphoid cells contain different levels of H and L Ft according to their lineage, proliferative status, and anatomical site (Dörner, *et al.*, 1983; Vezzoni *et al.*, 1986). Increased Ft synthesis has been reported to occur during cell differentiation (Fibach *et al.*, 1985). Several studies have shown an association between the development of cancer and

increased serum Ft levels (Moroz *et al.*, 1984). This secreted protein may bind to subpopulations of circulating lymphocytes and ultimately affect their cell-mediated immunity (Moroz *et al.*, 1977).

Mechanisms of regulation of Ft synthesis vary depending on the cell type. In cells in which stored iron is used for intracellular needs, such as proliferating cells, there is a transcriptional control where the mechanism of regulation may depend on mRNA concentration (Cairo *et al.* 1985). It has been shown that there was an increase in the relative concentration of mRNA during the differentiation process indicating either changes in transcription to produce the appropriate Ft to meet the needs of each cell type or changes in mRNA stability (Theil, 1987). It has been shown that the increased Ft mRNA levels in differentiating Friend cells arises from increased transcription rather than decreased mRNA breakdown (Beaumont *et al.*, 1987). Iron also increases Ft transcription in these non-specialized iron storage cells. It has been shown that iron administration causes an increase in transcription of Ft L mRNA (Cairo *et al.*, 1986).

On the other hand, in specialized iron storage cells i.e. hepatocytes, the mechanism of regulation depends more on translational control. Iron administration to rats (Kohgo *et al.*, 1980; Bomford *et al.*, 1981), and to cell cultures (Goto *et al.*, 1983; Rittling and Woodworth, 1984; Rogers and Munro, 1987) stimulates subunit synthesis by an actinomycin-resistant mechanism (Zahringer *et al.*, 1976; Drysdale and Munro, 1966) involving mobilization of L subunit mRNAs from a cytoplasmic pool of inactive messages on to polyribosomes (Aziz and Munro, 1986; Rogers and Munro, 1987).

Ferritin genes for both H and L subunits have a highly conserved 28 base sequence in their 5 untranslated region (5 UTR) that is essential for the translational regulation of the Ft L (Aziz and Munro, 1987) and H mRNAs (Hentze et al., 1987) by iron. Deleting part of the 5' UTR containing the conserved 28 base sequence (Aziz and Munro, 1987) eliminates the response to iron, suggesting that iron-sensitive factor(s) in the cytoplasm may bind to this sequence and regulate the availability of both Ft messages for translation. Leibold and Munro (1988) have identified an 87 Kd cytoplasmic protein interacting with RNA sequence in 5' UTR of Ft H and L subunit mRNA. Similar potential sequences are also present in the 3' untranslated region of the TfR mRNA (see below). The degree of complex formation between mRNA and the cytoplasmic protein has been shown to be affected by treatment of rats or cells with iron. This confirms that intracellular iron levels regulate Ft synthesis by influencing the specific association of this repressor protein to the iron responsive element (IRE) on the Ft H and L subunit mRNA.

Inflammation can also increase Ft synthesis (Konijn *et al.*, 1981). Campbell *et al* (1989) have reported that inflammation causes a shift of Ft mRNA to the polyribosomes in liver and spleen. A possible role of an inflammatory cytokine i.e. tumour necrosis factor in modulating Ft synthesis by inducing Ft gene expression has been suggested (Torti *et al.*, 1988). Moreover Rogers (unpublished cited by Campbell *et al.*, 1989) has shown that exposure of hepatoma cells to IL-1 for 12 h shifts Ft mRNA to polyribosomes, suggesting that this cytokine may be the modulator of the inflammatory reaction by which Ft synthesis is enhanced. As far as lymphoid cells are concerned, Ft, or a subpopulation of isoFts, may play a role in the regulation of the cellular growth of lymphocytes and the immune response (reviewed by Broxmeyer, 1989) as well as mediating T-cell surveillance (Dörner *et al.*, 1980). Synthesis of Ft by lymphocytes is increased by *in vitro* culturing irrespective of whether a mitogen is present of not (Pattanapanyasat *et al.*, 1987; 1988), suggesting that a mechanism for increasing Ft synthesis by activation may exist that is distinct from iron stimulation.

The biological importance of Ft can be summarized by the following functions, though these functions may be related :

- Storage of iron in specialized iron storage cells such as hepatocytes and macrophages, for use by other cells. These cells tend to contain the largest amount of Ft, and usually show the largest increase in Ft levels under conditions of iron overload.
- Storage of iron for subsequent intracellular metabolic use (in cells that are undergoing proliferation, differentiation, and development).
- Storage of iron for detoxification (in iron overload), which probably constitutes the most evident role of Ft.

1.1.4.2 Haemosiderin

The other iron sequestering protein haemosiderin is the major iron storage protein in iron overload. Hs is a purely lysosomal product which ultrastructurally resembles Ft. The main difference between Ft and Hs is that the latter is water insoluble, probably due to the lack of an intact protein shell. The structure of Hs is less well defined than that of Ft. It consists of iron containing granules visible in the electron microscope as massive aggregates of electron dense particles of irregular shape (Dickson et al., 1988) resembling Ft iron cores. These particles consist essentially of hydrated polymeric ferric oxide with some phosphate and peptide groups (Weir et al., 1984). It is generally assumed that Hs is a degradation product of Ft following its polymerization (Hoy and Jacobs, 1981). These polymers may be incorporated to lysosomes and converted into Hs. On the other hand, others are still doubtful if Hs is really a product of Ft (St Pierre et al., 1988; Dickson et al., 1988). Iron is much less readily mobilised from Hs than from Ft, and it is generally considered to be an unreactive form of storage iron, which helps in reducing unwanted release of reactive free iron in iron-rich tissues (O'Connell et al., 1986a).

1.1.5 Cellular iron uptake: the transferrin cycle

A primary function of Tf is to donate iron to erythrocyte precursors in the bone marrow, which have a large requirement for synthesis of haemoglobin. Iron is also required by non-erythroid cells for many metabolic needs, though the amounts needed are very small compared with erythroid cells.

1.1.5.1 Structure and regulation of the transferrin receptor

Iron uptake by erythroid and non-erythroid cells is a multistep

process. The chemical properties of iron, its extremely low solubility in aqueous solution, and its high reactivity in forming of active oxygen species, demand that in all steps of its transport from Tf into the cell, it is bound to a ligand. The initial step is the interaction and the attachment of Tf to a specific cell membrane receptor (TfR), which was first noted by Jandl *et al* (1959) on reticulocyte cell membranes. Since then large numbers of studies have been carried out exploring this field. It is now generally agreed that iron is transported from FeTf into the cell by a receptor mediated system.

The TfR is transmembrane glycoprotein (Hamilton, 1979) with a molecular weight of about 180 Kd. The receptor is known to be composed of two disulphide-linked subunits of about 90 Kd each (Seligman *et al.*, 1979; Enns and Sussman, 1981a; 1981b). Each subunit contains glycan chains, which account for about 5% by weight of the receptor (Enns and Sussman, 1981a), and are composed of sialic acid, N-acetyglucosamine, galactose, and palmitic acid (Omary and Trowbridge, 1981). Each subunit can bind one Tf molecule. The structural features of the Tf molecule which are involved in its interaction with the receptor are still not known.

The gene for the human TfR has been identified and cloned (McClelland *et al.*, 1984). Using *in situ* hybridization, it was shown that the gene is mapped on the long arm of chromosome 3 in close proximity to Tf and melano-Tf (p97) genes (Rabin *et al.*, 1985). During its biosynthesis, the human TfR undergoes extensive modifications. Those modifications include asparagine linked glycosylation, dimer formation, intersubunit disulphide bond formation, acylation with

26

palmitate, and phosphorylation. Examination of the promoter regions of Tf genes and TfR genes indicates some similarities in orientation (Bowman, *et al.*, 1988).

There is a reverse correlation between receptor number and iron supply in cell culture (Rao et al., 1985; Taetle et al., 1985; Ward et al., 1984). The level of intracellular iron regulates the expression of the receptor, as the iron chelating agent desferrioxamine (DFO) increased 2.5-fold the in vitro translation of TfR by polyadenylated mRNA isolated from K562 erythroleukaemic cells (Rao et al., 1985), while treatment with ferric ammonium citrate and FeTf produced 25 and 50% reductions, respectively. By using light microscopic and immunochemical methods (Sciot et al., 1987) and immunoelectron microscopy (de Vos et al., a disappearance of hepatic TfR expression was reported in 1988), haemochromatosis when iron overload is severe. It has been shown that DFO treatment of K562 cells causes an early increase of TfR mRNA, while haemin treatment has the opposite effect (Louache et al., 1985). The changes in the level of surface TfR are mainly due to mechanisms that affect TfR mRNA pool. It has been shown that the regulatory domain is located in the 3' untranslated region (3' UTR) (Owen and Kühn, 1987), which Müllner and Kühn (1988) have demonstrated regulates the expression of the TfR through differential degradation of mRNA according to cellular iron needs. While treatment of cells with DFO produces no significant effect on TfR gene transcription nor on the nuclear processing of primary transcripts, cytoplasmic TfR mRNA becomes 20-fold more stable. Full induction of TfR mRNA requires about 15 h. After the addition of iron salts, the TfR mRNA decays with a half life of 1.5 h. These observations suggest that one mechanism for

this regulation may be an iron-dependent control of TfR mRNA stability.

Casey et al (1988) have identified 5 potential stem-loop structures resembling the Ft mRNA iron regulatory element (IRE) (see section 1.1.4.1) on the sequence of the 3' untranslated region of human TfR mRNA. When two of these 5 elements were inserted into the 5' untranslated region of an indicator gene transcript, they conferred iron regulation of translation and confirmed the importance of the IRE regions in conferring the property of iron regulation of mRNA. However, with the TfR IRE in this location (5') they found that raising intracellular iron caused increased indicator protein synthesis and lowering intracellular iron caused decreased indicator protein synthesis, a regulation pattern opposite to that seen with the TfR gene. These experiment showed that the IRE location within the mRNA, i.e., 5' or 3', determined the pattern of this iron regulation. It has been shown that the effect of iron on TfR mRNA is exerted through the binding of a cytoplasmic protein to the regulatory region of TfR mRNA, and binding activity is inversely correlated with the level of intracellular iron (Müllner et al., 1989). They proposed that the stem-loop is the substrate of an RNAase that is sterically hindered by the binding of several iron regulatory factor molecules to the adjacent palindromes. This cytoplasmic protein seems to be similar to the one identified by Leibold and Munro (1988) which binds to the untranslated region of the Ft mRNA (see section 1.1.4.1). This IRE-binding protein may therefore be involved in regulation of expression of more than one protein involved in cellular iron metabolism. Rouault et al (1988) have reported that the 3' regulatory region of TfR mRNA can compete with the 5' Ft mRNA IRE for the interaction with IRE-binding

protein, resulting in opposing effects on the biosynthetic rate of Ft and the TfR. However, the mechanism of regulation of TfR and Ft may differ according to the cell type. In human macrophages, iron apparently causes an increase in both TfR mRNA and Ft synthesis. Recently, Neupert *et al* (1990) have isolated and purified the IRE-binding protein which binds to specific palindromic elements in the 5' and 3' untranslated sequences of Ft and TfR mRNA respectively and found it to consist of equimolar amounts of two proteins with molecular weights of 95 and 100 Kd when analysed by SDS/PAGE.

The proliferative status also regulates TfR expression. A highly proliferative status of cells is generally associated with a high density of the TfR (Sutherland et al., 1981). Krönke et al (1985) reported increased TfR transcription after activation of peripheral T-cells with PHA and phorbol myristate acetate (PMA). It has been shown that the factors that stimulate cell growth increase the expression on the cell surface of TfR (Lombardi et al., 1989). They showed 5-fold reduction in TfR transcripts within 24 h of DMSO treatment and these reductions proceeded loss of surface-expressed TfR. It has been reported that both the proliferation and iron dependent changes in TfR mRNA levels occur mainly post-transcriptionally (Kühn et al., 1989). Iron level and proliferative status appear to act on different mechanisms, as deletion of sequences on the 3['] noncoding region of the cDNA abolishes the iron dependent, but not the proliferation dependent regulation (Owen and Kühn, 1987). It has been reported that activated mouse spleen cells show 50-fold increase in TfR mRNA without any induction of TfR gene transcription.

1.1.5.2 Cellular iron uptake from transferrin

The uptake of iron from Tf is not random. Association constants for the binding of Tf to its receptor have been found in the range of 5×10^{6} -2.3 x 10^{9} M⁻¹ depending on the temperature, pH, type of cell, and the degree of saturation (Hamilton *et al.*, 1979; Young and Aisen, 1980; Ward *et al.*, 1982). It has been shown that there is a marked preference of receptors for diferric over monoferric Tf (Huebers *et al.*, 1984). Using the interaction of different forms of ¹²⁵I-labelled Tf and rabbit reticulocytes Young *et al* (1984) have found that the association constant for receptor binding is 4.6 x 10^{6} M⁻¹ for apoTf, 2.5 x 10^{7} M⁻¹ for monoferric Tf (C-terminal), 2.8 x 10^{7} M⁻¹ for monoferric (Nterminal), and 1.1 x 10^{8} M⁻¹ for diferric Tf. It therefore seems to be that the amount of iron taken up depends on the number of each molecular species present.

The effect of pH on the binding of apoTf and diferric Tf to reticulocyte membrane receptors was investigated by Morgan (1983), who found that the binding of apoTf to the cell membrane was of high affinity below pH 6.5. By contrast, diferric Tf showed high binding between pH 7.0 and pH 8.0 with much weaker binding below pH 6.5.

The second step of iron uptake is the internalization of Tf, which is achieved by the general process of receptor mediated endocytosis. Whether phosphorylation-dephosphorylation of the receptor may give a signal for internalization, and play a role in directing receptor traffic from the surface on to the endosome network and back into the surface is a matter of considerable controversy. It has been reported that TfR undergoes protein kinase C-dependent phosphorylation and internalization in response to treatment by phorbol diesters (Kohno et al., 1985; May and Tyler, 1987; May et al., 1986). FeTf has been found to induce a 10-fold increase in the activity of protein kinase C in CCRF-CEM cells probably through increasing synthesis of the enzyme, while apoTf and iron in the form of Fe-citrate or complexes of Tf with several other metals, did not increase protein kinase activity (Phillips et al., 1987). However, gene transfection experiments have failed to prove a role for receptor phosphorylation in mediating TfR internalization or cycling (Rothenberger et al., 1987; Davis and Meisner, 1987; Zerial et al., 1987; McGraw et al., 1987). Binding of Tf to its receptor generates a lateral mobility of TfR complex to achieve an area of increased numbers of ligand receptor complexes. However there is still a controversy over whether the internalization process is spontaneous and occurs independently of ligand binding (Watts, 1985) or whether it requires Tf bound to the receptor to switch on the process (Klausner et al., 1984a). The internalization process is achieved by a specific microdomain invagination of the cell membrane which is supported by clathrin to form coated-pits. Little is known about the mechanism by which receptors are concentrated selectively in coated-pits prior to internalization. The mechanism might involve passive diffusion of receptors in the membrane and their selective retention in coated-pits (Hopkins, 1985), or direct movement mediated by receptor-protein interactions. Alternatively, the aggregation of receptors could induce the formation of coated-pits. The coated pits are pulled from the cell surface into the cytoplasm through assembly of the clathrin coats. The vacuoles generated rapidly lose their clathrin, producing vesicles which associate and fuse generating the socalled endosomes (Willingham et al., 1984). Internalization is rapid and an energy requiring process (Karin and Mintz, 1981).

The mildly acidic environment within the endosome of pH 5-6.5 (van Renswoude *et al.*, 1982) protonates the anion $(CO_3^{2-} \text{ or } HCO_3^{-})$ in the binding site adjacent to the iron and provokes iron release (Crichton, 1985). It has been shown that methylamine, ammonium chloride, and chloroquine, which increase vesicular pH, decrease the release of iron from the protein to the cell (Morgan, 1981) without inhibiting internalization of the Tf. Other studies employing mutants of CHO cells, which are defective in endosomal acidification, have demonstrated that such cells cannot remove iron from internalized Tf (Klausner *et al.*, 1984b).

Little Tf is degraded during the process of iron delivery (Karin and Mintz, 1981), and instead of becoming separated apo-Tf remains tightly coupled to the receptor inside the endocytic vesicle, escapes lysosomal degradation, and returns intact together with its receptor to the cell surface by an unidentified mechanism of some kind of reverse endocytosis within a microtubule-associated endosome (Willingham *et al.*, 1984; Willingham and Pastan, 1985). At the surface the apo-Tf dissociates from the receptor into the extracellular milieu, since the apo-Tf-TfR complex is poorly stable at neutral pH. The Tf is then available for many plasma-to-cell cycles, as is indicated by its long plasma half life (7-10 days) compared with its passenger iron (1.5-2 h) (Katz, 1961).

1.1.6 Metabolism and distribution of intracellular iron

Little information is available on the fate of iron within the cell. Once iron is released from the Tf inside the cell, the metal undergoes a series of intracellular transfers. In non-erythroid cells intracellular iron exists in a number of forms that can be summarized as iron containing enzyme systems, stored iron (Ft and Hs), and a "labile pool".

The iron released from Tf may initially enter lysosomes, and it then somehow finds its way across the lysosomal membrane. Despite these uncertainties about the mode of transport of iron to the cytosol, transport of iron across biological membranes usually involves a ferricferrous transition (Romslo, 1980). Low molecular weight complexes of Fe³⁺ have been postulated as the species which transport iron through membranes (May and Williams, 1980).

Once in the cytosol, iron passes to an unidentified pool which is dialysable and available for binding by iron chelating agents. The precise identity of the ligands which bind the iron in this pool are not known, but the multiplicity of possible reactions of iron with low molecular weight molecules such as cysteine, glutathione, ascorbic acid, glucose, fructose, and other molecules such as ATP and other nucleotides (Romslo, 1980), has given rise to the concept of an intracellular transit pool, which constitutes a pivot of intracellular iron metabolism and may act as a precursor for functional and storage compounds (Lynch *et al.*, 1974). The idea of a "labile pool" or "transit pool" has emerged from difficulties in explaining intracellular iron metabolism without postulating such an intermediate pool (Jacobs *et al.*, 1977) between the extracellular

(transport) and intracellular (storage iron and iron containing proteins) forms of the metal. In reticulocytes, iron complexes with a molecular weight below 2000 d were detected (Pollack and Campana, 1981; Bakkeren *et al.*, 1985). In mature erythrocytes, GTP and ATP chelated iron were found in an amazing concentration of up to 100 μ M (Bartlett, 1976). It has also been shown that rat liver, kidney, heart, brain, spleen, and pig brain, have small but significant amounts of iron in a form available to DFO (Linder *et al.*, 1983).

In reticulocytes, the bulk of intracellular iron is found in haemoglobin. In non-erythroid cells the proportion used for haem synthesis is much less, only 5% being found in haem compounds in PHA-stimulated lymphocytes incubated with Tf for 3 h (Bomford *et al.*, 1986). The major proportion of the intracellular non-haem iron is found in Ft and Hs. After 2 h incubation of isolated rat hepatocytes with ⁵⁹FeTf almost 70% of iron can be recovered in Ft (Young *et al.*, 1985). The other pool of iron consists of iron incorporated into a number of nonhaem iron proteins such as electron transport iron sulphur proteins, oxygenases, aconitase, and ribonucleotide reductase.

Little is known about the mechanisms by which intracellular iron is mobilised from Ft. The presence of apoTf does not appear to be essential for the mobilization of iron from macrophages (Saito *et al.*, 1986). It thus appears that the inverse Tf cycle (Dautry-Varsat *et al.*, 1983; Klausner *et al.*, 1983) would probably not be appropriate to explain the mechanism of intracellular iron mobilization.

34



PART TWO

LITERATURE REVIEW OF IRON AND LYMPHOCYTE FUNCTION

.

1.2.1 The lymphocyte

The function of the immune system is to discriminate between self and non-self, and mount responses to rid the body of potentially life threatening foreign organisms, eg. viruses. In higher vertebrates adaptive immune responses, i.e. responses which give rise to memory, are mediated by lymphocytes. Two main families of lymphocytes can be distinguished, B cells, and T-cells. The T-cells first differentiate into immunocompetent cells by spending a training period in the thymus, while B cells undergo their education in the bone marrow itself. These two interactive subpopulations have many characteristics in common; morphologically, in their habitat, and in sharing a number of cell surface molecules. However they are distinct from each other in their density, their electrophoretic mobility, their half life, their migration after intravenous injection, in a number of cell surface molecules expressed only on T or B cells, and most importantly in the mechanisms of their effector functions.

The immune response occurs most effectively in structured secondary lymphoid tissue. These tissues become populated by cells of reticular origin and by macrophages and lymphocytes derived from bone marrow stem cells. The lymph nodes filter and screen lymph flowing from the body tissues while the spleen filters the blood. Antigen is presented to lymphocytes by macrophages and by specialized antigenpresenting cells in T- and B-cell areas within the lymphoid tissue. Blymphocytes exhibit antigenic specificity. They proliferate in response to a particular antigen and differentiate into non-proliferating but antibodysecreting plasma cells. T-lymphocytes also exhibit antigenic specificity, and they also proliferate and differentiate in the presence of antigen, releasing substances called lymphokines, which have important biological properties. While T-cells do not synthesise antibodies, they do fulfill many other immunologically important functions such as helping B-cells to synthesise antibodies, killing tumour cells, and regulating the immune response.

1.2.2 T-Lymphocyte interaction with antigen or mitogen: activation and proliferation

Activation of T-cells involves a shift from a quiescent cell, which is in the G_0 phase of the cell cycle into G_1 and S phase, and the cycle then proceeds through the G_2 phase towards cell division. T-cell activation takes place in sequential stages. Interaction of the antigen or mitogen with its membrane receptor is a crucial first step. Under physiological conditions, such ligand-receptor interaction occurs at the interface of the plasma membranes of an antigen-specific T-cell and an antigen presenting cell (APC) or target cell. These ligand-receptor binding events result in the transduction of these events into intracellular biochemical signals in the form of "second messengers" which influence specific targeted genes, receptive to these signals, that can become transcriptionally active or inactive. The manifestations of T-cell activation include the production of lymphokines, the appearance of new cell surface proteins (which include growth factor receptors), the acquisition of cytolytic effector function, and as a consequence of the production of growth factors and their receptors, proliferation.

Activation of the T-cell induced by an antigen on the surface of an antigen presenting cell must involve an interaction with the T-cell antigen receptor (Ti). This is an immunoglobulin-like heterodimer composed of α - and β -chains which functions as a receptor for combination of antigen and a major histocompatibility complex (MHC)-encoded glycoprotein (Dembic et al., 1986; Bjorkman et al., 1987). An alternative heterodimer TCR containing two different chains i. e. γ and δ has been identified on the surface of a small fraction of T-cells (Chien et al., 1987). The TCR is found on the surface of the majority of mature Tlymphocytes (Aparicio et al., 1989) and may exist as a non-disulphidelinked heterodimer or as a single chain in non-covalent association with a T-cell specific molecular complex composed of at least 5 distinct chains, collectively termed T3 or CD3 to form the (T3/Ti or TCR) complex (Brenner *et al.*, 1985). The function of the TCR is a matter of considerable speculation. A suggested role is as an intermediary that delivers transmembrane signals subsequent to antigen receptor occupancy. This is supported by the findings that antibodies directed against CD3 function as polyclonal agonists which are able to mimic stimulation with antigen leading to early manifestation of activation (O'Flynn et al., 1985; Oettegen et al., 1985), and later to lymphokine production (Chang et al., 1982; van Wauwe et al., 1984), expression of lymphokine receptors (Meuer et al., 1984a; Schwab et al., 1985; Tsoukas et al., 1985; Ledbetter et al., 1986), and proliferation (Van Wauwe et al., 1980).

In addition to its role in antigen-induced T-cell activation, the T3/Ti complex appears to be important in the activation of T-cells by the T-cell-specific mitogenic lectins PHA and Con A. Both lectins bind to

large numbers of T-cell surface glycoproteins (Sitkovsky *et al.*, 1984), Although the cell surface molecules responsible for the ability of these lectins to stimulate T-cells are still unknown, studies with solubilised cell surface proteins have demonstrated that Con A can bind to CD3 but not Ti, whereas PHA can interact with the Ti heterodimer but not to isolated CD3 chains (Kanellopoulos *et al.*, 1985). Evidence supporting the role of the T3/Ti complex in PHA- and Con A-induced T-cell activation is the observation that Jurkat mutant cells which fail to express the T3/Ti complex lose the capacity to produce IL-2 in response to either PHA or Con A (Weiss *et al.*, 1984; Weiss and Stobo, 1984). Moreover, reconstitution of the T3/Ti expression in one of these mutants by transfection resulted in the restoration of the PHA and Con A responsiveness of those cells (Weiss *et al.*, 1986).

CD2, the molecule which functions as sheep erythrocyte-binding protein on the surface of human T-cells and which has as many as 6 distinct epitopes (Martin *et al.*, 1983; Yang *et al.*, 1986) is also involved in the activation of T-lymphocytes and thymocytes, including those lacking T3/Ti antigen receptor complex (Sayre *et al.*, 1987). It has been demonstrated that certain combinations of α -CD2 monoclonal antibodies were mitogenic to T-cells, as measured by proliferation or IL-2 production (Meuer *et al.*, 1984b; Brottier *et al.*, 1985). Individual α -CD2 antibodies are insufficient in inducing T-cell activation. Other reports suggested that α -CD3 and α -CD2 mAb can synergize in inducing proliferation in cultures prepared with highly purified T-cells (Yang *et al.*, 1986).

A crucial number of mitogen molecules must bind to the lymphocyte

surface in such a way as to provide a sufficient surface stimulus. Maximum commitment of lymphocytes to DNA synthesis requires some 18 h exposure to mitogen (Wedner and Parker,1976). Following the binding, a redistribution of the ligand-receptor complex occurs. This interaction leads to rapid changes in the surface membrane, and to "cap" formation, as indicated by reacting lymphocytes with fluorescein-labelled Con A at 37° C (Rao, 1982).

Transduction of a transmembrane signal to T-lymphocyte nuclei via the antigen/mitogen receptors seems to occur via a limited number of biochemical pathways utilized by a variety of cell types, when they transit from a non-growing to a growing phase. The antigen/mitogen receptors on T-cells are linked to a polyphosphoinositide-specific phosphodiesterase (PPI-PDE), such that ligation of these receptors provokes polyphosphoinositide (PI) breakdown and formation of hydrolytic products that function as cellular second messengers (Fig.1). The precise nature and functions of the molecule(s) coupling antigen/mitogen receptors on T-cells to the PPI-PDE are still largely undefined. Inositol 1,4,5-triphosphate (IP_{2}) , a water soluble compound, and 1,2-diacyl-glycerol (DG) are released from hydrolysis of phosphatidyl inositol biphosphate (PIP₂) (Imboden and Stobo, 1985), which is found in the cell membrane in minute quantities. These two second messengers trigger two parallel pathways that act in concert to elicit a physiological response. The DG translocates Ca²⁺ phospholipiddependent protein kinase C (PKC) from the cytosol to the cell membrane and increases its activity by increasing its affinity for Ca^{2+} (Imboden et al., 1985), whereas IP₃ increases cytosolic free Ca^{2+} concentrations by



Figure 1Lymphocyte triggering transmembrane signals.Modified from Isakov et al (1986).

causing its release from intracellular stores i.e. mitochondria and endoplasmic reticulum (Berridge and Irvine, 1984) probably by binding to specific receptors within the endoplasmic reticulum (Streb et al., 1983). Calcium ions bind to and thereby activate calmodulin, a cytoplasmic protein which, in the presence of Ca²⁺, can activate a variety of different enzyme pathways including Ca²⁺-dependent protein kinases(Cheung, 1980; Means and Dedman, 1980; Berridge and Irvine, 1984). This is followed by a prolonged phase of Ca^{2+} influx through the plasma membrane (June et al., 1986). The increase in Ca²⁺ causes the activation of PKC which mediates phosphorylation of numerous proteins, yet unidentified. Calcium ionophores and phorbol esters can mimic parts of the activation process, whereas neither agent alone is mitogenic. Phorbol ester can bind and activate PKC without noticeable changes in cytosolic free Ca^{2+} concentration (Gelfand *et al.*, 1985). Ca^{2+} ionophore A23187 increases cytosolic free Ca²⁺ influx without influencing PKC activity (Tsein et al., 1982). The combination of Ca²⁺ ionophore and phorbol esters duplicate the role of DG and IP₃ and display strong synergism in inducing the genes for c-fos, c-myc, IL-2, IL-2R and IFN- γ , and in inducing T-lymphocytes to proliferate (Kumagai et al., 1987, 1988).

As mentioned above, after antigen/mitogen interaction with resting T-cells and the transduction of this interaction into intracellular biochemical signals, these activation signals, in turn, induce the *de novo* expression of certain sets of cellular genes required for T-cell proliferation. These include those encoding lymphokines such as IL-2 and its high affinity cellular receptor. The binding of IL-2 to its high affinity receptors results in their internalization and increases their rate of

removal from the cell surface resulting in a decrease of the level of those surface receptors which are replaced by low affinity receptors (Depper *et al.*, 1985). The subsequent binding of IL-2 to these receptors results in a second wave of gene activation leading to mitogenesis (Robb *et al.*, 1984; Stern and Smith, 1986).

One of these genes activated as a result of the binding of IL-2 to its receptor is the TfR gene (Neckers and Cossman, 1983). This is discussed in detail below.

1.2.3 The role of transferrin in proliferation and iron metabolism of lymphocytes

The proliferation of lymphocytes is a dynamic process which is highly controlled under normal conditions by stimulating, enhancing, and suppressing biomolecule-cell interactions. The iron binding proteins are one group of these molecules.

The importance of iron-containing Tf for the transformation of lymphocytes is well established (Tormey *et al.*, 1972; Dillner-Centrelind *et al.*, 1979; Brock, 1981, Anderson *et al.*, 1982; Brock and Mainou-Fowler, 1983; Brock *et al.*, 1986). The need of these cells to acquire iron is sufficient to account for the Tf requirement, although it has been proposed that Tf may fulfill an additional role unrelated to its iron-donating properties, i.e. its binding to its receptor generates a proliferation signal (Brock and Mainou Fowler, 1983; Seligman,1983). Manger *et al* (1986) have produced an α -TfR monoclonal antibody recognising an epitope different from the Tf binding site which stimulates the production of IL-2 by HUT 78, a T-cell leukaemic cell line, even in Tf-free medium. Others have demonstrated that Tf exercises its growth factor effect through acting as an electron acceptor for a diferric Tf reductase in cell membranes independently to donating iron to cells (Navas *et al.*, 1986). However, Brock *et al* (1986) have demonstrated that in lymphocyte cultures Tf fulfills its role as a growth factor exclusively as an iron donor.

As regards the question of whether Tf is the obligatory iron donor to proliferating lymphocytes, there are conflicting and contradictory reports. Some authors have shown that the addition of iron without Tf can give good proliferation and differentiation (Rudland *et al.*, 1977; Tanno and Takishima, 1982; Titeux *et al.*, 1984). On the other hand, others have reported that iron itself cannot support proliferation (Phillips and Azari, 1975; Brock, 1981). It has been demonstrated that Tf could be replaced by the lipophilic chelator pyridoxal isonicotinoyl hydrazone (PIH) which can promote proliferation by donating iron via a Tfindependent route (Brock and Stevenson, 1987).

It is generally agreed that DNA synthesis is the main event dependent upon Tf in lymphocyte function, though there may be other cellular metabolic events that are also iron-dependent. Iron uptake was shown to precede DNA synthesis (Brock and Rankin, 1981), because of the need of proliferating cells for iron for the production of the iron containing enzyme ribonucleotide reductase, which is a rate limiting step in lymphocyte proliferation (Hoffbrand *et al.*, 1976; Leberman *et al.*, 1984; Kay and Benzie, 1986). Like other non-proliferating cells, resting lymphocytes express few TfRs. Larrick and Cresswell (1979) were the first to show that when human peripheral blood lymphocytes were activated by mitogenic lectins such as PHA or Con A, TfR became prominently displayed at cell surface. This observation has been confirmed by many other investigators (Galbraith *et al.*, 1980; Omary *et al.*, 1980; Neckers, 1984). TfR expression is now recognised as one critical event in a complex programme of gene activation induced by mitogenic agents and culminating in progression of resting G_0 phase lymphocytes through G_1 and S phase of the cell cycle (see section 1.1.5.1).

The role of TfR in the iron metabolism of proliferating cells is still not clear, since despite comparable receptor numbers, iron uptake rates by mitogen stimulated lymphocytes are low compared to erythroid precursors (Bomford et al., 1983; Brock and Mainou-Fowler, 1983; 1984). It has been suggested that this Young and Bomford, hyperexpression of TfRs may permit normal immune function to be maintained when Tf saturation is reduced eg. in inflammation or iron deficiency. It has also been reported that the receptor density may vary with the growth phase of the cell (Larrick and Cresswell 1979). Little binding of Tf occurs in resting lymphocytes (Galbraith et al., 1980; Brock and Rankin, 1981; Khalfoun et al., 1986). This indicates that the cells regulate their requirements for iron by the expression of TfR (Iacopetta et al., 1982). Furthermore, interference with TfRs by specific antibodies can affect the cell cycle and growth (Trowbridge et 1982). Anti-TfR mAbs that block binding and iron uptake in al., activated lymphocytes inhibited proliferation, whereas antibodies that bind to TfR but do not block Tf binding are without effect (Mendelsohn et al., 1983; Brock et al., 1986; Taetle et al., 1986; Kemp et al.,

Similarly proliferation of murine lymphocytes in medium 1987). containing FCS can be blocked by addition of apo-mouse Tf, which presumably binds to the murine TfRs and prevents iron uptake from the bovine Tf in the medium, which has much lower binding affinity to the murine lymphocyte TfR than mouse Tf (Brock et al., 1986). Treatment of mitogen-stimulated lymphocytes with substances that inhibit proliferation such as IFN- α (Besancon et al., 1985), cyclosporine (Prince and John, 1986), or calcium channel-blocking agents (Neckers et al., 1986), also reduce expression of TfRs. The availability of Tfbound iron in the extracellular medium also influences expression of TfRs; the addition of microbial iron chelators eg. DFO or a high affinity iron chelating agent such as picolinic acid (Pelosi et al., 1986), stimulates the cells to increase the number of TfRs. On the other hand, if the culture medium is supplemented with additional iron the number of TfRs decrease (Ward et al., 1984).

As mentioned above T-lymphocyte proliferation is controlled by a series of premitotic signals generated by interaction of growth factors and their membrane receptors. In normal T-cells, TfR expression is tightly regulated and linked to prior expression of tissue specific growth factor receptors and their interaction with appropriate growth factors. It has been reported that TfR expression in PHA or PHA/tetradecanoyl phorbol acetate-stimulated cells is dependent on the interaction of IL-2 with its receptor (Neckers and Cossman, 1983; Krönke *et al.*, 1985). A short period of interaction of lectin with lymphocytes allows activation of the TfR gene, even in the absence of IL-2, although production of IL-2 greatly amplified TfR expression (Pelosi-Testa *et al.*, 1988). This effect

is exclusive to IL-2 since addition of other recombinant cytokines did not induce TfR expression. However, others have demonstrated that this depends on the route of activation and have described an IL-2independent pathway for induction of TfR mRNA by stimulating the cells with a combination of phorbol 12, 13 dibutyrate/ionomycin (PDB) (Kumagai, et al., 1988). In this study accumulation of TfR mRNA occurred early in PDB/ionomycin-stimulated T-cells (6 h) in contrast to PHA-stimulated T-cells in which TfR mRNA reached a peak only between 12-24 h. Other studies have been trying to define a temporal sequence of gene activation in human T-cells stimulated with PHA and PMA using nuclear transcription (Krönke et al., 1985). These studies indicated that the proto-oncogene c-myc is activated rapidly, reaching peak transcription within 6-9 h. These findings are consistent with the earlier studies of Kelly et al (1983) who showed that c-myc expression is followed by IL-2R and IFN- γ transcription which peak at 9-15 h. IL-2 gene transcription is also initiated early but does not reach maximal levels until 24 h. Transcription of these four genes was not inhibited by the addition of cycloheximide, indicating that gene expression was not dependent upon the protein products of the other genes. In contrast, the TfR gene was not expressed until 24 h after mitogen activation, and did not reach peak levels until 48 h (Krönke et al., 1985). Addition of cycloheximide at the initiation of culture completely blocked TfR gene transcription, suggesting a dependence on proteins derived from other genes.

Tf is also synthesised in the course of activation by Thelper/inducer lymphocytes. Tf synthesis by these cells appears to be part of a normal autocrine reaction in which growth factors and their respective receptors act as premitotic signals for DNA synthesis that is followed by cell division and proliferation. It has been shown that the Tf gene is expressed at a specific stage during the premitotic events of Tlymphocyte proliferation, by the T4⁺ "inducer" subset of lymphocytes (Lum *et al.*, 1986), and that Tf transcription is an intermediate event in the IL-2 autocrine cycle, and occurs after IL-2 transcription and prior to expression of the IL-2R and TfR. The nature and origin of Tf synthesis by lymphocytes as well as its impact on the autocrine pathway of Tlymphocyte proliferation await further study.

1.2.4 Effect of iron deficiency and iron overload on lymphocyte function

Both iron deficiency and iron excess have been shown to alter cellular immune and non-specific defence functions. It has been demonstrated that when the iron saturation of Tf is very low, the rate of proliferation of mitogen stimulated lymphocytes is decreased (Brock, 1981; Phillips and Azari, 1975). It has also been shown that supplementation of lymphocyte cultures with serum from iron-deficient mice does not permit optimal proliferation due to the reduced Tf saturation (Mainou-Fowler and Brock, 1985). Furthermore, It has been demonstrated that splenic lymphocytes as well as purified T- and B-cells from iron deficient mice showed a depressed transformation response to mitogens (Kuvibidila et al., 1983) and decreased cytolytic activity (Baliga et al., 1981). The effect on cellular immune function of excess iron *in vitro* has been studied by a number of investigators. Bryan *et al* (1981) demonstrated that high concentrations of ferric citrate could inhibit a human mixed lymphocyte reaction. Iron was shown to act on the responder cells rather that the stimulator cells. Keown and Descamps-Latscha (1983) reported that excess iron, in a variety of different forms, could inhibit antigen-induced lymphocyte proliferation and cytotoxic T-lymphocyte (CTL) sensitization, but not CTL effector function. In their studies they used varying concentrations of ferric iron as ferric citrate, ferric nitrate, and also ferric chloride at pH 7.2 where solubility may be a problem. It has also been reported that lymphocytes from iron overloaded rats showed a marked reduction of proliferative capacity after a mitogenic stimulus and a dramatic decrease of their capacity to repair DNA damage (Pietrangelo *et al.*, 1988).

Clinical and experimental iron-overload are associated with a high incidence of infection (Weinberg, 1978; 1984) and neoplasia (Powell et 1971). One possible reason for this association is that ironal., overload may interfere with normal immune surveillance. In clinical iron overload a small but significant amount of iron is 'free' or non-Tf bound. It has been shown that in iron overload, non-Tf bound iron can be present (Hershko et al., 1978; Batey et al., 1980). It has been demonstrated that non-Tf bound iron Fe³⁺ has an inhibitory effect on the generation of cytotoxic T-lymphocytes by affecting all the major Tlymphocyte subsets. Good *et al* (1986) have found that Fe^{3+} reduced both the cloning efficiency of human CD4⁺ precursor lymphocytes and the rate of clone growth of the T-cells that did proliferate. The same group has found that low levels of exogenous Fe³⁺ similar to those concentrations reported in the non-Tf bound fraction of the serum of patients with iron-overload, significantly reduced the generation of CTL

by an effect on both regulatory lymphocytes and CTL-precursors.

Spleen cells from iron-overloaded animals had also markedly reduced ability to generate allo-specific CTL, though a frequency analysis of CTL-precursors revealed that these same animals had normal numbers of allo-specific CTL-precursors, suggesting that a regulatory cell defect was responsible for the reduced ability to generate effector cells in the mixed lymphocyte culture (Good *et al.*, 1987). Ironoverload can thus place a limit on the number of available T-helper cells *in vivo* apparently by a reduction in the number of functional helper Tcells. Bryan *et al* (1986) found that iron (Fe³⁺) at extremely high concentrations suppressed expression of the human T-cell CD4 molecules, on mitogen activated cells. They showed that this was a specific defect, since iron did not alter expression of other markers on activated cells, including CD8, Ia and the thermostable sheep red blood cell receptor.
AIMS OF THESIS

Although iron is needed by lymphocytes for their proper function, the way that iron affects these cells is not fully understood. One particular area in which there is a lack of knowledge is how these cells respond to different forms and levels of iron, and how this affects the way they handle iron. Therefore, the work presented in this thesis has attempted to investigate uptake of iron by mouse lymph node cells donated in different forms and amounts, and relate it to its effect on proliferation. It has also examined where this iron is located inside the cell by looking at the relative incorporation of the metal into different intracellular compartments. One of the reasons for the conflicting results reported on the effect of non-Tf bound iron on proliferation of lymphocytes could be that these cells have the ability to synthesise transferrin. Therefore the possibility that mouse lymph node cells make Tf was also investigated.

Similar studies have also been performed on the effect of iron on human cells in order to see whether these two species respond differently to different conditions of iron. This has also included an investigation of the effect of Lf on lymphocyte proliferation. In particular the role of Lf in promoting proliferation of human lymphocytes by sequestering excess iron has been investigated. Numerous studies have demonstrated a regulatory role in the immune response for Lf, but there are conflicting reports concerning whether this protein fulfills a similar role to Tf. As mentioned in the preceding review it is not clear which human T-cell subsets are primarily affected by high levels of iron. It was therefore decided to study the effect of different forms and amounts of iron on the expression of T-cell markers to investigate whether different T-cell subsets respond in different ways to iron in varying forms and amounts.

Finally, because it is often stated in the literature that transformed cells generally have reduced needs for nutrients and growth factors compared with their normal counterparts, a similar study was carried out on the T-transformed line CCRF-CEM to examine whether iron was required in similar amounts and forms.

CHRPTER TWO

THE EFFECT OF DIFFEHENT LEVELS AND FORMS OF IRON ON MURINE LYMPHOCYTES

2.1 INTRODUCTION

An immune response is a complex process, involving activation and proliferation of lymphocytes, and a complex series of cellular interactions between T- and B-lymphocytes and monocytes, to produce antibodies and to recruit and activate phagocytic and cytotoxic cells. Although direct interactions are probably involved in such responses, soluble factors also play an important role. In addition, other types of factors have been shown to interact with these cells. One such group of factors includes iron and the molecules that bind it, namely Tf, Lf, and Ft. The ability of lymphocytes to proliferate is closely linked with iron, and both inadequate and excessive amounts of the metal can be deleterious (Dallman, 1987; Good *et al.*, 1988). Iron is an important requirement for these metabolically active cells which undergo differentiation and cell division, it being needed in particular for the synthesis of iron containing proteins and enzymes involved in DNA synthesis.

Tf-bound iron appears to be the form in which iron must be supplied (Brock, 1981; Brock and Mainou-Fowler, 1983; Taylor *et al.*, 1987). It has been demonstrated that the ability of Tf to promote lymphocyte proliferation is closely related to its ability to donate iron to the cells (Brock *et al.*, 1986). Little proliferation occurs when apo-Tf is added. Optimal transformation occurs in the presence of Tf which is between 30 and 70% saturated (Brock, 1981). There have been conflicting reports on whether addition of iron in forms other than Tf leads to inhibition or enhancement of the lymphocyte response to mitogens. While some

reported that chelate iron cannot support proliferation (Phillips and Azari, 1975; Brock, 1981; Taylor et al., 1987) others have reported good proliferation (Tanno and Takishima, 1982). Therefore to investigate this matter in more detail, the effect on mouse lymphocyte proliferation of Tf together with two chelators, different in their chemical properties has been investigated. One is a low affinity hydrophilic chelator, nitrilotriacetate, which maintains iron in a soluble form, unlike iron salts, as it is less susceptible to hydrolysis (Bates and Schlabach, 1973). The second chelator, pyridoxal isonicotinoyl hydrazone, is lipophilic and unlike many other chelators was found to be able to donate iron to erythroid precursors for use in haem synthesis (Ponka et al., 1982). Mouse lymph node lymphocytes were chosen as an in vitro assay system because they are easily obtainable, and the use of cells from an animal model allows variables which could complicate the interpretation of the results to be eliminated as mice are inbred and age sex etc. are much easier to control. Thus they respond more uniformly than humans.

Much of the knowledge of incorporation of iron into different intracellular iron compartments is derived from studies of erythroid cells. As far as lymphocytes are concerned this area of iron metabolism is almost unexplored. The few studies which looked at this field in cells of the lymphomyeloid series (Bomford *et al.*, 1986; Mattia *et al.*, 1986) have demonstrated that the uptake of iron by the cell is initially to the non-Ft compartment, and this is followed by a time-dependent fractional accumulation of iron into Ft which occurs gradually and increases over a 2-5 h period. However in these studies they cultured the cells for a very short time and looked at cellular distribution of iron at an early time after iron administration, and there are no data concerning intracellular distribution of iron in cultures over a longer period up till now. Moreover, in the studies mentioned above, FeTf was the exclusive source of iron. Therefore, the work reported in this chapter has attempted to define more closely the effect of different iron forms on lymphocyte transformation *in vitro* in relation to their iron donating ability. This involved the study of the ability of proliferating lymphocytes to acquire iron from Tf and the two iron chelates, i.e FeNTA and FePIH; to investigate the action of these chelating compounds and the effect of their iron content on cellular proliferation, and to see whether iron taken up from these chelates is handled differently from iron acquired from Tf. By looking at the pattern of iron and Ft accumulation and intracellular pathways of iron metabolism, a clearer picture of the effect of these iron carriers on lymphocyte function could be drawn.

One of the reasons for the conflicting results discussed above (section 1.2.3) concerning the ability of non Tf-bound iron to promote lymphocyte transformation, may be explained by the fact that lymphocytes can themselves synthesise Tf which then mediates proliferation. In man activated CD4⁺ T-cells synthesise Tf (Lum *et al.*, 1986), but in mouse and rat there is so far no convincing evidence of Tf synthesis by lymphocytes. Therefore, Tf synthesis in the mouse lymph node cells upon activation was investigated to confirm that mouse lymphocytes can synthesise Tf as suggested by previous work with human cells (Soltys and Brody, 1970; Nishiya *et al.*, 1980; Broxmeyer *et al.*, 1983; Lum *et al.*, 1986).

In summary therefore, the work reported in this chapter aims to investigate:

- 1) Uptake of iron by the mouse lymphocyte donated in different forms and amounts in relation to its effect on transformation.
- 2) Incorporation of iron into different intracellular compartments.
- 3) The ability of mouse lymph node cells to make Tf.

2.2 MATERIALS

2.2.1 Animals

Balb/C mice of either sex of 16-40 weeks of age were used throughout, and obtained from departmental animal house stock.

2.2.2 Reagents

When necessary, media were prepared in iron-free apparatus i.e. plastic or glassware rendered iron-free by acid washing in 10% HCl overnight then rinsed in deionised distilled water three times.

2.2.2.1 Phosphate Buffered Saline (PBS)

This consisted of NaCl (8 g/l), KCl (0.2 g/l), Na₂HPO₄ (1.15 g/l) and KH₂PO₄ (0.2 g/l) (Analar grade, BDH Chemicals Ltd, Poole, Dorset, England). Solutions were made in deionised distilled water, and when required, aliquots were sterilised by filtration. PBS was made low iron by adding 1% (w/v) NaHCO₃.

2.2.2.2 Human Serum Albumin (HSA) solution

A solution of 1% (w/v) HSA (Behringwerke, West Germany) was prepared in RPMI-1640 medium (w/o Hepes; Flow Laboratories, Irvine, Ayrshire, Scotland), was sterilised by filtration and stored at 4° C. The solution was tested by radial immunodiffusion assay (kindly done by Dr S. McGregor) and found not to contain any detectable Tf.

2.2.2.3 Concanavalin A (Con A) solution

A stock solution of Con A (10 μ g/ml; Sigma, Dorset, England) was prepared in low iron PBS, sterilised by filtration, dispensed in 100 μ l aliquots, and stored at -20° C. Each aliquot was used once only.

2.2.2.4 Ferric nitrilotriacetate (FeNTA)

This was prepared using a molar ratio of 4:1 of NTA to iron to ensure the formation of low molecular weight rather than hydrated polymeric complexes (Spiro *et al.*, 1967). One volume of a freshly prepared solution of FeCl₃ (20 mM, BDH) was added dropwise, with constant stirring, to an equal volume of NaNTA (80 mM, pH 7.0; Sigma) and the pH adjusted to 5 with 1 M NaOH to prevent the NTA-free acid precipitating. The resulting concentration of FeNTA was 10 mM (pH 5.0)..

2.2.2.5 Ferric pyridoxal isonicotinoyl hydrazone (FePIH)

A stock solution of 0.3 mg/ml of PIH (Dr P. Ponka, McGill University, Canada) was made in 5-10 ml deionised water, 50 μ l of 1 N HCl was added, and the solution was mixed vigorously and left under the hot tap to aid dissolution of the chelator before being filter sterilized.

2.2.2.6 Standard Culture Medium

This was prepared by supplementing RPMI-1640 culture medium (w/o Hepes; Flow) with 0.3 mg/ml L-glutamine (BDH), and 100 IU/ml penicillin and 100 μ g/ml streptomycin (Flow). Finally, 10% final concentration of fetal calf serum (FCS) (Flow) was added. For serum

free medium, HSA (1 mg/ml final concentration), 2-mercaptoethanol (2-Me, final concentration 50 μ M; BDH), and human Tf (Behringwerke) previously prepared in PBS containing 1% (w/v) NaHCO₃ (Analar grade, BDH) to ensure iron removal, usually 50 μ g/ml, were added instead of FCS. Different Tf saturations with iron were achieved by adding to Tf solutions appropriate amounts of FeNTA, which is known to maintain iron in the solubilised form and allows it to be readily available for binding to Tf (Bates and Schlabach, 1973).

2.2.2.7 Reagents for α-naphthyl acetate esterase staining

2.2.2.7.1 Fixative. This consisted of formol calcium solution, pH 6.7, containing 10% (v/v) formaldehyde (BDH) and 1% (w/v) CaCl₂ in distilled H₂O.

2.2.2.7.2 Phosphate buffer (0.06 M, pH 5.0). This was prepared by mixing 98.5 ml of KH_2PO_4 (9.08 g/l) and 1.5 ml of Na₂HPO₄.2H₂O (11.88 g/l). The buffer was sterilised by autoclaving.

2.2.2.7.3 Hexazotized pararosaniline. This solution was prepared by mixing equal volumes of two solutions which were prepared as follows:

•Solution A This was a freshly prepared solution of 4% (w/v) NaNO₂ (BDH).

•Solution B This was prepared by dissolving 1 g of Pararosaniline (Waxes & General Laboratories Supplies, London, England) in 20 ml distilled H_2O to which 5 ml of concentrated HCl was added. The solution was gently warmed to dissolve the pararosaniline, allowed to stand at room temperature to cool, and finally filtered. The filtrate was then stored in the dark at 4° C.

2.2.2.7.4 α -naphthyl acetate esterase stain. This consisted of adding 2.4 ml of hexazotized pararosaniline and 10 mg of a freshly prepared solution of α -naphthyl acetate (Sigma) in 0.4 ml acetone to 40 ml of the phosphate buffer. Finally the pH of the complete reagent was raised to 5.8 with 2N NaOH before being filtered.

2.2.2.8 Reagents for Poly Acrylamide Gel Electrophoresis for ferritin

2.2.2.8.1 Acrylamide solution (40%)

38.96 g Acrylamide (BDH)

1.04 g Bisacrylamide (BDH)

Made up to 100 ml with H_2O , filter sterilized, and kept in cold.

2.2.2.8.2. Sample buffer

50% (v/v) Glycerol (BDH)

0.25 M Tris-HCl (pH 6.8; Bohringer Mannheim GmbH, W.Germany)

0.0125% (w/v) Bromophenol blue (BDH)

2.2.2.8.3 Tank buffer

24 g Tris (Bohringer)

115.2 g Glycine (BDH)

Made up to 4 litre with H_2O

2.2.2.8.4 Fixer

10% (v/v) TCA (BDH) 10% (v/v) Acetic acid (BDH) 30% (v/v) Methanol (BDH)

2.2.2.8.5 Protein stain

0.25% (w/v) Coomassie blue

50% (v/v) Methanol

10% (v/v) Acetic acid

The solution was stirred overnight and filtered before use.

2.2.2.8.6 Perl's Prussian blue reagent

8% (w/v) Ferricyanide mixed 1:1 with 8% (v/v) concentrated HCl (both freshly prepared in double distilled H_2O).

2.2.2.8.7 Destaining solution

20% (v/v) Methanol 10% (v/v)Acetic acid in H_2O

2.2.2.9 Reagent for immunoprecipitation of transferrin

2.2.2.9.1 3D Buffer

1% (v/v) Triton X-100 (BDH)
0.5% (w/v) deoxycholic acid (BDH)
0.1% (w/v) sodium dodecyl sulphate (SDS; BDH)
0.1 M NaCl
10 mM Na-phosphate buffer, pH 7.5

The buffer solution was sterilized by filtration and kept at room temperature.

2.2.2.10 Reagents for SDS-Poly acrylamide gel electrophoresis for transferrin

2.2.2.10.1 Urea buffer

10% (w/v) SDS

8 M Urea (BDH)

10 mM Tris-HCl, pH 7.0

The buffer solution was sterilized by autoclaving.

2.2.2.10.2 Sample buffer

50% (v/v) glycerol
0.25 M Tris-HCl, pH 6.8
5% (w/v) SDS
5% (v/v) β-mercaptoethanol
0.0125% (w/v) bromophenol blue

2.2.2.10.3 Tank buffer

24 g Tris 115.2 g glycine

4 g SDS

Made up to 4 litre with H_2O .

2.3 METHODS

2.3.1 Preparation of cell suspension and culture conditions

Cell suspensions were prepared from peripheral (brachial, axillary, and inguinal) and mesenteric lymph nodes of the mice. The lymph nodes were removed from freshly killed mice under sterile conditions, and the cells were gently teased out into RPMI-1640 supplemented with streptomycin, and L-glutamine as described in section penicillin, 2.2.2.6. The cell suspension was mixed well, transferred into a sterile plastic universal and left to stand for 1-2 min for any tissue debris to settle out. The cell suspension was then carefully aspirated into a clean plastic universal and the cells were washed twice at 160 g for 2-3 min in the culture medium. The suspension contained 85-95% viable cells as measured by eosin exclusion. The cells were finally suspended in the culture medium at a concentration of 2×10^6 viable cells/ml. Unless otherwise stated cells were cultured in conical test tubes with Con A at the appropriate concentration. The cells were cultured at 37° C in an atmosphere of 5% CO_2 , 95% air for 48 h.

2.3.2 Measurement of proliferation

Cell suspensions were prepared as described above. Proliferation responses were assayed by plating 100 μ l aliquots of the cells in a conical bottomed microtitre culture plate (Sterilin) and pulsing with 1 μ Ci/well of ³H-thymidine (specific activity 52 Ci/mmol; Radiochemical Centre,

Amersham, England) for 4 h before terminating the incubation period. The cells were then harvested on glass fibre mats using a cell harvester (Skatron, Lierbyen, Norway), which were usually left overnight on the bench to dry. Individual discs, each with contents of a single well, were added to 2 ml of scintillation fluid (LKB, Croydon, Surrey, England) and counted for 5 min on a scintillation counter (Packard, Pangbourne, Berkshire, England).

2.3.3 Uptake of iron by proliferating mouse lymphocytes cultured with different transferrin saturations

Uptake of iron was determined by incubating the lymph node cells which were prepared as described in section 2.3.1 with Con A (1 μ g/ml) in serum free medium and transferrin at 50 μ g/ml saturated to various degrees with ⁵⁹Fe. This latter was prepared by diluting ⁵⁹Fe-citrate (specific activity 11 μ Ci/µg Fe; Amersham) with FeNTA as a source of cold iron. Sufficient iron was added to apo-Tf to give the required saturation at least 2 h before the ferric Tf solutions were added to cultures (Zapolski and Princiotto, 1977), taking into account endogenous iron present in the medium which is in the average of 5 ng/ml (Brock, 1981). After incubation for 2 days in iron-free plastic conical test tubes, the cells were spun down at 1200 rpm for 3 min then washed 3 times with Hanks's solution. To avoid problems of adsorption of isotope to culture tubes, cells were transferred to new tubes after suspension in the third wash solution. Radioactivity associated with supernatant, washes, tubes, and the cells was counted in a gamma counter (Compugamma 1282, LKB Wallac, Croydon, Surrey, England). To determine the degree of transformation in different samples, 0.5 ml was removed from each cell suspension, and then split into 4 x 100 μ l aliquots in a 96 well conical bottomed plate. The cells were pulsed for 4 h with 1 μ Ci/well of ³H-thymidine, harvested, and counted for 5 min. The radioactive iron was tested and found not to interfere with measurement of uptake of labelled thymidine.

2.3.4 Isolation of ferritin

This was prepared from 100 g of mouse livers following the method reported by Linder and Munro (1972) for preparation of human Ft, except that AcA-22 (LKB, Bromma, Sweden) was used for the gel filtration step. The tissue was minced with scissors and a homogenate was made in four volumes of water using an MSE homogenizer (Crawley, All materials were kept in an ice bath. Sussex, England). The homogenate was heated to 70° C for 10 mins then cooled on ice and centrifuged for 15 min at 2000 g to remove heat-inactivated proteins. The pH of the supernatant was adjusted to 4.8 with glacial acetic acid, left for 4 h, and a further precipitate removed by centrifugation. The pH was adjusted to 7.2 with 1 M K₂HPO₄, solid ammonium sulphate was added to give 50% saturation and the solution was left overnight at 4° C. The precipitate was removed by centrifugation at 2000 g, dissolved in the minimum amount of PBS, and dialysed against 6 changes of the same buffer until the reaction for sulphate with 1% BaCl₂ was negative. The sample was fractionated by gel filtration on AcA 22 in a column $(1.2 \times 90 \text{ cm})$ eluted with PBS, and 1.5 ml fractions collected. The tubes with red colour, indicative of Ft, were pooled and centrifuged at

 $11 \ge 10^4$ g for 4 h in an L2-65 Beckman centrifuge to sediment the Ft. After redissolving, protein content of the ferritin preparations was measured by the method of Lowry *et al* (1951) using BSA (Sigma) as a standard. Purity was checked by polyacrylamide gel electrophoresis as described in the next section. Mouse spleen Ft was prepared following similar procedure and was kindly provided by Dr X. Alvarez-Hernandez.

2.3.4.1 Agarose Polyacrylamide gel electrophoresis For ferritin

The LKB 2001 Vertical Electrophoresis system was used with a non-denaturing gel consisting of 1% agarose and 4% acrylamide gel, and a 1.5 mm spacer. The agarose solution was prepared by dissolving 345 mg of agarose (Sigma) in 4.37 ml of 1.5 M Tris-HCl, pH 8.8, and 8.5 ml H₂O, and brought to the boil on a hot plate stirrer. When dissolved, the agarose solution was put in a water bath at 50° C. The acrylamide solution was prepared by mixing 4.37 ml of 40% acrylamide (see section 2.2.2.8.1) with 4.37 ml 1.5 M Tris-HCl, pH 8.8, and 8.5 ml H₂O, and put in the same water bath. When both solutions were equilibrated to temperature, they were mixed rapidly and 450 µl of ammonium persulphate (BDH) (freshly prepared) and 25 µl TEMED (Sigma) were added immediately. The mixture was immediately poured into a prewarmed mould with the comb already inserted in it, using a prewarmed glass pipette. The comb was extracted very carefully after the gel had solidified, and the resultant wells were then filled with tank buffer (see section 2.2.2.8.3). The upper reservoir was placed on top of the gel and sufficient tank buffer was added to cover the electrode wire.

The tank was filled with the remaining buffer, and the upper reservoir and gel were placed in the tank. The samples were prepared by adding the Ft solution to the sample buffer (see section 2.2.2.8.2). Two equal volumes of the samples were loaded into two separate wells each containing 20 µg of the protein using long narrow round pipette tips (Bioquone Ltd, Yorkshire, England). The gel was run at 40 mA for approximately 6 h with water cooling. The gel was then incubated for 1 h in fixer solution (see section 2.2.2.8.4), and then cut into two halves each containing one sample, one being stained for protein by incubation in Coomassie blue solution (see section 2.2.2.8.5) for 1 h, and the other for iron by incubation in Perl's Prussian blue reagent (see section 2.2.2.8.6) for 1 h. The gel stained for protein was incubated in the destaining solution (see section 2.2.2.8.7) overnight on a rotary shaker. Both gels were dried by freeze drying using a dryer gel plate connected to a freeze dryer machine. One major and two minor bands were observed all of which stained for iron as well as protein.

2.3.5 Preparation of antibodies against ferritin

The inoculum was prepared by emulsifying complete Freund's adjuvant (Difco) and a solution of MLFt (100 μ g/ml) (1:1) to give a final volume of 1 ml in a syringe of the same capacity. To make a stable emulsion the syringe was vibrated on a vortex for 20 min with the plunger and needle secured with tape. The stability of the emulsion was tested by allowing a small drop to fall into a beaker of water, and the emulsion was judged ready to be injected when the drop did not diffuse in the water. The scheme of immunization of rabbits consisted of multiple

subcutanous inoculations in various sites (1 ml vaccine per rabbit) followed by 10 days rest, and then a similar repeat inoculation. After another 10 days a booster was given (50 µg of Ft in 1 ml PBS). After a week, approximately 20 ml blood was collected from the ear vein of each rabbit into a glass universal and allowed to clot at room temperature. The serum was then pooled, transferred to plastic conical-bottomed test-tubes (Sterilin) and centrifuged twice at 3000 rpm for 5 min to remove any erythrocytes. The antibody titre was checked by the ring precipitin test as follows; a small volume of Ft solution (10 μ g/ml, 100 μ g/ml, and 1 mg/ml) was gently layered on top of a similar volume of antiserum in a Durham tube. The formation of strong precipitin bands with all three concentrations of the protein indicated a very good titre. The antisera were salt fractionated with 33% (w/v) (NH₄)₂SO₄ (3 times) (by adding saturated $(NH_4)_2SO_4$ solution, 50% of the original antisera volume). The IgG fraction was dialysed against PBS to remove $(NH_4)_2SO_4$ and stored in aliquots at -20° C.

2.3.6 Affinity chromatography

The following Sepharose 4B affinity columns; Sepharose-normal rabbit immunoglobulins (S-NRIg), Sepharose-rabbit- α -mouse liver Ft (S-R α -MLFt), Sepharose-normal sheep immunoglobulins (S-NSIg) (NSIg was purchased from Serotec), Sepharose-sheep α -human transferrin (S-S α -HTf) (S α -HTf was purchased from Serotec), together with Sepharose-MLFt (for labelling α -MLFt with ¹²⁵I; see section 2.3.10), were prepared using similar procedures. The required amount of freeze dried CNBr-activated Sepharose 4B (Pharmacia Ltd, Milton Keynes, England) was swollen in 1 mM HCl for 15 min then washed with the same solution on a sintered glass filter (200 ml/g of gel). The

gel was finally washed and resuspended in the coupling buffer (0.1 M Na₂CO₃, pH 7.0 containing 0.5 M NaCl). The proteins to be conjugated were previously dialysed against the coupling buffer for 2 days, the concentration then adjusted to 5-10 mg/ml, as judged by spectroscopic absorption at 280 nm, and finally they were mixed with gel in a ratio of 2:1 (v/v). The mixture was left rotating in an end-over-end mixer for 2 hat room temperature, after which the supernatant was aspirated. Excess ligand was washed away with coupling buffer on a glass filter and any remaining active groups blocked by treatment with 2 volumes of 1 M glycine for 2 h at room temperature. The final product was washed alternately with 3 cycles of 0.1 M sodium acetate buffer (pH 4.0) containing 0.5 M NaCl and the coupling buffer to remove traces of noncovalently adsorbed materials. The conjugates were stored at 4° C in PBS containing 1% (w/v) BSA and 0.02% (w/v) sodium azide. The binding capacity of S-R α -MSFt and S-S α -HTf were measured by counting the radioactivity of a known concentration of ¹²⁵I-Ft and ¹²⁵I-Tf solution before and after passing each one through a column (1 ml syringe barrel) containing 50 μ l of the appropriate packed coupled gel. The capacity of the affinity gel was found to be 5 and 7 μ g/ μ l of settled gel for 2 batches of the S-R α -MLFt, 4.2 for S-R α -MSFt and 5.4 μ g/ μ l of settled gel for S-Sα-HTf.

2.3.7 Iron uptake and intracellular distribution of iron in proliferating mouse lymphocytes cultured with different carriers

Intracellular distribution of iron was investigated in proliferating lymphocytes that had been cultured with Con A (1 μ g/ml) in serum-free

medium containing different amounts of iron in different forms. Three forms of iron were investigated, ferric transferrin, ferric nitrilotriacetate, and ferric pyridoxal isonicotinoyl hydrazone. The transferrin at 50 μ g/ml was used at two different saturations with iron, one low (14%) and the other high (71%). The other iron binding agents were used with iron content equivalent to that of the corresponding ironcontaining Tf (10 and 50 ng/ml respectively). The following protocol was adopted:

FeNTA (1 mM) was mixed volume to volume with ⁵⁹Fe-citrate (specific activity 1331 MBq/mg Fe) and this trace labelled FeNTA was either used as such or added to Tf or PIH. The Fe:chelator ratio was kept constant throughout all the experiments (1:5 for FeNTA and 1:2 for FePIH). The cells (2 x 10⁶ viable cells/ml) were cultured in plastic conical test tubes for 48 h, after which they were washed 3 times with Hank's solution, then lysed by adding 1 ml of 1% Triton X-100 in PBS containing 1 mM desferrioxamine (DFO) (CIBA Laboratories, Horsham, England) to the pellet of cells. The addition of DFO at this stage, which reacts with iron to form a highly stable complex, was intended to chelate any available iron at the moment of cell lysis, hence reducing to minimum potential movement of labile iron to other pools as a consequence of cell disruption. The lysates were mixed vigorously and left for 5 min before being centrifuged in a microcentrifuge for 10 min at 10000 rpm, and the precipitates, which consist of insoluble cellular components, after a further washing with Triton X-100/PBS were kept for counting radioactivity. The wash supernatants were added to the original Triton X-100/PBS lysate supernatants. These two together were passed through affinity columns. The affinity columns were made

in a 1 ml syringe barrel containing 50-75 µl of each immunoabsorbent gel. Fig. 2 shows a diagram of one column. The column contains two specific immunoabsorbents (S-R α -MLFt and S-S α -HTf) together with two gels coupled to the immunoglobulins of the corresponding normal sera to correct for non specific binding, i.e. S-NRIg and S-NSIg for S-R α -MLFt and S-S α -HTf respectively (see section 2.3.6). The columns were equilibrated and eluted with Triton X-100/PBS, then the lysate supernatants were passed twice through each column to wash out liquid in The different parts of the column were carefully the dead space. dismantled and radioactivity associated with each gel counted in a gamma counter. The filtrates from the immune absorbents were kept for further fractionation into low (<10 Kd) and high (>10 Kd) molecular weight materials using Amicon ultrafiltration cells (Centricon 10 Microconcentrator; Amicon Co, Lexington, Mass., USA) equipped with PM 10 membranes. The Amicon cells were centrifuged in a fixed angle rotor centrifuge (Super Minor, Mk2, MSE) at 1250 g for 90 min. The retentates which consist of the high molecular weight fraction consist of ⁵⁹Fe bound to soluble proteins (other than Ft or Tf). This fraction was called the intermediate molecular weight fraction-⁵⁹Fe and it probably consists mainly of iron in enzymes and haemprotein. The filtrates containing low molecular weight iron compounds consist of iron bound to DFO.

2.3.8 α-naphthyl acetate esterase (ANAE) staining

A cell suspension of lymph node cells was prepared as described in section 2.3.1. Approximately 6-7 x 10^5 in vivo stimulated lymph node



Figure 2 Diagram of an affinity column used in fractionating lysate supernatants for intracellular distribution of iron

cells or lymphocyte depleted cells (see section 2.3.9) were deposited on clean glass slides in a Shandon cytocentrifuge at 1000 rpm for 90 sec and then air dried. Preparations were then fixed in formol calcium (see section 2.2.2.7.1) at 4° C for 10 min and washed in running tap water for 20 min at room temperature. Finally the slides were incubated with the staining reagent (see section 2.2.2.7.4) for 90 min at 37° C. The slides were then washed gently in running water for 10 min, left to dry at room temperature, and then counterstained with 2% methyl green (Hopkins & Williams, Essex, England) for 20 sec. Finally the cells were rinsed in running water, completely air-dried, and mounted with Histomount (National Diagnostics, Somerville, New Jersey, USA). Slides were examined using a Leitz-Welzlar microscope (X 100). T-lymphocytes show small redish brown spots in their cytoplasm, and monocytes and macrophages are characterized by diffuse redish brown staining material throughout the whole cell, while B lymphocyte and null cells do not stain.

2.3.9 Immunoprecipitation method For transferrin

Tf synthesis was assayed in three systems:

1) Normal lymph node cells: Lymph node cells were obtained from six mice as described in section 2.3.1. The cells were preincubated in 5 ml of Tf-free medium consisting of RPMI-1640 (w/o cysteine or cystine; Select-Amine Kit, Gibco Ltd, Paisley, Scotland) in a conical test tube for 1 h at 37° C in an atmosphere of 5% CO₂, 95% air, to allow exocytosis of any endogenous Tf to occur. Cells were then washed twice with PBS, resuspended in 2 ml of RPMI-1640 (w/o cys) containing 100 μ Ci of ³⁵S-cysteine (specific activity 1021.8 Ci/mmol; DuPont NEN,

•.

Dreireich, W. Germany), and incubated at 37° C in a CO_2 incubator for 5 h. The cells and supernatant were separated, and 9.4 µl of phenylmethyl sulphonyl fluoride (PMSF; 0.21 M in dimethylformamide) and 3ml 3D buffer (see section 2.2.2.9.1) containing 14.4 µl of PMSF were added to supernatant and pellet of cells respectively to give a final concentration of 1 mM. Both supernatant and cells were kept at -70° C until ready for use.

2) Con A stimulated lymph node cells: The cells were cultured at $2 \ge 10^6$ cells/ml in RPMI-1640 containing 10% FCS and $4 \ge 10^6$ ml con A for 36 h. The cells were then harvested and treated as described above for the non-stimulated cells.

3) In vivo stimulated lymph node cells: Six mice were inoculated into the footpad with 50 μ l of an emulsion of PBS/complete Freund's adjuvant (1:1). Three weeks later, the mice were sacrified and the stimulated popliteal nodes were removed. Cells were prepared and then treated as described above for the non-stimulated cells.

Detection of Tf synthesis in adherent and non-adherent *in vivo* stimulated lymph node cells was also performed. After preparing the cell suspension, the *in vivo*-stimulated lymph node cells were resuspended at a concentration of 5 x 10⁶ cells/ml in RPMI-1640 complete medium (section 2.2.2.6) containing 10% FCS. The cell suspension was placed in a Falcon 100 x 5 mm plastic Petri dish (5 ml/dish), and incubated for 1 h at 37° C in a CO₂ incubator. For adherent cells, 2 ml of RPMI-1640 (w/o cys) containing ³⁵S-cysteine was added receiving a total of 100 μ Ci. Non-adherent cells were treated as described above.

Another methodological approach was used to investigate Tf synthesis in lymph node macrophages. This consists of performing an immunodepletion of B and T-lymphocytes before doing the assay. The lymphocytes were eliminated by incubating the in vivo stimulated lymph node cells (10⁷ cells/ml) with a cocktail of 1:1000 diluted α -mouse IgG (Serotec, Oxford, England), a 1:2000 diluted α -Thy 1 (Serotec), and a 1:10 diluted low-Tox-M rabbit serum complement (Cederlane Laboratories Ltd, Ontario, Canada) for 1 h twice. At the end of the incubation viability was determined using eosin exclusion and was found to be 27-34%. The ANAE stain revealed that the proportion of macrophages increased from 15% in non-depleted in vivo-stimulated lymph node cells to 54% in lymphocyte depleted cells. The cells were washed 3 X and counted before being endogenously labelled with ³⁵S-cysteine as described above. Non-depleted in vivo stimulated mouse lymph node cells were also assayed as control. The same number of viable cells of both group of cells (depleted and non-depleted) was assayed.

After incubating the cells with 35 S-cysteine, the supernatant was split into three 500 µl aliquots in 1.5 ml microvials. Each aliquot was treated as follows:

- Test: To this sample 12 µl of rabbit α-mouse Tf (provided by Dr J.H. Brock) was added.
- Competition: To this sample 40 µg of mouse Tf (provided by Dr J.H. Brock) was added followed by 12 µl of rabbit α-mouse Tf.
- Control: To this sample 8 µl of normal rabbit serum was added.

The samples were incubated for 1 h at room temperature. IgGsorb (The Enzyme Center, Malden, Mass., USA) (100 μ l of 60 mg/ml in 3D buffer, freshly prepared each time), protein A coupled to an insoluble base, was then added to all tubes. The tubes were left incubating on an end-over-end rotator at room temperature for 1 h. The IgGsorb was spun down at 6500 rpm on a microfuge for 15 sec. The supernatants were discarded and the IgGsorb was washed 3 times with 750 μ l of 3D buffer for 15 sec at 6500 rpm on the microfuge, ensuring that IgGsorb was well suspended between each wash. The washed IgGsorb was resuspended in 70 μ l of urea buffer, and incubated at room temperature on an end-over-end rotator for 1 h. The tubes were then spun at full speed (13000 rpm) on the microfuge for 5 min. The pellets were discarded and the supernatants were kept.for an SDS polyacrylamide gel electrophoresis analysis.

2.3.9.1 SDS Polyacrylamide Gel Electrophoresis for transferrin

The standards for the electrophoresis analysis were prepared in separate microvials and 40 μ l of urea buffer (see section 2.2.2.10.1) added to each one. The high molecular weight standard mixture (Sigma) (20 μ l) was used alongside unlabelled mouse Tf (10 μ g) and ¹²⁵I-labelled human Tf (1000-3000 cpm) (kindly supplied by Miss N. Moughal) to assess the position of the Tf band on the gel and the autoradiograph respectively. To all tubes; samples, and standards, 17 μ l of sample buffer (see section 2.2.2.10.2) was added and mixed well. The cap of each tube was pierced with broad-gauge syringe needle and the tubes

heated in boiling water for 5 min

The LKB 2001 Vertical Electrophoresis System was used with 10% agarose gel and 1.5 mm spacers.

The running gel was prepared as follows:

8.75 ml 40% acrylamide (see section 2.2.2.8.1)
8.75 ml 1.5 M Tris-HCl, pH 8.8
16.3 ml H₂O
0.7 ml 10% (w/v) SDS

After mixing well, 450 μ l of 10% (w/v) ammonium persulphate (freshly prepared) and 25 μ l TEMED were added, mixed by inversion, and the gel solution pipetted into the gel former to a mark made previously at ~1 cm down from the bottom end of the comb. A few ml of water-saturated butanol were added to remove bubbles and ensure a straight surface. When the gel had solidified, the butanol was tipped out and the remaining space washed twice with water. Excess water was removed with a tissue. The stacking gel was prepared as follows:

1 ml 40% acrylamide (see section 2.2.2.8.1) 1.25 ml 0.5 M Tris-HCl, pH 6.8 7.5 ml H₂O 100 μl 10% (w/v) SDS

After mixing 100 μ l ammonium persulphate and 10 μ l TEMED were added and after inserting the comb, the stacking gel was poured. The comb was extracted very carefully after the gel had solidified. Equal volumes of each sample (~120 μ l) were loaded by making up the volume with tank buffer (see section 2.2.2.10.3). A constant voltage giving a current of 25 mA was applied until the dye entered the running gel, then the voltage was increased to give a current of 45 mA. When front line of the dye was ~ 1 cm from the bottom edge of the gel, the electrophoresis the gel was carefully removed, the stacking gel was was stopped, scraped off, and the running gel covered with fixer (see section 2.2.2.8.4) in a plastic dish and incubated for 30 min until the dye turned yellow. The gel was then incubated in stain solution (see section 2.2.2.8.5) for 1 h, after which it was incubated overnight with destaining solution (see section 2.2.2.8.7). The gel was rinsed twice with water before being incubated with enhancer (EN³HANCE, New England Nuclear, Boston, Mass., USA) for 1 h. The last incubation was done in water for 30-60 min until the gel was uniformly opaque. The gel was carefully transferred to a piece of blotting paper, placed on a gel dryer plate, and covered with cling film. The plate was connected to a vacuum freezer-dryer for 4-6 h to dry. Autoradiography was carried out using a Kodak XAR-5 film at -70° C for 3-7 days.

2.3.10 Labelling of rabbit antiferritin with ¹²⁵I

A solid phase procedure of labelling in which the specific antibodies are bound to the antigen coupled to a gel was used to label the specific R α -MLFt IgG with ¹²⁵I. This was achieved using the Bolton and Hunter reagent (Amersham International plc., Buckinghamshire, England). The R α -MLFt was attached to a Sepharose-mouse liver ferritin complex (S-LFt) (see section 2.3.6) to protect its binding site, using the method of Alvarez-Hernandez and Loria (1980) with some modifications as follows:

An excess of R α -MLFt (0.5 ml) was passed several times through a small column containing 50 µl of packed S-MLFt complex. The S-MLFt-R-α-MLFt complex was washed 5 times with 0.1 M borate buffer pH 8.5 and left with minimum volume of liquid. The dissolved ¹²⁵I-Bolton and Hunter reagent was dried under a gentle stream of dry nitrogen gas (the nitrogen was passed through a desiccant tube packed with CaCl₂ and silica gel in order to remove any trace of moisture) in a fume hood at room temperature. The S-MLFt-R-α-MLFt was added to the dried ¹²⁵I-Bolton and Hunter reagent and the vial containing the reaction mixture agitated periodically for 15 min on ice. After reacting the mixture with 0.5 ml of 0.2 M glycine in 0.1 M borate buffer, pH 8.5 for 5 min to block remaining reagent, the complex was removed from the vial and passed through a column of Sephadex G25 (Pharmacia) in a disposable 10 ml plastic pipette (Sterilin) to separate the ¹²⁵I-labelled α -MLFt from the other labelled products of the conjugation reaction. One ml of potassium thiocyanate (3 M in 0.1 M borate buffer, pH 9.0) was added to the top of the column and elution was started after 5 min to allow the KSCN to enter the gel bed to release the protein from the antibody. Equilibration and elution were carried out with 0.05 M phosphate buffer, pH 7.5 containing 0.25% (w/v) gelatin to minimize the loss of labelled protein by adsorption. Fractions of 1 ml were collected by hand and an aliquot of each was counted. The presence of SCN⁻ was tested with a drop of 10% (w/v) FeCl_3 which gives a dark red colour. The high molecular weight fractions were pooled, dialysed against PBS to remove SCN⁻, and stored at -20° C. Similar procedure was used to label R α -MSFt (the antibody was provided by Dr X. Alvarez-Hernandez).

2.3.11 Immunoradiometric assay for ferritin (IRMA)

The assay for Ft was based on a two site-IRMA (Miles *et al.*, 1974) with some modifications (Alvarez-Hernandez *et al.*, 1981), the principle of which is the binding of a cold antibody to a surface, allowing the antigen to react, and detecting how much is present by a second labelled antibody. This gives a direct dose-response curve.

Preliminary experiments showed that a multiplate (cat No 76-364-05, S-MRC-96 clear, U-shaped wells-clear, Linbro chemical Co., New Haven, Conn., USA) absorbed more protein than other kinds of plastic plates.

2.3.11.1 Titration of first and second antibodies

The titration of the first antibody was done by maintaining constant the amount of the second antibody (4.5×10^4 cpm/well) and using known Ft concentrations. The titration of the second antibody was carried out by maintaining the first antibody constant. The aim of these experiments was to find the lowest concentration giving a linear response with an adequate working range. Finally, conditions and the working protocol were chosen as follows:

Samples were prepared by lysing the cells by freeze-thawing in PBS to release intracellular Ft. A 1: 500 dilution (50 μ l) of the first antibody made in 0.01 M freshly-prepared sodium carbonate buffer pH

9.2, was incubated overnight to allow adsorption to the wells. Unbound antibody was removed by washing 3 times with 100 μ l of PBS. The plate was then incubated for 4 h with 100 µl of 0.5% (w/v) BSA (RIA grade; Sigma) in PBS to prevent non-specific binding of antigen or of the second antibody. It has also been shown that this improves the linearity of the standard curve (Alvarez-Hernandez et al., 1981). Unbound albumin was removed by washing 3 times with 100 μ l of PBS. Different concentrations of standard ferritin (0-500 ng/ml) or the unknown samples (50 μ l) in at least two different dilutions, were incubated overnight, each one in quadruplicate. All samples were made in PBS containing 1% BSA. Unbound antigen was removed by washing 3 times with 100 μ l of PBS. The plates were incubated overnight with 50 μ l of a dilution containing approximately 4.5 - 5 x 10^4 cpm of the labelled R- α -MLFt. Unbound antibody was removed by washing with 100 μ l of PBS. Finally the wells were cut up with scissors and counted in a gamma counter for 5 min.

2.3.12 Expression of results and statistical analysis

Differences between mean values were analysed by Student's t-test and judged significant when P values were less than 0.05. Unless otherwise stated the results are presented as typical individual experiments representative of a number of separate experiments which gave similar results.

2.4 RESULTS

2.4.1 Titration of concanavalin A concentrations

Lymph node cell suspensions were prepared as described in section 2.3.1. The cells were cultured in serum-free medium containing 50 μ g/ml of 75% saturated Tf, serum-containing medium, or serum-free medium with no addition for 48 h and proliferation responses were assayed as described in section 2.3.2. The concentration of Con A which induced optimum transformation differed from one batch to another. A typical titration pattern is shown in Fig. 3. The concentrations of Con A which induced maximum transformation were 0.25 for cells cultured in serumfree media and 2.25 µg/ml for serum containing medium. Although maximum proliferation in cells cultured in serum free medium with or without Tf was obtained at the same concentration of the mitogen, the proliferative response of cells lacking Tf was very much lower than when Tf was present in the culture medium. A new batch of the mitogen was required during the course of this work. Before use it was titrated and gave 1 and 4 μ g/ml for serum-free medium and serum containing medium respectively (not shown). Therefore, in the subsequent experiments the concentrations of Con A used were 0.25 or 1 µg/ml for serum-free cultures and 2.5 or 4 µg/ml for serum-containing cultures.according to the batch used.



Figure 3 Titration of Con A concentrations in stimulating mouse lymphocyte. mean + SD (n = 4)

2.4.2 Uptake of iron and thymidine by proliferating mouse lymphocytes cultured with transferrin at different iron saturations

Cells were cultured with Con A (1 μ g/ml) in the presence of different iron saturations of Tf ranging from 30% to 120% for 48 h, and the iron associated with the cells was measured as described in section 2.3.3. From Fig. 4 it is seen that iron uptake increased with increasing Tf saturation up to 90% with about 10% of iron being taken up. Beyond this level, when the amount of iron present added as FeNTA exceeded the binding capacity of Tf up to the equivalent of 120% saturation, the iron uptake increased more rapidly with more than 15% of iron present being taken up. To investigate whether iron uptake from higher levels of saturation occurred at the same rate of increase or if there is a limit at a certain level above which the cells cannot taken up more iron, a new experiment of iron uptake with higher concentrations of iron up to the equivalent of 240% saturation of Tf was performed. Proliferation was assayed in parallel as described in section 2.3.2. The results (Fig. 5) show that iron uptake increased at about the same rate up to 120% saturation, as found in the previous experiment. With a further increase in iron present in the cell culture up to the equivalent of 180 and 240% saturation of Tf, the rate of increase in iron uptake rose sharply with 35 and 42% of iron being taken up respectively, indicating that this latter extracellular level of iron (168 ng/ml Fe) did not lead to cellular iron saturation. The two graphs (Fig. 4 and 5) of iron uptake versus Tf saturation show that the proportion of iron taken up to iron present in the medium was roughly constant up to 45% saturation of Tf at 5.2-6.7% of



Figure 4 Iron uptake by Con A-stimulated mouse lymphocytes from transferrin of different iron saturations (representative of 5 individual experiments)


Figure 5

Iron and thymidine uptake by Con A-stimulated mouse lymphocytes cultured with different saturations of transferrin with iron (representative of 4 different experiments). Open points and vertical bars represent mean counts \pm SD (n = 4)

iron present in the medium being taken up. When this level of saturation was exceeded up to around the complete saturation level of Tf, this proportion markedly increased. Within this range of Tf saturation (45%-90%) this proportion was 10.3-11.5% of iron present in the medium being taken up. However, beyond the level of complete saturation of the protein, this proportion was directly proportional to the amount of iron present in the medium.

Proliferation, assayed in parallel showed that at lower iron saturations (15-45%), the proliferative response of the lymphocytes was significantly lower than when Tf was fully saturated with iron. An increase in proliferation was seen as the saturation was increased up to about complete saturation. Thereafter, there was a sharp decline of proliferation once this saturation level was exceeded, and proliferation was reduced to below control level when the saturation of Tf exceeded 200%.

The results obtained here at higher iron saturation of Tf (around the complete saturation) differed from earlier findings (Brock, 1981) in which it was found that optimal proliferation occurred with 30-70% saturation of Tf. In view of this, a further assay of proliferation was performed using larger numbers of different concentrations of iron ranging from 0 to 240% saturation. The results (Fig. 6) were similar to the previous experiment (Fig. 5) with a proliferation peak around 100% saturation of Tf followed by a decrease. Iron free (apo-)Tf was also assayed in this experiment and found to cause a slight but significant increase compared with cells incubated without Tf (P<0.001), perhaps due to slight iron contamination in the culture medium.



Iron concentration (ng/ml)

Figure 6 Thymidine uptake by Con A-stimulated mouse lymphocytes cultured with different saturations of transferrin with iron (typical experiment representative of 4 individual experiments) Points and vertical bars represent mean counts \pm SD (n = 4).

2.4.3 Proliferation of lymphocytes cultured with different iron chelates

The effect of two different forms of iron on proliferation of Con Astimulated lymph node cells was examined. The two chelators were FeNTA, the chelator used to load Tf throughout all the experiments where FeTf was used and a lipophilic iron chelator FePIH. The cells were cultured as described in section 2.3.1 in the presence of increasing FeNTA concentrations. Proliferation was assessed as described in section 2.3.2 to investigate whether adding FeNTA alone inhibited proliferation in the same way as it did when iron loaded Tf was present. **Fig. 7**, which shows thymidine uptake versus iron content of the medium indicates that at very low iron concentration 10 ng/ml, the chelator FeNTA did not have any significant effect compared to control cultures with no addition. At higher iron concentrations a progressive inhibitory effect was seen, and at 60 ng/ml, proliferation was completely abolished.

Fig. 8 shows the effect of FePIH on lymphocyte proliferation. The ratio of Fe:PIH was kept constant at 1:2 throughout all the experiments. In contrast to FeNTA, FePIH was effective in promoting proliferation of Con A-stimulated mouse lymphocytes. At low concentration of iron (<40 ng/ml), the complex FePIH was not effective, but when iron concentration was increased in the medium up to 160 ng/ml, proliferation increased and optimum proliferation occurred. The proliferation at this concentration was as good as that with 90% saturated FeTf (63 ng/ml iron), which was used as a positive control.



Iron concentration (ng/m1)

Figure 7 The effect of FeNTA on proliferation of Con A-stimulated mouse lymphocytes (typical experiment representative of 3 individual experiments). Points, columns and bars represent mean counts \pm SD (n = 4).





Figure 8

The effect of FePIH on proliferation of Con Astimulated lymphocytes (results of a typical experiment representative of 4 separate experiments). Points, columns and vertical bars represent mean counts \pm SD (n = 4). Higher concentrations were inhibitory.

2.4.4 Iron uptake and its intracellular distribution in proliferating mouse lymphocytes cultured with different iron carriers

In order to determine the fate of the iron taken up by proliferating lymphocytes from different sources, uptake of radioactive iron delivered by FeTf, FeNTA, and FePIH into the proliferating Con A-stimulated lymphocytes and the subsequent distribution of the metal among intracellular iron compounds was carried out. The cells were lysed in the presence of DFO and fractions were separated as described in section 2.3.7. In the absence of Tf total uptake of iron from both chelates at both concentrations of iron tested was greater than the amount of iron taken up from Tf by the cells (Fig. 9). Nevertheless iron taken up from FePIH, was much lower than that taken up from FeNTA at both iron concentrations tested. At 10 ng/ml, iron taken up by cells from FeNTA was some 4.5-fold the amount taken up from Tf, while at 50 ng/ml of iron, the amount of iron taken up from this chelate was 9 times the amount taken up from Tf, 40% of iron present in the medium being taken up by the cells in the former case. Iron donation to the cells by the other chelate FePIH, was about 1.5-fold higher than the amount of iron taken up from Tf at both concentrations tested.

Since there appeared to be differences in iron delivery by the three carriers to cells, it was of interest to determine the relative intracellular distribution of iron taken up from these forms of iron. This was carried out as described in section 2.3.7 which allowed intracellular iron to be



Figure 9 Iron uptake by Con A-stimulated mouse lymphocytes cultured with different iron carriers (representative of 4 separate experiments)

assigned to five different compartments consisting of insoluble materials, iron-bound Ft, iron-bound Tf, iron bound to non-Ft non-Tf macromolecule soluble compounds, and iron capable of being bound to DFO.

The major proportion of iron taken up from the iron chelate FeNTA was found in the insoluble fraction which contains cellular debris and probably haemosiderin. This was true at both concentrations of iron tested (Fig. 10 and 11). However, when Tf was present this fraction represented a much lower proportion compared with cells cultured with FeNTA. This was also the case for cells cultured with FePIH. This suggests that much of the iron taken up from FeNTA is not used for cellular metabolism. The chelatable fraction gave fairly similar figures as far as FeTf and FeNTA are concerned. However, with FePIH the proportion of this fraction was slightly higher compared with the proportion found with the two other carriers, at both concentrations of As the iron concentration in the medium was increased the iron. proportion of the low molecular weight fraction showed a noticeable increase with all carriers. However, in absolute figures this fraction was much higher with FeNTA compared with the amount found in this fraction with the two other carriers.

The proportion of intermediate molecular weight-associated iron, defined as soluble non-Ft non-Tf and non chelatable fraction, which probably represents iron being incorporated into enzymes and haemproteins, was much larger in the cells cultured with Tf compared with FeNTA, as it was also in cells cultured with FePIH, especially at 50 ng/ml iron. This indicates that these cells have an increased need for



Figure 10

Intracellular distribution of iron in Con Astimulated mouse lymphocytes cultured with different iron carriers at <u>10 ng/ml_iron</u> (representative of 4 separate experiments)

Ft Fr:Ferritin FractionTf Fr:Transferrin FractionI Fr:Insoluble FractionIMW Fr:Intermediate MolecularWeight FractionWeight FractionLMW Fr:Low Molecular Weight
Fraction



Figure 11 Intracellular distribution of iron in Con A-stimulated mouse lymphocytes cultured with different iron carriers at <u>50 ng/ml Fe</u> (representative of 4 seprate experiments)

Ft Fr:	Ferritin Fraction		
Tf Fr:	Transferrin Fraction		
I Fr:	Insoluble Fraction		
IMWFr:	Intermediate Molecular		
	Weight Fraction		
LMWFr:	Low Molecular Weight		
	Fraction		

metabolically active iron. This was not due to the findings that these cells take less overall, as this fraction was greater in absolute figures than when FeNTA was present.

The ferritin iron fraction formed only a small proportion with all iron carriers, particularly with the chelators.

Iron bound to Tf was found in small amounts in the cells incubated with chelates in absence of any exogenous Tf. This could be due to cross reaction between HTf and MTf with α -HTf which was used to fractionate iron-bound Tf. This Tf might be synthesised by proliferating mouse lymph node cells as it has been reported that human lymphocytes can synthesise Tf in the course of activation (Lum et al., 1986). To investigate whether such cross reaction did occur, 1 mg of ⁵⁹Fe- labelled 50% saturated MTf was passed through two affinity columns, one containing 25 µl of S-S α -HTf and the other column containing 25 µl of S-NSIgs, to correct for non-specific binding. It was found that 60% of the radioactivity was retained in S-S- α -HTf (Table 1), with low non-specific binding (3%), which confirms that there is a cross reaction between human and mouse Tf.

In view of these results, it was decided to investigate Tf synthesis in mouse lymph node cells using an immunoprecipitation method.

2.4.5 Detection of Transferrin synthesis by stimulated mouse lymph node cells

In order to identify Tf synthesis by lymph node cells, the proteins secreted by the cell were endogenously labelled with ³⁵S-cysteine and a

Table 1Binding of ⁵⁹Fe-mouse transferrin to α-
human transferrin affinity column

⁵⁹ Fe binding to affinity column (cpm x 10^3)				
column I		colu	mn II	
S-S-a-HTI	Filtrate	S-S Ig	Filtrate	
972	644	43	1443	
972	644	43	1	

S-S-α-HTf:	Sepharose-sheep antihuman transferrin
S-S-Ig:	Sepharose-sheep immunoglobulins

Tf-immunoprecipitation performed on the supernatant, which was then analysed in an SDS-polyacrylamide gel. Tf synthesis was tested in three systems; in unstimulated, Con A-stimulated, and *in vivo*-stimulated mouse lymph node cells as described in section 2.3.9. In *in vivo*stimulated cells a major band of immunoreactive radiolabelled material similar to mouse Tf was present which was barely detectable when excess cold Tf was added to the supernatant before α -mouse Tf in a competition test or when non-immune serum was used (**Fig. 12**). In Con Astimulated cells, the supernatant showed an extremely faint band, giving a suggestion that there is a less pronounced synthesis of Tf in those cells compared to the cells stimulated *in vivo*. In contrast, in unstimulated cells Tf was not produced in detectable amounts as there was an equally faint band in all three lines.

Lymph nodes contain together with lymphocytes, other cell types i.e. endothelial cells and macrophages. The number of the latter increased sharply within the node after *in vivo* stimulation as judged by ANAE staining (Fig. 13). Thus it is possible that the Tf synthesis seen in *in vivo* stimulated lymph node cells could come from cells other than lymphocytes. It was therefore necessary to investigate Tf synthesis in adherent and non-adherent lymph node cells as described in section 2.3.9. Fig. 14 shows that both groups of cells synthesise Tf upon stimulation. The band corresponding to Tf synthesised by adherent cells (test line) was stronger than the one corresponding to Tf synthesised by lymphocytes. This estimate depends on the efficiency of the separation step of adherent and non-adherent cells. Therefore, to be more accurate, it was decided to use a more specific method consisting of an



Figure 12 Autoradiography of ³⁵S-cysteine incorporated into transferrin by unstimulated, Con A-stimulated and *in vivo*-stimulated lymph node cells

- a: Test (+rabbit α-mouse transferrin)
- b: Competition (+ mouse transferrin then rabbit α -mouse transferrin)
- c: *Control* (+ normal rabbit serum)



Figure 13In vivo-stimulated lymph node cells stained with α-
naphthyl acetate esterase satin. Cells indicated by one
narrow are negative cells (B-lymphocytes or null
cells), double arrows point to T-lymphocytes and
triple arrows to monocytes or macrophages (X 1000).

-



Figure 14 Autoradiography of ³⁵S-cysteine incorporated into transferrin by adherent and non-adherent *in vivo*-stimulated lymph node cells

- 1: Test (+rabbit α -mouse transferrin)
- 2: Competition (+ mouse transferrin, + rabbit α -mouse transferrin)
- 3: *Control* (+ normal rabbit serum)

immunodepletion step of lymphocytes before performing the assay as described in section 2.3.9. The efficiency of this method was checked by assessing the viability of cells at the end of the incubation period with the antibodies which dropped to around 34%. ANAE stain of depleted cells revealed that the proportion of macrophages rose from 15% in nondepleted preparations to 54% in the depleted cells. Fig. 15 shows a band corresponding to Tf appears in the test line of the non-depleted lymph node cells. However, lymphocyte depleted cells show a very much stronger band in the test line. Since the same number of cells was assayed in both cases, the less strong band seen in the non-depleted lymph node cells would correspond to Tf synthesised by the macrophages diluted with lymphocytes in the whole lymph node cells. The conclusion is that macrophages are responsible for Tf synthesis upon *in vivo* stimulation of the lymph node.

2.4.6 Intracellular ferritin levels in proliferating mouse lymphocytes cultured with different iron carriers

Turning back to iron incorporation into ferritin which was found to be very low with practically no increase when iron was increased in the medium (section 2.4.4), it was decided to investigate this matter in more detail. Ft synthesis is known to be stimulated by iron in most tissues tested, especially in liver. Therefore, it was of interest to determine the effect of iron on intracellular levels of Ft in lymphocytes, and to investigate its accumulation during Con A stimulation in the presence of different carriers i.e. FeTf, FeNTA, and FePIH at two different iron concentrations (10 and 50 ng/ml) using an IRMA.



- **Figure 15** Autoradiography of ³⁵S-cysteine incorporated into transferrin by B and T lymphocyte-depleted and non-depleted *in vivo*-stimulated lymph node cells
 - 1: Test (+rabbit α -mouse transferrin)
 - 2: Competition (+mouse transferrin, + rabbit α -mouse transferrin)
 - 3: *Control* (+ normal rabbit serum)

The antibody was labelled with the Bolton and Hunter reagent as described in section 2.3.10. The α -MSFt was tested in the IRMA as 1st and 2nd antibody and found to be very weak giving a low dose response curve (Fig. 16). On the other hand, the α -MLFt gave a good dose response with liver Ft as standard (Fig. 17), hence it was decided to carry out the experiments with only the MLFt- α -MLFt, using a dilution of 1:500 of the antibody.

During the optimization of the method the dilution of the two antibodies was chosen on the basis of finding the lowest working range (Fig. 17). The assay did not present the so called hook effect, which can give rise to false low responses at high concentrations of the antigen, up to at least 0.5 μ g/ml. The sensitivity of the assay was quite good, as low as 5 ng/ml.

Cellular Ft content was generally low with all iron carriers, but was slightly higher in the presence of Tf, and lower in the presence of FePIH (Table 2). At 50 ng/ml iron in the medium, the cells showed only a very modest increase in their intracellular Ft levels compared to cells cultured with 10 ng/ml iron, which indicated that stimulated mouse lymphocytes do not show any marked response to iron by increasing Ft synthesis.



Figure 16 Immunoradiometric assay standard curves with spleen Ft- α -spleen Ft.



Figure 17Titration of first antibody (α-mouse liver ferritin)
with mouse liver ferritin

Table 2Intracellular ferritin levels in Con A-stimulated mouse
lymphocytes cultured in the presence of different iron
carriers

Amount of iron added (ng/ml)	Form of iron	Ferritin content (ng/10 ⁶ cells)
0		$11.6 \pm 3.2^*$
10	Fe Tí	16.6 ± 3.0
(= 14% Tf sat.)	FeNTA	14.8 ± 2.5
	FePIH	BD
50	Fe TY	24.9 ± 4.8
(= 71% Tf sat.)	FeNTA	19.6 ± 5.0
	FePIH	13.3 ± 2.6

* mean of 4 separate experiments ± SDBD: Below Detection

.

2.5 DISCUSSION

Measurement of incorporation of radiolabelled precursors, i.e. ³Hthymidine into DNA was used throughout this study to follow proliferation. Therefore, it was of prime importance to titrate the mitogen. In the absence of Con A the proliferation response was very much lower than when it was present. However, there was still some incorporation of ³H-thymidine by the cells. This might have been the result of non-specific stimulation of some lymphocytes or due to the presence of mesenteric lymph node cells which often contain a proportion of cells that have been antigenically stimulated in vivo prior to being cultured. The background stimulation observed was greater when the cells were cultured in serum-containing medium compared to cultures in serum free-medium. Nevertheless, it was always very much lower than the response of mitogen-containing cultures. It was also found that a higher concentration of Con A was needed to induce an optimum proliferation response in serum-containing medium than when serum-free conditions were used. This was probably due to Con A being bound to serum proteins in the medium, causing a reduction in the total number of mitogen molecules available for activation of the cells.

2.5.1 Uptake of iron and thymidine by proliferating mouse lymphocytes cultured in the presence of different transferrin saturations

The uptake of iron by lymphocytes from different iron saturations of Tf showed a gradual increase at Tf saturations below complete

saturation. However, when iron was added beyond the binding capacity of Tf, iron uptake increased dramatically. This suggests that uptake of Tf-bound iron proceeds via the process of Tf-TfR endocytosis, and was controlled via TfR expression, which is modulated by the intracellular iron level rather than the rate of cellular proliferation per se (Pelosi-Testa et al.. 1988). However, in the presence of excess free iron in the medium uptake rose sharply, suggesting that this uptake took place in a non-specific uncontrolled manner, resulting in an excessive accumulation of the metal inside the cell. The findings of this study agreed well with previous observations concerning other types of cells which showed that cultured Chang cells (Bailey-Wood et al., 1975), liver slices, reticulocytes, and macrophages (Morgan, 1981) took up iron from Tf at rates which depend on the percentage saturation. Below the complete saturation of Tf the iron uptake graphs (Fig. 4 and 5) revealed two linear phases. The ratio of iron taken up to the amount of iron present in the medium was constant up to around half saturation of Tf (45%), but when saturation exceeded this level, the proportion of iron uptake to iron available increased. This suggests that diferric Tf donates iron to lymphocytes at a higher rate than monoferric Tf. The initial step in the uptake of iron involves the binding of Tf to specific cell membrane receptors. Transferrin iron exchange has been shown to involve a random loading of iron binding sites, one iron atom at a time (Huebers et al., 1984). On the other hand, the uptake of the FeTf complex by receptors on cell membranes is not random, there being marked preference of receptors for diferric over monoferric Tf. It has been shown in rat hepatocytes (Young and Aisen, 1981) and in rabbit reticulocytes (Young et al., 1984), that the strength of interaction of Tf with the receptor

decreased from diferric to apo-Tf. Diferric Tf was also found to be a better donor for erythroid cells than monoferric Tf (Brown et al., 1975; Hahn et al., 1975; Lane, 1973). However, Verhoef et al (1978) have reported that iron uptake by rat bone marrow cells, fibroblasts and lymphoblasts incubated for 3 h was not influenced by the degree of iron saturation of Tf. The reason for this discrepancy could be the short time of incubation used in this study which does not allow uptake to occur. Brock and Rankin (1981) have shown that uptake of iron in mitogenstimulated lymphocytes is insignificant up to 20 h, and that it occurs mostly between 30 and 70 h incubation. The amount of iron taken up from Tf therefore, depends on the number of each form of the molecule present in the medium, and because of its lower affinity, apo-Tf does not compete significantly for receptors. Thus the degree of transformation of lymphocytes in medium containing partially saturated Tf will most likely depend on the proportion of the different Tf molecular species present. Hence, this study, indicated that at low Tf saturations there are not enough iron loaded Tf molecules and the proliferation was low. An increased rate of transformation is associated with an increased percentage of saturation of Tf with iron, up to the complete saturation of Tf where optimum proliferation occurs. It has been shown that proliferation of mouse lymphocytes was lower in media supplemented with serum from iron-deficient mice than when serum from normal or iron-repleted mice was used (Mainou-Fowler and Brock, 1985). The addition of sufficient iron to bring the iron level of the deficient serum to that of normal serum improved its ability to promote proliferation.

The results of this study suggest that the amount of iron bound to Tf may be important in determining the degree of proliferation. This is in line with the suggestion that the growth promoting effect of FeTf may be related to the delivery of iron to a specific intracellular site, which is probably related to the iron-requiring enzyme ribonucleotide reductase (Hoffbrand *et al.*, 1976). It seems therefore, logical that with Con Astimulated mouse lymphocytes, the higher the saturation of Tf with iron the better the *in vitro* proliferation. However the slight enhancement in proliferation seen when lymphocytes were cultured with 'iron-free Tf' compared to control cultures with no addition might suggest that Tf alone may have an enhancing effect. The more likely explanation is that apo-Tf in fact becomes slightly saturated with iron up to 8% (Brock, 1981), due to endogenous iron contamination of the culture medium.

In contrast, the addition of iron beyond the binding capacity of Tf produced the opposite effect on DNA synthesis, i.e. increasing iron levels result in reduction of transformation, indicating that there is no simple correlation between external iron supply and increased DNA synthesis. The presence of Tf might protect the cells from the toxic effects of iron, since non-Tf bound iron added as FeNTA either above saturation of Tf molecules present in the medium or alone was found to be inhibitory to cell transformation. This could be due to iron in this form mediating cytotoxicity as discussed below.

The results obtained in the present study at higher iron saturations of Tf contradict previous findings which demonstrated an optimum effect of Tf around physiological saturation levels (30-70%) with a falling off at higher saturations (Brock, 1981). The discrepancy could lie in a methodological difference between the two systems. In Brock's system, FeTf solutions were made up immediately before use, thus assuming that completion of iron binding occurred almost instantaneously. However, Zapolski and Princiotto (1977) measured iron binding to Tf by elution from an anion exchange resin column and found that complete iron binding to Tf is not instantaneous, but requires up to 1 h. Therefore, binding of iron to Tf may not have reached completion, and thus, free iron may have existed in these solutions which when added to lymphocyte cultures might have been able to inhibit proliferation as discussed above, even if unsaturated Tf was present in the cultures. It was shown in the present study that iron in the form of FeNTA, which was used to load Tf, is inhibitory to proliferating mouse lymphocytes (see section 2.4.3). Aruoma and Halliwell (1987) reported that iron-loaded Tf (Fe₂Tf) showed no protective ability, but does not itself accelerate OH' production unless chelating agents are present in the reaction mixture, especially if the protein is incorrectly loaded with iron. On the hand in a study reported by Sibille et al. (1987) using saturated Tf loaded by a specific procedure (Bates and Schlabach 1973) which was carefully freed of exogenous iron by two cycle of gel filtration they could not find any stimulation of OH' formation by iron loaded Tf at pH 7.4. Moreover, Gutteridge et al (1981) showed that Tf was an inhibitor of iron-dependent lipid peroxidation at pH 7.4. All the experiments in this work used FeTf solutions which had been allowed at least 2 h for binding to occur. This difference in experimental technique may, therefore, account for the difference in the results between the two studies when high iron levels were used. Similarly, Broxmeyer et al (1983) have reported that iron saturated Tf, but not apoTf decreased the production of granulocytemacrophage colony stimulating factor (GM-CSF) by PHA- or Con-A-

stimulated human lymphocytes. However these authors achieved Tf saturation by addition of excess of iron salt followed by removal of unbound iron by dialysis and passage through an iron exchange column. It is known that great care has to be taken when iron is added to Tf and it requires the presence of an excess of a low molecular weight chelator, such as citrate or NTA, to prevent hydrolysis, polymerization, and non-specific binding of iron (Bates and Schlabach, 1973). Therefore the inhibitory effect reported by Broxmeyer's group could be due to non-specific iron bound to the Tf molecules becoming detached during the assay and acting as a catalyst for reactions which produce free oxygen radicals.

2.5.2 The effect of different iron chelates on mouse lymphocyte proliferation

Lymphocytes may take up or respond to iron in the medium in two different ways. Firstly they may take up iron physiologically from Tf under the control of cell metabolism and specific receptors on the cell surface. Iron taken up in this form may be used immediately by the cell and thus permit an increase in the cellular metabolic response to mitogens. The other way is uncontrolled non-specific capture of iron and/or transport across the cell surface. The results of this study seem to indicate that uptake of iron from FeNTA is probably an example of the latter. The finding that FeNTA did not promote proliferation suggests that iron bound to NTA was probably handled by cells in a way that failed to make the metal available for metabolic use. It has been reported that free iron leads to formation of polymeric complexes (Spiro *et al.*, 1967). It is possible that this polynuclear iron may bind non-specifically to the

plasma membranes of the cells and exert a toxic effect as will be discussed in the next section. Nevertheless FeNTA, unlike iron salts is known to be quite an efficient chelator in maintaining iron soluble in the extracellular milieu, but intracellularly the iron may polymerise and bind non-specifically to the membrane, as will be discussed in the next section. The finding that as little as 40 ng/ml iron of FeNTA significantly lowered the rate of proliferation to below control levels suggests that the lymphocyte is extremely sensitive to oxidative injury. In comparison, only when the concentration of iron of FeNTA was as high as 10 μ g/ml, was the growth rate of RL34 rat liver cells cultured in serum-free conditions affected, while the corresponding of NTA alone gave the same growth as that of the control cells (Yamada et al., 1987). Many in vivo and in vitro experimental studies exist showing that FeNTA-induced organ injury is related to lipid peroxidation. A single injection of FeNTA was sufficient to produce some hepatic and pancreatic injury and to initiate lipid peroxidation in the liver (Yamanoi et al., 1982). Fodor and Marx (1988) compared ferrous-ammonium sulphate, ferrous-ascorbate, ferric-citrate and ferric-NTA, and showed that FeNTA was by far the most effective in inducing lipid peroxidation of rabbit small intestinal microvillus. The subject of free radicals affecting lymphocytes is discussed more fully later (section 2.5.3)).

The other iron chelate FePIH has been found to support good proliferation of Con A-stimulated mouse lymphocytes. This chelate was identified by Ponka *et al* (1982) and unlike Fe-citrate and FeNTA was found to be able to donate iron to erythroid precursors for use in haem synthesis. In the present study, it was found to be as effective as FeTf in promoting lymphocyte proliferation at iron concentrations ranging from 40 to 160 ng/ml. These results are in agreements with a report that FePIH permitted growth of embryonic kidney cells (Landschulz *et al.*, 1984) and that the response of these cells was physiological since the cells in the model system used convert to epithelial cells and form tubules. The same group have shown that some other low molecular weight iron chelators such as Fe-acetate, FeNTA, Fe-ascorbate, and Fe-citrate, and unchelated ferrous sulphate could not support proliferation to the same extent as FePIH and FeTf. The present study also confirmed the findings of Brock and Stevenson (1987) with mitogen-stimulated mouse lymphocytes cultured *in vitro* with FePIH in the absence of Tf.

The chemical features of the complex FePIH indicate that it is lipophilic. Landschulz and Ekblom (1985) have found that FePIH has an octanol/saline partition coefficient near 1. This suggests that it can easily and passively enter and leave the phospholipid bilayer of the membrane. The chelator could therefore bypass the TfR and deliver iron directly through the lipid bilayer. Nevertheless, this could not explain why FePIH is more efficient than FeNTA in promoting proliferation since FeNTA was found to donate iron intracellularly (Brock and Rankin, 1981). The finding that levels of iron >160 ng/ml were inhibitory to the cells when bound to PIH suggests that although iron bound to PIH bypasses the TfR route, it is possible that its uptake is less well controlled than that of Tf-bound iron and any excess over that needed for biosynthetic processes becomes toxic for the cells. As will be discussed in the next section, the pattern of intracellular iron distribution suggests that while both chelators bypass the specific uptake mechanism needed to acquire Tf-bound iron, they take different routes once internalized.

The findings that FePIH is equivalent to Tf for supporting proliferation of the cells has some important implications, as it provides another piece of evidence to show that the growth-promoting activity of Tf can be accounted for by its role in providing iron to the cells. Therefore it seems unlikely that Tf could promote cell growth through interaction with its receptor i.e. as a mechanism for transduction of growth signals as has previously been suggested (Brock and Mainou-Fowler, 1983). This conclusion is in agreement with the study of Perez-Infante and Mather (1982) who clearly showed that Tf cannot stimulate growth in the absence of iron, since the effect of apo-Tf on cell growth was blocked by a complete removal of contaminating iron with DFO. Some other iron containing molecules, such as Fe-dextran may also replace the growth promoting effect of Tf on mitogen stimulated lymphocytes (Tanno *et al.*, 1982), as do Hb and haem with 3T3 cells (Young *et al.*, 1979).

2.5.3 Uptake and intracellular handling of iron from different carriers by mouse lymphocytes

The differences in the ability of the two chelators in promoting proliferation compared with Tf, discussed in previous section, suggested that there might be differences in the manner of iron acquisition. To this end, a set of iron uptake experiments were performed using FeTf, FeNTA, and FePIH

The amount of iron taken up from FeNTA was found to be greater

than from FeTf or FePIH, especially at 50 ng/ml iron where a sharp increase in iron uptake was seen. These results are in accordance with the report that mouse lymphocytes took up iron more rapidly from FeNTA and Fe-citrate complexes than from Tf (Brock and Rankin, 1981). Similar findings have been reported with other cell types such as Chang liver cells (White and Jacobs, 1978), which were found to take up 30 times more iron when FeNTA is used as donor compared with FeTf. However, Taylor *et al* (1988) found that iron uptake from ⁵⁹FeNTA added directly to cultures of mitogen-stimulated peripheral blood lymphocytes was quantitatively very similar to that from ⁵⁹FeTf. This contrast with the results of the present study is probably due to the fact that their cultures contained 5% serum, the Tf of which could take up ⁵⁹Fe in situ from ⁵⁹FeNTA.

As discussed in the previous section the explanation of the large difference in iron uptake between FeNTA on one hand and FeTf and FePIH on the other is that iron taken up from the former is not used for metabolic needs, and its accumulation is probably not controlled by any mechanism, whereas iron acquired from FeTf is under control of intracellular metabolism, as will be discussed later. As far as FePIH is concerned, although it can pass through the plasma membrane easily, because of its higher affinity for the metal the chelator would still be able to bind iron intracellularly and donate it to the cell at a slower rate compatible with metabolic needs. Indeed, Taylor *et al* (1988) suggested that although iron utilization is related to cellular activity, the uptake mechanism is only activated when an increase in iron metabolism has exhausted internal stores. However any control of iron uptake from FePIH is achieved by an unknown means other than TfR expression. Pronase, which inhibits Tf binding to cells probably by digesting TfRs has virtually no effect on the uptake of ⁵⁹FePIH (Ponka *et al.*, 1979). On the other hand, FeNTA is less likely to be able to traverse the membranes, as mentioned in the previous section. However, Brock and Rankin (1981) have demonstrated that FeNTA donates iron intracellularly to mitogen stimulated lymphocytes and the iron may therefore remain trapped within endocytic vesicles. It could also be that binding of iron to the membranes occurs after entry of the chelator to the cell as will be discussed below.

After determining the differences in radioactive iron delivery to the cell by the different carriers, the subsequent distribution of the metal among intracellular iron compounds was carried out in order to determine the fate of iron taken up, and to interpret their differential effect on proliferation. This was carried out by disruption of the cells and studying the cytosolic phase using affinity chromatography and ultrafiltration.

After its entry to the cell the intracellular pathways of iron are not well defined. After being released into the cell, iron finds its way to multiple specific sites. Since transport of uncomplexed iron is very unlikely, once iron is inside the cell it is presumed to be bound by a carrier. Jacobs (1977) suggested the presence of a labile unidentified low molecular weight iron transit pool in the cytosol which was available for binding by iron chelating agents and which maintained a dynamic equilibrium with many iron-requiring enzymes. Iron chelators could in principle modify iron metabolic pathways by competing with or adding to this cellular iron pool. However, opinions are divided about the existence of such a pool. The identity of the iron carrier molecules in this fraction are still undefined. There are two possibilities. One is that specific iron-transporting molecules do not exist in the cytosol and that the iron is transported through a variety of iron-chelating amino acids and salts (reviewed by Romslo, 1980). The second possibility is that there is a specific iron-binding and transporting factor with a very high turnover, which makes it very difficult to detect (Bakkeren et al., 1985). Presumably such species are required both within the endosome and in the cytosol. The movement of iron through the labile pool is supposed to be linked to several factors, such as metabolic use of iron for haem synthesis and other iron-containing enzymes (Wrigglesworth and Baum, 1980; Schneider and Erni, 1981). The metabolic state of the cell would dictate the flux of iron from this common pool. Incoming iron would also contribute to the flux into this pool, and it would be affected by every intracellular event involving iron mobilization.

The fraction chelatable by DFO, which probably corresponds to this pool, was found to represent a relatively small proportion in lymphocytes cultured with all carriers tested. These findings are in agreement with the report of Bomford *et al* (1986) that in mitogenstimulated human lymphocytes, the non-Ft non-haem compartment is initially the largest and the most active for maintenance of cell iron prior to incorporation into different compartments, but gradually this fraction decreased and remained constant after 3 h (only 15% of cellular iron). However, this compartment could also contain iron in non-haem ironcontaining enzymes, since in their study they did not further fractionate this compartment. In the present study this low molecular weight fraction in absolute figures was very much higher in cells cultured with FeNTA at 50 ng/ml iron since these cells take more iron overall compared to cells cultured with FeTf or FePIH. Abnormal expansion of the chelatable pool might be associated with an increased amount of toxic low molecular weight iron which may enhance free radical formation and lead to cell injury as discussed below in the case of cells cultured with FeNTA.

In cells cultured with FeNTA iron was found predominantly in an insoluble non-Ft macromolecular form which probably contains mostly cell membranes, while in cells cultured with FeTf and FePIH, this fraction represents only a small proportion of intracellular iron. This is in line with reports that the high proportion taken up form FeNTA, but not iron originating from Tf was in the pellet of cells probably bound to the cell membrane of mouse lymphocytes (Brock and Rankin, 1981). White and Jacobs (1978) have also shown that iron delivered from FeNTA to Chang cells was found largely in the membrane pellet and very little was in the form of Ft. However, it has been reported that the major part of iron donated by FeNTA to fetal rat hepatocytes was present in the cytosol and not membrane bound as reported with Chang cells (Lescoat *et al.*, 1989), but they suggested that this might be related to iron exchange between FeNTA and Tf resulting in part of the iron entering the cells in a Tf-bound form.

Three hypotheses can be advanced here to explain the nature and origin of the insoluble fraction. Firstly iron in this fraction could simply consist of iron trapped in the membrane. Secondly iron in this fraction could be bound to Hs or Hs-like molecules resulting from saturation and then degradation of Ft to which it was initially bound. Thirdly, iron is
this fraction does not all come classically from Hs, but results from the formation of polymer complexes which then bind to the membranes, especially when iron is taken up non-specifically.

In the case of the first hypothesis, the high proportion of iron in the insoluble fraction in cells cultured with FeNTA could be a result of iron being trapped in the cytoplasmic membrane. Fe³⁺ can only remain soluble at neutral pH by displacing water with ligands of strong chelating agents. The NTA:Fe ratio 5:1 used throughout this study is supposed to be safe and prevent polynuclear complex formation in the medium. However, Marx (1989) has reported that the mononuclear iron form of Fe³⁺ only occurs with an excess of citrate to iron of at least 500:1 and not 20:1 as it was previously thought (Bates and Schlabach, 1975). Therefore some polymers may also be present in the system used in this study when FeNTA was used. Electron microscopic evidence for the presence of electron-dense materials on the surface of lymphocytes incubated in the presence of Fe-citrate have been presented (Nishiya et al., 1980). Landschulz and Ekblom (1985) showed that FeNTA has a low partition coefficient between octanol and saline (<0.01), implying that iron associated with the cells probably consists of polynuclear iron bound non-specifically to the cell, and they concluded that FeNTA cannot traverse the lipid bilayer. However, White and Jacobs (1978) have shown that iron in the membrane pellet of Chang cells cultured with FeNTA was not removed by DFO, which suggests that the binding of iron to the membrane occurs after entry into the cell rather than extracellularly, assuming it is bound to the membrane. This finding does not support the hypothesis that iron uptake from FeNTA occurs by adsorption of the chelate on the cell surface rather than intracellularly.

With regard to the second hypothesis, it is known that lymphocytes are not specialised for iron storage the potential capacity of these cells to synthesise Ft to sequester iron in a harmless form is not high (Summers et al., 1975; Lema and Sarcione, 1981; Summers and Jacobs, 1976; Dörner et al., 1980; Pattanapanyasat, 1988; 1989b, and confirmed in this study, see section 2.4.6). As a consequence, the large amount of iron taken up from FeNTA would be quickly transported from the labile iron pool and load the limited amount of Ft present, which would result in rapid Ft degradation and the generation of Hs. It is known that the excessive loading of Ft is followed by lysosomal uptake of the molecules and their subsequent degradation (Trump et al., 1973; Weir et al., 1985), which is generally considered to be a relatively inactive form of storage iron (O'Connell et al., 1986a, 1986b). The polymerization of Ft molecules in solution has been recognized for some time, and Niitsu and Listowsky (1973) have reported that this process preferentially affects iron-rich molecules, with the incorporation of rich molecules into lysosomes, degradation of the protein shell, and formation of Hs. Indeed, It has been shown that while Ft is the major iron protein present in the normal liver, in iron overload Hs predominates (Selden et al., 1980). Therefore, it might be assumed that in the present experiments, this process is accelerated when cells are incubated with FeNTA due to the high level of uptake. However even when the concentration of FeNTA in the medium is low (10 ng/ml) and the subsequent iron uptake comparable to that from Tf at 50 ng/ml, the abnormal distribution between cytosol and membrane iron remains with a larger accumulation in the latter. Ultrastructural studies of liver in artificially iron-loaded animals (Trump et al., 1973) or in human liver

biopsy material (Iancu *et al.*, 1977) suggest that iron loading results in the accumulation of Ft in lysosomes at which sites it is converted to Hs. This is in agreement with other studies with liver and heart cells (Iancu *et al.*, 1987; Jacobs *et al.*, 1978; Link *et al.*, 1985) which found that unlike the physiological uptake of Tf iron which is primarily located to cytosol, low molecular weight iron complexes, such as FeNTA and Feammonium citrate are rapidly confined to siderosomes. This hypothesis could be valid especially in cases of physiological uptake of the metal which implies that excess internalized iron would pass through Ft before the former is converted to Hs. However, in cases of non-physiological uptake it could be that Hs is not the only insoluble iron storage compound and polymers of iron complexes might also exist.

Turning to the third hypothesis, it is possible that once inside the cell, FeNTA might result in abnormal intracellular iron processing due to dissociation of iron from the carrier and release of large quantities of free iron. This could saturate all available natural low molecular mass compounds capable of binding iron inside the cytosol. This iron could accumulate, polymerise and precipitate, leading to subsequent non-specific lysosomal incorporation of some of these polymers, adding to the to iron in Hs derived from Ft. Therefore the insoluble fraction of these cells would in fact consist of both free and lysosomal-bound iron polymers together with Hs. Although there was no evidence that polymerization occurs in the cytosol, and FeNTA is thought to be an excellent chelator in maintaining extracellular iron in a soluble form, this might occur if the chelate was subject to degradation by the metabolism of the cells. The hypothesis that FeNTA could be subject to degradation by

the metabolism of the cell is supported by the findings of Nakamoto *et al* (1986) who incubated Ehrlich ascites tumour cells with ⁵⁹FeNTA and Fenitrilotri(¹⁴C)acetate at NTA:Fe of 5:1 and found that the intracellular ratio NTA:Fe decreased as the extracellular concentration of FeNTA increased reaching almost 1:1 at an iron concentration of 3.6 mM.

Thus the conclusion is that the way the cells handle iron may depend on the form in which it was supplied. If the second hypothesis is valid in the case of cells cultured with FeTf, the third one is more likely to be true especially in the case of cells cultured with FeNTA, although the first hypothesis can not to be excluded in this case.

If FeNTA binds non-specifically to the membrane and polynuclear iron complexes were unable to cross the membrane as discussed above they may exert their toxic effect at this site by providing active iron which may participate in hydroxyl radical generation which promotes lipid peroxidation, leading to cell damage. This view is in accordance with the report of Soyano *et al* (1985) that the inhibitory effect of FeNTA on lymphocyte proliferation is due to formation of iron polymers which affect the membrane of these cells. Landschulz and Ekblom (1985) have also reported that the toxic effect of FeNTA on kidney tubules is due to alteration of their membranes by polynuclear iron.

If FeNTA was able to enter the cell as discussed above, is it iron that is incorporated into Hs after Ft degradation following excessive uptake into the cell that is responsible for the inhibitory effect on lymphocyte proliferation, or is it formation of polymeric iron after degradation of the chelator and liberation of iron inside the cell? Alternatively, toxicity might be a function of the abnormally large intermediate pool which in absolute figures is very much greater when cells were cultured with FeNTA at 50 ng/ml.

When cells were exposed to a lower iron concentration (10 ng/ml) in the form of FeNTA, toxicity did not occur (as judged by ³H-thymidine uptake which was similar to control cultures) and the cells were probably able to sequester iron in Ft, with formation of Hs which also represented the major fraction of cellular iron of these cells. However when cells were cultured with 50 ng/ml iron, this fraction represented the same proportion as when cells were cultured with 10 ng/ml and yet proliferation was reduced as compared with control, suggesting that possibly there is polymeric iron in this fraction. If this assumption is true most of iron in this fraction in cells cultured with 10 ng/ml iron would probably be in Hs and very little or none in the form of polymers. A close positive correlation has been shown between enhanced lysosomal fragility and liver Hs content (Selden et al., 1980). Hs accumulation within tissues could provoke lysosomal damage by increasing lipid peroxidation (Selden et al., 1980). However this does not necessarily imply that Hs is the cause of cytotoxicity. On a unit iron basis, Hs has been found to promote OH' generation to a much smaller extent than Ft, which suggests that iron in the form of Hs is far less active in promoting lipid peroxidation (O'Connell et al., 1985). Moreover, addition of Hs to normal liver homogenates is incapable of promoting lipid peroxidation (Bacon et al., 1985). It could thus be argued that conversion of Ft into Hs is biologically advantageous, in that it would diminish the occurrence of oxygen-radical reactions in the presence of excess iron in the medium. It is therefore likely to be that the iron deposits in the cell represent

immobilized metabolically inactive iron. Hence Hs formation appears to be a protective mechanism and overload of the intermediate pool may be responsible for toxic deposition of iron.

In conclusion, rather than being an instrument of cellular damage, the formation of Hs in response to iron loading may initially represent an important mechanism protecting cellular organelles against iron toxicity. However, with increasing iron uptake this protective mechanism may no longer be sufficient and polynuclear iron formed after degradation of FeNTA following abnormal saturation of low molecular iron complexes in the intermediate pool may be responsible for its participation in peroxidation damage.

What makes FePIH better than FeNTA in this respect? PIH has a very high and specific affinity to Fe^{3+} especially at neutral pH (Richardson *et al.*, 1989) Its chelating efficiency on a weight-perweight basis is equal to or slightly better than DFO (Hershko and Weatherall, 1988), and since the latter is known to inhibit irondependent OH production (Gutteridge *et al.*, 1979; Halliwell, 1985), at least under certain conditions (Braughler *et al.*, 1988), this would suggest that FePIH must be very efficient in preventing the reduction of Fe^{3+} , and thus preventing it participating in radical formation. This view is supported by the report of Mello Filho *et al* (1984) which showed that 2,2'-bipyridine, an iron chelator related to PIH, prevents both DNA damage and the killing of mouse cells by H_2O_2 by entering the cells and chelating their intracellular iron. However, bipyridine is an Fe^{2+} chelator and the mechanism of action might be different.

The results of size of the intermediate molecular weight fraction fit

very well with the results of thymidine uptake. This fraction, which represents iron incorporated into non-Ft non-Tf iron-containing proteins and thus reflects the metabolic activity of the cell, was found to represent relatively the largest proportion in cells cultured with FeTf or FePIH. Iron in this fraction is bound to many enzymes and coenzymes. Probably the most important of all as far as these cells are concerned is ribonucleotide reductase (Brown *et al.*, 1969). The importance of this enzyme in DNA synthesis was shown by Brockman *et al* (1971) who reported impaired incorporation of ³H-thymidine into DNA by a leukaemic cell line when the cells were cultured in medium containing inhibitors of the ribonucleotide reductase system. However, in cells cultured with FeNTA this fraction represents a smaller proportion.

The finding that the fractionation profile of iron acquired from FePIH is similar to that of iron acquired from FeTf indicates that the former complex might be capable of donating its iron to the intermediate low mass compound(s), in which iron is fed *en route* to functional ironcontaining enzymes or the storage compound Ft, at equal rate to FeTf. Alternatively, FePIH might simply jump this step and the actual complex fulfills this role. The last probability is more likely to be true since FePIH can pass directly to the cytosol (Landschulz and Ekblom, 1985) and its ability to bind iron more strongly than FeNTA would enable it to deliver its load to the mitochondria to join the normal metabolic pathways. This might also reflect the fact that one part of the PIH molecule, pyridoxal, a form of vitamin B_6 , is a natural biological substance (Ponka *et al.*, 1979). This same argument can be applied to explain the means by which FePIH donates iron to the cell, in which case one cannot exclude the possibility that PIH crosses biological membranes using a transport system for pyridoxine derivatives. However, Solomon (1982) demonstrated that pyridoxal uptake by red cells was not affected by ATP depletion or addition of inhibitors of anion transport channels. This seems to indicate that red cell uptake of pyridoxal is not dependent upon a specific transport mechanism and both pyridoxal and PIH may just diffuse non-specifically across the cell membrane. On the other hand, Ponka *et al* (1979) have reported that ATP seems to be required for iron mobilization from reticulocytes by PIH, suggesting that the whole chelate unlike its pyridoxal moiety, may be taken by an active transport system.

It is worth noting that with all carriers the overall intracellular distribution of iron among different intracellular compartment was not altered by changes in the amount of iron available. These results are in agreement with the findings of White and Jacobs (1978) with Chang cells.

In previous reports using cultured fibroblasts (Octave *et al.*, 1981), human erythroleukaemic cells (Klausner *et al.*, 1983), or established liver cell lines such as Chang cells (White *et al.*, 1976), the majority of iron taken up from Tf was sequestered in Ft. In sharp contrast, the Ft-iron fraction was by far the lowest in all cultures of lymphocytes. However at both concentrations tested it was slightly larger in cells cultured with FeTf. Under increased metabolic demands for iron the major source of iron to be used would be that present in the low molecular weight fraction. In view of these results, which suggest a very limited ability of these cells to incorporate iron into Ft, it was of interest to investigate Ft accumulation in response to iron presented in different forms.

The IRMA for mouse Ft used in this study to measure the lymphocyte content of Ft detected predominantly the L type, since the antibody used was an α -mouse liver Ft. Although there was a relative difference in the level of Ft in proliferating mouse lymphocytes cultured with three different iron carriers when the iron concentration in the medium was increased, this difference was not very pronounced in comparison to other cell types such as monocytes (Dörner et al., 1983b). This suggests that there is a basal level of Ft which could not be greatly increased by the addition of any form of iron to the incubation medium. However, Jones et al (1983) have reported that normal human peripheral blood lymphocytes contain a relatively high proportion of H subunits, and incubation of these cells with iron increases preferentially the H subunits (Worwood et al., 1984). This could partly explain the relatively low Ft content detected in these cells in the present study. A low increase in L-rich Ft in response to iron has also been reported by other authors (Pattanapanyasat et al., 1988; Pattanapanyasat, 1989a). However, the control of Ft synthesis is much less well studied in mouse lymphocytes than in human lymphocytes, and it is not really known if mouse Ft behaves similarly. Nevertheless, since the cells in the present work were incubated for a relatively long period its was more relevant to assess the L-rich ferritin which unlike H-rich Ft is more stable and better suited for long-term iron storage (Drysdale, 1988). Together with the proliferative activity of the cells this can give a relative idea of how these cells are succeeding as far as iron detoxification is concerned. The results of the present study are in agreement with some reports which showed that addition of either iron (Summers et al., 1975) or iron saturated Tf (Summers and Jacobs, 1976), failed to induce human

lymphocyte Ft synthesis in vitro. These results are also in accordance with the findings of Pattanapanyasat et al (1988) which showed that there was a significant increase in Ft synthesis in lymphocytes cultured in the presence of PHA when compared to non-stimulated cells, and that the presence of FeNTA made little difference to the rate of Ft synthesis. Lema and Sarcione (1981) have reported that administration of increasing amounts of iron to rats in vivo induced both liver and peripheral blood lymphocyte Ft synthesis, but incubation of liver and lymphocytes with increasing concentrations of iron in vitro stimulated liver but not lymphocyte Ft synthesis. These data suggest that induction of lymphocyte Ft synthesis observed after iron administration in vivo resulted from secondary stimulatory mechanisms rather than to iron per se. The results of the intracellular Ft content together with the results of the intracellular distribution of iron lend support to the theory of Mattia et al (1986) who proposed that the regulation of intracellular iron distribution is dependent upon absolute Ft levels and is unaffected over a range of amounts of iron delivered to the cells

The fact that proliferating mouse lymphocytes cultured with both physiological and non-physiological forms of iron showed no marked increase in synthesis of Ft could be interpreted as indicative that these cells are directing most of their iron either to a metabolically inert pool in the case of FeNTA (which failed to support proliferation) or into enzymatic needs (in the case of cells cultured with FeTf and FePIH) as judged by the intracellular distribution of iron (see section 2.4.4). In cells specialized for iron storage where the potential accumulation of the metal is high, iron can regulate the synthesis of Ft by translational control mechanisms. It has been generally considered that the main function of Ft in these cells is to store iron not immediately required for metabolic activity and release it to metabolic iron pools when required. It is well known also that most cells and tissues, when loaded with iron, incorporate the metal into Ft. Increased synthesis also occurred during inflammation and during cell differentiation (Fibach *et al.*, 1985). Since the main function of Ft in lymphocytes is storage of iron for intracellular use (house keeping) rather than for use by other cells the mechanisms of regulation could differ.

2.5.4 Transferrin synthesis by mouse lymph node cells

It is well established that the major site of synthesis of Tf is the liver (Morgan, 1981). Nevertheless, extrahepatic sites of Tf synthesis may exist (section 1.1.2.1). In the course of the present study the initial suggestion of Tf synthesis by mouse lymph node cells came when Tf was detected after chromatographing the lysate of cells cultured with ⁵⁹Fe-labelled chelator in the absence of any exogenous Tf. Early studies reported that the human lymph node (Prunier *et al.*, 1964) makes Tf, and it has also been reported that human lymphocytes might have the capacity to secrete Tf (Soltys and Brody, 1970; Nishiya *et al.*, 1980; Broxmeyer *et al.*, 1983). This was confirmed by the study of Lum *et al.*(1986) who showed that activated human helper T-lymphocytes synthesise Tf. Therefore, to investigate whether this is also true for mouse lymphocytes a study of Tf synthesis by those cells was carried out.

In this study the mouse lymph node was found to be one of the extrahepatic local sites of Tf synthesis. Tf was found to be actively synthesised by lymph node cells upon *in vivo* immunological stimulation but not by quiescent cells. However, the present study has failed to demonstrate the synthesis of the other smaller Tf-like molecule (60 Kd) which was found to be synthesised by activated human lymphocytes (Lum *et al.*, 1986). Moreover *in vitro* Con A-stimulated cells failed also to show an active synthesis of the protein. All this might suggest that another type of cell could be responsible for Tf synthesis within the lymph node. Therefore, it was of interest to determine which particular cell populations within the tissue are involved.

It has been known for some time that macrophages can synthesise Tf (Phillips and Thorbecke, 1966; Haurani et al, 1973). It has also been suggested that endogenous Tf might act as a vehicle for the release of iron from macrophages (Haurani and Ballas, 1984). These cells, which are in close contact with lymphocytes within the node, were found to increase sharply after in vivo stimulation of the node. All this makes the macrophage the most likely candidate. In subsequent experiments it was confirmed that these cells are responsible for Tf synthesis in the stimulated lymph node. As mentioned earlier, the initial detection of newly synthesised Tf was in lysates of Con A stimulated lymph node cells cultured in Tf-free medium and was possible because of its content of radioactive iron. The reason for the failure of the immunoprecipitation method to detect Tf synthesis in these cells might be explained by the possibility that the detection by the radioactive iron content of the protein is more sensitive than the autoradiography of ¹²⁵I-labelled Tf. But

whether this iron was incorporated by Tf molecules endogenously or occurs after lysing the cells is not known.

The mechanism that regulates the expression of the Tf gene in the lymph node and the physiological role of Tf during the immunological activation process are presently unknown. It is possible that the Tf gene in these cells is not expressed in the resting stage under normal conditions, and that in the host immune response the gene is switched on to provide a source of available iron in an autocrine/paracrine manner to support localized proliferation of those rapidly dividing cells. This is in agreement with finding of Vostrejs *et al* (1988) who reported that Tf synthesis in a lung cancer cell line increased more than 10-fold when cells entered active phases of the cell cycle, a time when Tf and particularly iron is necessary for cell division to proceed.

The need for a local supply of Tf to these cells may result from the restricted environment within the lymph node during immunological stimulation, when iron is being withheld from circulation (Weinberg, 1984). During acute phase reaction such as bacterial infections or neoplasia the body responds with a protective reduction in the concentration of available iron resulting in a rapid drop in the iron saturation of circulating Tf (Zarrabi et al., 1977; Lee. 1982: Finkelstein et al., 1983), by mechanism(s) which are not fully defined Moreover, in both aseptic and microbial inflammation Tf yet. concentration is reduced (Kumar et al., 1978). This is assumed to be a characteristic of the host defence mechanism against microbial pathogens and neoplasia (Ballantyne, 1984; Hunter et al., 1984), since under these conditions serum iron is less accessible to the pathogen. On the

other hand, under these conditions, shortages in iron supply could affect the proper function of the immune cells which are liable to find themselves in a state of suddenly increased activity caused for example by invasion of foreign organisms. This is especially true within lymphoid organs where a sudden increase in demand for iron may not be met fast enough resulting in impaired immune response. Cells of the lymph node would then be better equipped if they could acquire iron independently of plasma Tf by synthesising their own Tf locally in order to proliferate and exert their effector functions. Therefore, macrophages may serve as an intermediary in the transport of iron to proliferating lymphocytes under certain conditions. Specialized cellular proliferation in vivo by tissues that are not well vascularized might be limited by insufficient delivery of Tf-bound iron from plasma. A striking example of these tissues which is quite similar in certain aspects to the lymph node, is the testis in which Tf synthesis by Sertoli cells provides iron to proliferating spermatocytes (Skinner et al., 1980). The brain, which is separated from blood stream by the so-called blood-brain barrier, has been shown to synthesise Tf (Levin et al., 1984; Aldred et al., 1987), which contributes to maintaining homeostasis in the extracellular environment of the brain.

Thus lymphocytes may be capable of modulating their own supply of iron through influencing the regulation of macrophage Tf synthesis via lymphocyte-macrophage contacts and/or diffusible substances. By analogy to this, Le Magueresse *et al* (1988) have found that removal of germ cells contaminating the Sertoli cell cultures resulted in a significant decrease in Tf secretion. This hypothesis awaits further investigation. CHAPTER THREE

1 . N.

THE EFFECT OF DIFFERENT LEVELS AND FORMS OF IRON ON HUMAN LYMPHOCYTES

, 1

3.1 INTRODUCTION

The preceding chapter reported evidence for the importance of iron availability on the proliferation of mouse lymphocytes in response to mitogens and that iron in the form of FeNTA, but not Tf-or PIH-bound iron had an inhibitory effect. Iron bound to NTA was probably toxic to the cells. Although the use of cells from an animal model has allowed variables which could complicate the interpretation of the results such as infections, age and nutritional differences to be eliminated, it was of prime importance to perform comparable studies on human lymphocytes. If the inhibitory effect of iron under the conditions reported in the previous chapter with mouse cells is applicable to human lymphocytes, the immunological status of humans may be seriously impaired under similar conditions. It is thus intrinsically more valuable to study human lymphocytes which would also enable comparison to be made with clinical situations. Therefore one of the aims of the work presented in the present chapter is to determine to what extent the findings with the mouse cells apply to human cells, and to investigate whether the way cells from these two species behave in the presence of different levels and forms of iron is different. It is known that these cells are exposed to different conditions in vivo, as Tf saturation in the plasma of mice is higher that in humans. In vivo saturation of mouse Tf with iron is 65-80% (Puschmann and Ganzoni, 1977; Kuvibidila et al., 1983) against about 30% in humans (Morgan, 1981). Thus the former may be expected to be better equipped than the latter in coping with excess iron. It could therefore be hypothesized that the inhibitory effect of excess iron would be more

pronounced in the human system. The work reported in the present chapter is therefore an extension of these studies to normal human lymphocytes, which examines the effect of different forms of iron on lymphoblastic transformation.

The results reported in the previous chapter also gave some evidence that mouse lymphocytes do not have a marked capacity to synthesise Ft in response to iron in the medium. To investigate whether the same is true in human lymphocytes, Ft synthesis in proliferating normal human lymphocytes has been investigated in more detail, by looking at the effect of a larger range of iron concentrations on cellular Ft content. The results reported with mouse cells suggested that lymphocytes were unable to proliferate because they could not efficiently handle excess iron by increasing Ft synthesis as discussed above (section 2.5.3).

Several studies have suggested that besides its well known antimicrobial activity, Lf posses several characteristics which could implicate it as potential modulator of immune cell function (Duncan and McArthur, 1981; Brock, 1985). In this respect, the release of this protein and its local accumulation in inflammation are of particular relevance. Therefore Lf might have a role in controlling lymphocyte proliferation. It was also hypothesized that Lf could sequester excess iron in the medium and could prevent its toxic effect on the cells, and hence enable the cells to proliferate when excess iron is present. However no definite proof for this is available. Therefore, in this study Lf was tested for its effect on human lymphocyte proliferative response and in particular to address the question of whether this protein which has a much higher affinity constant for iron than Tf (Aisen and Leibman, 1972) is capable of overcoming the inhibitory effect of excess iron when all the Tf present in the medium is saturated. The study was performed on human rather than mouse cells because it is quite easy to prepare sufficient quantities of human Lf from colostrum, whereas mouse Lf is not readily available.

In the previous sections (2.4.2 and 2.4.3) it was shown that iron modulates proliferation of T-cells. These cells consist of different subsets exerting different functions in the process of an immune response. The results reported in the previous chapter of a poor in vitro response of lymphocytes to mitogen stimulation in the presence non Tfbound iron in the form of FeNTA suggests that impairment may occur to the clonal expansion of these cells which is in line with previous reports (Good et al., 1986; 1987). Depression of the specific immune response in iron overload is less well documented than in iron deficiency, although disturbances have been reported in iron overload-related diseases (reviewed by de Sousa, 1989). However correlation between Tlymphocyte subset abnormalities and iron loading remains quite speculative. There is little information on whether different iron conditions can affect the ratio of lymphocyte subsets in vitro. Therefore, to address this question and to see whether the inhibitory effect on proliferation is ascribed to action on all T-cell subsets irrespective of surface phenotype, experiments investigating the effect of different concentrations of iron bound to chelates or Tf were performed to investigate which subsets are altered by iron. This would further our understanding of the immunological.regulatory properties of iron.

It is well established that transformed cells generally require less serum growth factors than normal cells (Draser and Irving, 1973). A non-Tf-mediated pathway of iron incorporation has been demonstrated in malignant cell-derived lines in culture (Taetle et al., 1985; Basset et suggesting that neoplastic cells might have different al., 1986), mechanisms for regulating their requirements of the metal. Therefore the study performed on normal human lymphocytes was extended to a human leukaemic cell line CCRF-CEM. This was carried out to investigate whether these cells, which have the property of unremitting proliferation, respond in the same way to iron as their normal homologue undergoing mitogen-stimulation, which leads to terminal differentiation. In addition, Tf synthesis in CCRF-CEM cells was investigated to see whether, as in the activated normal human T-lymphocytes (Lum et al., 1986), the Tf gene is expressed in these cells. This could serve as an extra means of acquisition of the iron that is needed by these constantly transformed cells. In this respect, endogenous cellular production of Tf could serve in an autocrine growth regulatory capacity, and uncontrolled Tf biosynthesis by these neoplastic cells could be an important mechanism by which a neoplasm might establish autonomous proliferation, and some evidence for this has already been presented (Kitada and Hays, 1985; Morrone et al., 1988).

In summary therefore, the work reported in this chapter aims to investigate:

 The effect of different forms and amounts of iron on transformation of human lymphocytes and the T-cell line CCRF-CEM and compare them with the results obtained with mouse cells.

,

- 2) The effect of iron on cellular Ft levels in human lymphocytes in relation to its effect on proliferation.
- 3) The effect of the Lf on lymphocyte proliferation.
- 4) The effect of different forms and amounts of iron on the expression of T-cell subset markers.
- 5) Tf synthesis by CCRF-CEM cells.

3.2 METHODS

3.2.1 Cell culture conditions of human mononuclear cells

Peripheral blood was obtained from healthy volunteers aged 19-48, collected into sterile plastic universals containing 50 IU/ml heparin (Leo laboratories Ltd, Princes Risborough, Bucks, UK) and mixed 1:1 with RPMI-1640 medium. Mononuclear cells were isolated by density gradient centrifugation through lymphocyte separation medium (Flow). The cells were washed twice with RPMI-1640 medium, viability assessed using eosin exclusion, and finally resuspended in complete medium composed of RPMI-1640 supplemented with penicillin-streptomycin at 100 IU/ml 0.3 mg/ml L-glutamine, and HSA at and 100 µg/ml respectively, 1 mg/ml. The cells were cultured at a concentration of 2×10^6 cells/ml in plastic conical test tubes in the presence of purified phytohaemagglutinin (Wellcome, Dartford, England) at appropriate concentrations. The cells were cultured at 37° C in an atmosphere of 5% CO₂, 95% air for 72 h. Proliferation responses were assayed as described in section 2.3.2 with mouse cells.

3.2.2 Indirect immunofluorescence

After terminating the incubation period, cells which had been cultured as above were washed twice with PBS/FCS (5%). Three tubes containing aliquots of 5 x 10^5 cells each were put on ice. The cells were spun at 800 rpm for 5 min at 4° C, the supernatant aspirated, and cell pellets were left with as little liquid as possible. To each tube was added 10 µl of a 1:20 dilution in PBS of one of the following monoclonal antibodies; α -CD3, α -CD4, and α -CD8 (Serotec). The tubes were well mixed on a vortex, and left incubating for 20 min on ice. During the incubation period each tube was vortexed twice. Tubes were then centrifuged and cells were washed three times in PBS/FCS to remove any unbound antibody. Cell pellets were left with as little liquid as possible and 20 µl of a 1:32 diluted fluorescein isothiocyanate (FITC) conjugated F(ab')2 fraction rabbit- α -mouse immunoglobulin (Serotec) was added to the cells which were again incubated on ice for 20 min. Following the second incubation period, the cells were washed as above and diluted to 0.7 ml with PBS/FCS. Cytopreparations were prepared by spinning 100 µl of each sample on to a microscope slide at 800 rpm for 5 min using a Shandon-Elliot cytocentrifuge. The slides were taken out, allowed to air dry and then fixed with ethanol/glacial acetic acid (95%/5%) for 20 min at -20° C. Following cell fixation, slides were washed by transferring them into 3-4 changes of fresh PBS. The slides were then mounted in 90% glycerol/10% PBS containing 0.1% sodium azide under a coverslip and sealed with nail varnish. The cells were observed under oil immersion in an Olympus BH-2 fluorescent microscope, which allows the observation of cells by both ultraviolet and phase contrast. This enables the fluorescent cells to be counted under UV light and the total mononuclear cells under phase contrast. Four random fields were chosen, and 200-280 cells were counted per slide.

3.2.3 Preparation of lactoferrin

The isolation of this protein was performed following the method reported by Johansson (1969), with some variations. Human colostrum (30 ml) (kindly provided by staff of the breast milk bank, Yorkhill Hospital, Glasgow, Scotland) was spun at 1250 g for 30 min at 4° C. Fat and debris were discarded, and the aqueous layer was collected. The defatted colostrum was diluted with 4 volumes of sterile saline. FeNTA (10 mM stock solution) was added (30 µl/ml of original volume of colostrum) which turned the colour of the colostrum solution to red-pink. Dry CM-Sephadex C50 (Pharmacia) (0.5 g) was added and the mixture gently stirred for 1 h. After allowing the gel, which turned to a reddish colour, to settle, the supernatant was aspirated, and the gel was washed with 3 changes of 50 ml of saline with 5 min gentle stirring each time. The gel was then packed into a small column made from a 20 ml syringe barrel with a Millipore prefilter in the bottom. To elute contaminating proteins a reservoir of 0.05 M Tris HCl, pH 8.0, was connected and allowed to run through until no further protein eluted, as judged by reading the optical density at 280 nm. The reservoir buffer was changed to 0.05 M Tris/2.0 M NaCl, pH 8.0, to elute the Lf and fractions of 1.5 ml were collected. The optical densities at 280 and 470 nm were measured (Fig. 18), and the Lf peak collected. The figure shows that there was a perfect match between the protein peak fractions and the red colour indicative of their iron content, demonstrating a good separation procedure. The fractions comprising the main peak were collected and the deep red solution was then concentrated by ultrafiltration and dialysed against 4 daily changes of PBS, pH 7.2 at 4° C. The solution was then



Figure 18 Optical densities of lactoferrin fractions at 470 and 280 nm eluted from a Sephadex CM 50 gel using 0.05 M Tris, pH 8.0 followed by 0.05 M Tris/2.0 M NaCl, pH 8.0.

dialysed against 3 daily changes of H_2O and freeze dried. To obtain apo-Lf, Lf was dissolved in 0.1 M acetate, 0.2 M Na_2HPO_4 and 0.04 M EDTA buffer (pH 4.0). The solution was dialysed overnight against the same buffer at 4° C. The EDTA was removed by a 24 h period of dialysis against PBS.

3.2.4 Cell line

CCRF-CEM (Foley *et al.*, 1965), a human T-leukaemic cell line was routinely grown in an atmosphere of 95% air and 5% CO₂ at 37° C in RPMI-1640 complete medium (see section 2.2.2.6) containing 10% FCS. Cells were maintained in log-phase growth at densities of 0.3 to 0.8×10^6 /ml by replenishing the cultures by 5-fold dilutions with fresh medium every three days. The cells were incubated initially in RPMI-1640 alone for 1 h, and washed twice, prior to being used in any experiment. This was done to allow exocytosis of any endogenous Tf from the culture medium to occur, thus minimizing interference.

3.2.5 Detection of transferrin synthesis by CCRF-CEM cell line

CCRF-CEM cells (4 x 10^7) were taken from cultures 24 h after addition of fresh medium, washed 3 times in PBS, and preincubated in RPMI-1640 (w/o Cys) for 90 min before being incubated with 100 µCi ³⁵S-cysteine for 5 h. The immunoprecipitation procedure was carried out as described in section 2.3.9 with the following modifications:

1) To test samples, 10 μ l of sheep α -HTf (SAPU) was added.

- 2) To competition samples, 30 μ g of HTf was added followed by 10 μ l of sheep α -HTf.
- 3) To control samples 8 μ l of normal sheep serum was added.

.

.

3.3 RESULTS

3.3.1 Titration of phytohaemagglutinin

Mononuclear cells were prepared as described in section 3.2.1, and cultured in serum-free medium containing either 50 μ g/ml of 75% saturated transferrin (serum-free) or 10% FCS. After 72 h the proliferation responses were assayed as described in section 2.3.2 for mouse cells. The optimum concentrations which induced maximum transformation were 1 and 5 μ g/ml for cells cultured in serum-free medium and serum-containing medium respectively (Fig. 19). The ratio of the responses for cultures in serum-free medium to FCS is in accordance with previous findings with Con A in the case of mouse lymph node cells (see section 2.4.1).

3.3.2 The effect of iron saturation of transferrin on human lymphocyte proliferation

The cells were cultured at 2×10^6 /ml in the complete medium (see section 2.2.2.6) in the presence of 1 µg/ml PHA and increasing iron saturation of Tf ranging from 0-200% at a concentration of 50 µg/ml. **Fig. 20** shows the proliferation of cells taken from three different individuals. Although there was a considerable variation between individuals, the same pattern of proliferation was seen. As the Tf saturation increased there was an increase in proliferation which reached a peak at around 100% iron saturation of the protein. Once this saturation level was exceeded, there was a noticeable decline of proliferation.



PHA concentration (µg/m1)

Figure 19 Titration of response to phytohaemagglutinin of human lymphocytes. Points and vertical bars represent mean counts \pm SD (n = 4).



Transferrin saturation (%)



3.3.3 The effect of FeNTA and FePIH on human lymphocyte proliferation

Cells were cultured at 2 x 10^6 cells/ml in the complete medium in the presence of 1 µg/ml PHA (see section 3.2.1). FeNTA and FePIH were added to the cultures at three different iron concentrations (10, 40, and 60 ng/ml). FeTf was used as a positive control with iron content equivalent to those of the two chelators i.e. iron saturations of 14, 56 and 84%. As expected, FeTf showed an enhancement of proliferation as the iron content of the protein increased up to 60 ng/ml, the equivalent of 84% saturation (Fig. 21). Though FePIH showed a significant enhancement as its iron content increased, it was somewhat less good than FeTf unlike the findings with mouse lymph node cells (see section 2.4.3). On the other hand FeNTA was not effective at promoting proliferation, and showed an inhibitory effect as the iron concentration was increased in the culture medium, compared with control cultures with no addition(P < 0.05 at 60 ng/ml iron).

3.3.4 The effect of iron saturation of transferrin on intracellular ferritin of proliferating human lymphocytes

The effect of Tf saturation with iron on intracellular Ft levels was investigated. Cells were cultured in serum-free medium in the presence of Tf of various iron saturations for 72 h, washed twice in PBS and resuspended at 10^6 /ml in 1 ml PBS containing 0.1% BSA, 1 mM PMSF and 0.7 µg/ml pepstatin. Lysates were prepared by three cycles of



Figure 21 The effect of different iron carriers on proliferation of PHA-stimulated human lymphocytes (representative of 5 separate experiments). The columns and the vertical bars represent mean counts \pm SD (n = 4). * P<0.005; ** P<0.05 compared with control (no addition), $\ddagger P < 0.01$; $\ddagger P < 0.025$ compared with culture containing equivalent amount of iron bound to Tf. freezing and thawing. Ft levels in lysates were determined by a ¹²⁵I radioimmunoassay method using the Magic^R Ft RIA kit (Ciba) which measures predominantly L type Ft (kindly performed by the Dept. of Biochemistry, Gartnaval General Hospital, Glasgow). The RIA failed to detect any Ft in lysates that were stored at -20 or -70° C. This indicates that Ft may lose its antigenicity upon storage at freezing temperature. Therefore, it was decided to test the samples straight after preparing them or storing them at 4° C overnight at the most if necessary. Cellular Ft content increased only slightly with increasing extracellular iron concentration up to the level of complete saturation of Tf where optimum proliferation was seen as judged by ³H-thymidine uptake (Table 3), which suggests that there was some correlation between the degree of transformation and Ft synthesis. However, when this level was exceeded there was no increase in intracellular Ft levels even when extremely high extracellular iron concentrations were present in the cultures. This might indicate that like proliferating mouse lymphocytes, human lymphocytes have a limited ability to synthesise Ft in response to iron.

3.3.5 The effect of lactoferrin on human lymphocyte proliferation

ApoLf or 90% saturated FeLf were added at 50 μ g/ml to cells cultured in serum-free medium (see section 2.2.2.6) containing varying iron saturations of Tf. The effect of apoLf and 90%-saturated FeLf added alone was also included and control cultures were set up with cells in medium alone.

Table 3	The effect of transferrin saturation on intracellular ferritin
	in proliferating human lymphocytes

	Control	Transferrin saturation (%)					
	(No Tf)	0	30	90	200	3200	
Ferritin (ng/ml/10 ⁶ cells)*	34±8	43 ±6	67±10	119±14	90±9	106 ±15	
³ H-thym. uptake (cpm/10 ⁵ cells) [†]	3315 ± 275	4603 ± 508	8955 ±796	12588 ± 982	7943 ± 801	453 ± 34	

* mean of triplicate experiments \pm SD †mean \pm SD (n = 4) .

Iron-loaded Lf did not have any significant effect on proliferation when added alone compared with control cultures with no addition (Fig. 22), which indicates that FeLf could not substitute for Tf in promoting lymphocyte proliferation. However, the iron-free form of the protein caused a significant decrease of transformation when added alone (P < 0.005) or to cells cultured in the presence of increasing iron saturation of Tf. Apo-Lf caused a slight but significant decrease of the proliferative response of these cells up to the level of 75% iron saturation of Tf(P < 0.025, P < 0.05, and P < 0.025 with cultures containing apoTf, 15% and 75% saturated-Tf respectively). On the other hand, FeLf did not cause any significant change in thymidine uptake when added to cells cultured in the presence of Tf at any saturation level. However at much higher iron concentration (200%), the presence of apo-Lf in cultures markedly restored blastogenesis. At higher concentration (600%) Lf lost its protective effect as proliferation dropped below control levels. Finally, at extremely high iron concentration in the medium (3200%), all proliferation was abolished.

3.3.6 The effect of lactoferrin on intracellular ferritin levels of proliferating human lymphocytes

In view of the results of the previous section it was of interest to investigate the effect of Lf on intracellular Ft levels to see whether the possible sequestration of iron from the cells by Lf reduced synthesis of Ft. Cellular Ft content was measured as described in section 3.3.4. **Table 4** shows that while FeLf did not have any effect on intracellular Ft levels, the iron-free form of the protein caused a slight decrease



Figure 22

The effect of lactoferrin on proliferation of PHAstimulated human lymphocytes cultured with or without transferrin (representative of 4 individual experiments). Columns and vertical bars represent the mean values \pm SD (n = 4). * P<0.0005; ** P<0.005; *** P<0.025; **** P<0.05 compared with control (no addition)

Table 4	The effect	of 1	lactoferrin	on	intracellular	ferritin	in
proliferating human lymphocytes							

.

	Control (No Tf)	ApoLf	FeLf (90% sat.)
Ferritin (ng/10 ⁶ cells)*	45 ± 6	29±4	40±8

* mean of duplicates \pm SD

.
compared to control cultures with no addition

3.3.7 The effect of iron saturation of transferrin on T-cell subsets

The following subsets; CD3, CD4, and CD8 were determined by an indirect immunofluorescence as described in section 3.2.2. The proportion of CD3⁺ cells in the cultures was not significantly affected by at any of the iron concentrations tested (**Table 5**). Neither apoTf nor a high concentration of iron in the medium (200%) had any effect on the proportion of these cells in the cultures compared to control cultures with no addition. However, the high level of iron (200% Tf saturation) decreased the ratio of CD4/CD8 by almost a half. This was due to a decrease in the proportion of CD4⁺ cells and a corresponding increase in the percentage of CD8⁺ cells compared to control cultures with no addition. In contrast, lower Tf saturations with iron and iron-free Tf did not show any effect on the CD4/CD8 ratio which remained constant as compared with control cultures with no addition. Finally at 75% saturation of Tf, the proportion of CD4⁺ cells was significantly higher compared to control cultures with no addition.

3.3.8 The effect of FeNTA and FePIH on T-cell subsets

The effect of iron in the forms of FeNTA and FePIH and in the absence of Tf on T-subset marker expression was investigated. An iron concentration of 50 ng/ml equivalent to Tf saturated to 71% was chosen,

156

Table 5The effect of transferrin saturation on expression
of CD3, CD4, CD8, by proliferating human
lymphocytes

	CD3+	CD4+	CD8+	CD4 +/CD8+
No addition	64.8±5*	40.4±5	19.2±2	2.1
Apo Tf	66.2±6	43.0±4	21.1±3	2.0
FeTf (30%)	71.8±5	51.2±6	23.3±3	2.2
FeTf (75%)	73.0 ± 6	53.2±6 ••	23.7±6	2.2
FeTf (200%)	64.1 ± 7	33.3±3 •••	27.9±3•	1.1

*mean of percentage of +ve cells \pm SD (n = 4) (representative of 3 separate experiments). •P<0.01; ••P<0.02; •••P<0.05 compared to control (no addition). and 71% saturated Tf was also included as a positive control. As it was seen in previous section with non-Tf bound iron in the presence of iron loaded Tf, iron as FeNTA lowered the ratio of CD4/CD8 cells (**Table 6**).This was the consequence of a considerable decrease of the proportion of CD4⁺ cells and an increase of the percentage of CD8⁺ cells compared to control cultures with no addition. FePIH did not have any significant effect on the ratio of CD4/CD8 as compared with control. On the other hand, the proportion of CD3⁺ cells in the presence of FeNTA in cultures was not significantly different from those containing FeTf or FePIH. The results obtained with 79% saturated Tf gave fairly similar results as compared to the previous experiment.

3.3.9 Proliferation of CEM cells in the presence of different iron saturations of transferrin

The cells were cultured in conical test tubes at 5×10^4 cells/ml in RPMI-1640 containing 1 mg/ml HSA in the presence of different iron saturations of Tf ranging from 0 to 3200% for 24 h. The degree of transformation was assessed as described in section 2.3.2 for normal lymphocytes. First of all it is of interest to notice the relatively high proliferative background of cells cultured in medium without Tf (Fig. 23). Nevertheless the addition of apo-Tf caused a noticeable increase in the rate of proliferation. At lower iron saturations of Tf proliferation of CEM occurred at almost the same rate with very little variation up to 200%. In the presence of larger amount of iron up to 1200%, a gradual decrease in proliferation occurred as the concentration of iron increased. Nevertheless good proliferation was still maintained even at this high level of iron. Only when this level was exceeded did proliferation drop

Table 6The effect of Fe-transferrin, FeNTA, and FePIH on
expression of CD3, CD4 and CD8 by proliferating
human lymphocytes

	CD3 ⁺	CD4 ⁺	CD8 ⁺	CD4 ⁺ /CD8 ⁺
No addition	68.8±6*	43.4±4	20.5±3	2.1
FeTf (50 ng/ml Fe) (=71% sat.)	71.1±5	51.3±5 •••	23.2±4	2.2
FeNTA (50ng/ml Fe)	66.6±6	31.2±4•	27.9±3**	1.1
FePIH (50 ng/ml Fe)	68.8±8	47.4±3	22.7±4	2.1

* mean of percentage of +ve cells \pm SD (n = 4) (representative of 3 separate experiments). • P < 0.01; ••P < 0.02; •••P < 0.05 ccompared to control with (no addition).



Transferrin saturation ($\% \times 10^{-1}$)

Figure 23 The effect of transferrin saturation on CCRF-CEM cell growth (representative of 5 separate experiments). Columns and vertical bars represent mean values \pm SD (n = 4).

below that of the control without Tf.

3.3.10 Proliferation of CEM cells in the presence of FeNTA and FePIH

Cultures containing 5 x 10^4 cells/ml were set up in RPMI-1640 containing 1 mg/ml HSA in the presence of FeNTA and FePIH with increasing iron concentrations ranging from 10 ng/ml to 1 µg/ml which is equivalent to a range of Tf saturation from 14 to 1429%. Cultures containing these iron saturations of Tf were included. Both chelates promoted good proliferation compared to controls without addition (Fig. 24). Indeed, there was no significant difference between either chelates and FeTf in supporting proliferation, indicating that the two chelates are equally effective at promoting proliferation of CCRF-CEM cells as FeTf. Surprisingly, the rate of proliferation did not drop below the control levels at high iron concentration, which indicates that these cells could handle very large amounts of iron in both physiological and nonphysiological forms, perhaps due to a greater ability to increase Ft synthesis.

3.3.11 The effect of iron saturation of transferrin on intracellular ferritin of CCRF-CEM cells

The effect of Tf saturation with iron on intracellular Ft levels in CCRF-CEM was examined to investigate whether these cells are able to show active response to iron by increasing Ft synthesis. Cells were



Transferrin saturation (%)

Figure 24

The effect of different iron carriers on CCRF-CEM cell growth (representative of 4 separate experiments). Columns and vertical bars represent the mean values \pm SD (n = 4). * P<0.005; ** P<0.01 compared with control (no addition). cultured in serum-free medium in the presence of Tf of various saturations for 24 h. The Ft levels were assayed as described in section 3.3.4 for normal human lymphocytes. Surprisingly, Ft content of cells (Table 7) or at least L-rich type Ft content at all Tf saturations tested was extremely low (in the order of 10-25 X less compared to normal human lymphocytes, see Table 3) and instead of showing as expected a marked increase of intracellular Ft content as iron concentration was increased in the culture medium the cells did not show any increase whatsoever.

3.3.12 Proliferation of CEM cells in the presence of lactoferrin

Cells were cultured at 5 x 10^4 cells/ml in RPMI-1640 containing 1 mg/ml HSA. ApoLf or 90% saturated Lf (50 µg/ml) were added either alone or to cultures containing FeTf or FCS. As was found with normal human lymphocytes apoLf and FeLf did not have any effect on the proliferation of CEM cells when added alone (Fig. 25). ApoLf was only inhibitory when added to cultures containing FeTf ranging from 0 to 75% saturation, while beyond this level of saturation up to 200% and with cultures containing FCS, it did not have any effect. In contrast at higher iron saturation (600%), apoLf has a slight but significant restorative effects on the modest inhibitory effect of high iron concentration on the proliferation of CCRF-CEM cells(P<0.01).

Table 7	The effect of	transferrin	saturation	on	intracellular
	ferritin in CCF	RF-CEM cell	s		

	Control _ (No TY) _	Transferrin saturation (%)				
		0	75	200	400	800
Ferritin (ng/10 ⁶ cells)*	3.5 ± 0.5	2.0 ± 0.3	3.0 ± 0.1	4.5 ± 0.6	4.2 ± 0.4	3.9 ± 0.7

* mean of duplicates \pm SD

.

•



Iron concentration (ng/ml)

Figure 25

Effect of lactoferrin on growth of CCRF-CEM cells cultured with or without transferrin (representative of 5 separate experiments). Columns and vertical bars represent the mean values \pm SD (n = 4). * P<0.001; ** P<0.01 compared with control (no addition).

3.3.13 Detection of transferrin synthesis by CEM cells

CCRF-CEM in logarithmic phase, i.e from cultures 24 h after subculturing, were subject to the immunoprecipitation method for detection of Tf synthesis as described in section 2.3.9. Fig. 26 clearly shows that these cells, which were cultured before the experiment in serum-containing medium, have the ability to synthesise Tf. The disappearance of the band corresponding to Tf when excess cold Tf was added to the system to compete with the newly synthesised radiolabelled Tf, or when the anti-serum was omitted demonstrate specificity of the assay



Figure 26

Autoradiography of ³⁵S-cysteine incorporated into transferrin by CCRF-CEM cells

- 1: Test (+ sheep α -human transferrin)
- 2: Competition (+ human transferrin, + sheep α -human transferrin)
- 3: *Control* (+ normal sheep serum)

3.4 DISCUSSION

The work reported in this chapter has attempted to push the investigation of the action of iron on lymphocytes further on by looking at its effect at different levels and forms on human lymphocyte transformation *in vitro* in comparison to mouse lymphocytes and a T-leukaemic cell line. Non-Tf bound iron in the form of FeNTA was shown to inhibit murine lymphocyte transformation, probably because of the inability of these cells to respond to iron by increasing Ft synthesis. To confirm and extend some of these effects seen on mouse lymphocytes, and to establish whether these findings apply to the normal human cells and their transformed counterpart is one of the aims of this chapter. It also attempts to define more closely the differential effect of iron on different T-cell subsets. The effect of another related iron-binding protein, Lf, was also investigated

3.4.1 The effect of iron saturation of transferrin, FeNTA, and FePIH on proliferation of human lymphocytes

There was a considerable variation in the absolute figures of the proliferative response between individuals. Despite this the same pattern of proliferation in the presence of increased iron saturation of Tf was obtained. Variability in human lymphocyte responses to mitogens have been reported by other investigators (Fan *et al.*, 1977; Platz *et al.*, 1976), and the presence of this variability has become an accepted problem in the clinical evaluation of human lymphocyte mitogen

responses. As in the case of mouse lymph node cells (section 2.4.2), the degree of transformation of human peripheral lymphocytes upon PHA stimulation cultured in medium containing apo-Tf (in practice 6-8% saturated) was lower compared with the response of the cells cultured in medium containing higher iron saturations of Tf, up to around 100% saturation. These results confirm the previous findings with mouse lymph node lymphocytes that one of the important factors in controlling the proliferative response of lymphocytes *in vitro* is the content of transferrin-bound iron in the culture medium.

As far as the effect of the chelators on human lymphocyte proliferation is concerned, it was found that as with mouse lymphocytes, iron-NTA cannot substitute for iron-Tf and at relatively high concentrations actually inhibits proliferation, but chelation with PIH was more effective. Nevertheless, the results suggest that iron taken up in this form may be used immediately by the cell and permit an increase in the cellular metabolic response to mitogens. Overall, the findings with human lymphocytes were broadly similar to the one obtained using the mouse system. However, the enhancing effect of both FeTf and FePIH on human lymphocytes was less than the effect on mouse cells. This was by almost a third (~600% increase at optimum saturation of Tf compared to control with mouse cells against $\sim 400\%$ with human cells). This is possibly due to the ability of human (Lum et al., 1986), but not mouse lymphocytes to make their own Tf, or at least to the possibility that the former could be actively engaged in synthesising this protein while the latter are less, as it is discussed in section 2.5.4. The same may be true in the case of FePIH which was not quite as efficient compared to its

effect on mouse lymphocytes.

Surprisingly, human lymphocyte seems to be more resistant to the inhibitory effect of FeNTA compared to murine cells, knowing that the latter were taken from the blood of these animals where iron saturation of the transferrin is known to be higher compared with humans. While FeNTA at 40 ng/ml iron did not have any significant effect on human lymphocyte, this concentration lowered the proliferative capacity of mouse lymphocytes by 56% as compared to control. However, when this concentration was increased to 60 ng/ml the proliferation of both mouse and human cells was affected, but only by 46% with human cells compared to 80% in the mouse one.

3.4.2 The effect of iron saturation of Tf on intracellular ferritin levels of proliferating human lymphocytes

Since lymphocytes are not specialized for iron storage, the sequestration of iron by Ft might serve a detoxification function. Unlike other cell types, mouse lymphocytes seem generally not to have any marked increase in intracellular Ft levels in response to iron presented in different forms (see section 2.4.6). To confirm this finding and to extend it to human lymphocytes, it was decided to investigate this in more detail. In the experiments with mouse cells, just two different concentrations of iron were used for each form of iron. In the present chapter the relationship between iron availability (in the form of increasing iron saturation of Tf) and cellular Ft content was looked at.

Ft synthesis by human lymphocytes cultured in the presence of Tf

increased only modestly with increasing intracellular iron concentration. This confirms previous evidence with mouse lymphocytes that these cells have a limited capacity to synthesise Ft in response to iron. However in the present study there was a suggestion that only Tf bound iron is capable of stimulating Ft synthesis since there was an increase as the saturation of Tf present in the culture medium increased up to around the complete saturation of Tf present. Thereafter, when all Tf present was loaded with iron, exposure of proliferating lymphocytes to non-Tf bound iron by further increasing the iron concentration did not show any further increase in Ft levels. Therefore, iron in a "free" form was not able to stimulate Ft synthesis in human lymphocytes. Alternatively, it could also be that Ft synthesis was already maximal with 100% saturation of Tf. These findings agree with the suggestion of Taylor et al (1987) that iron bound specifically to Tf does stimulate Ft synthesis in lymphocytes. Phillips and Rutledge (1984) have also reported that FeTf increased Ft synthesis. Recently Pattanapanyasat (1989b) has reported that addition of increasing levels of FeNTA to PHA-stimulated lymphocytes caused an overall increase in intracellular Ft levels. However in this study the cells were cultured in medium containing 10% FCS. The high concentration of partially saturated bovine Tf in the system used is capable of binding iron from FeNTA, resulting in an increased saturation of this Tf, which could explain the observed increase in cellular Ft content. The increase of Ft levels seen in the present study could be due to the effect of iron on the rate of Ft catabolism. Pattanapanyasat et al (1988) have suggested that iron may decrease the rate of catabolism thereby increasing cellular Ft content in response to iron. It is well established that monocytes show a higher intracellular Ft concentration than lymphocytes, and that these

cells exhibit an increase in Ft synthesis in vitro in response to iron in the culture medium (Summers *et al.*, 1975; Dörner *et al.*, 1983b). Therefore, the increase in Ft levels observed in the present study could be due to the adherent cell population not being separated out.

The apparent insensitivity of lymphocytes to iron present in the medium when in excess of the binding capacity of Tf suggests that these cells have a different control mechanism for Ft synthesis to other types of cells which are known to respond to much higher levels of iron bound to different carriers. It seems possible that in lymphocytes the stimulation of Ft synthesis by iron is normally at its maximum when all the iron present in the medium saturated all the Tf present, after which a balance between synthesis and degradation or conversion to Hs would result in an equilibrium being attained. In the initial process of degradation Ft protein loses its immunoreactivity (Jacobs et al., 1978) and can no longer be measured by the specific assay used here. Stimulation of lysosomal uptake by an increasing Ft load might result in a relatively short life of an individual Ft molecule and could account for the decline in the intracellular Ft levels seen. Worwood et al (1984) showed that peripheral blood mononuclear cells cultured in vitro accumulate Ft very rapidly, reaching a plateau after 10 h incubation, with little or no further increase in Ft occurring after 20 h.

The proliferation results show a good correlation with the Ft levels up to around the full saturation of Tf. At this point the proliferation response was at its maximum. However at higher iron concentrations the proliferation response dropped while cellular Ft content remained constant. This could be interpreted as demonstrating the protective effect of Ft synthesis and the onset of iron toxicity when it fails to match levels of iron donated to cells

3.4.3 The effect of lactoferrin on proliferation and ferritin levels of human lymphocytes

Lf, another member of the Tf family is an important constituent of the secondary granules in neutrophils. During an inflammatory response Lf is released from lysosomes and is readily detectable in inflammatory fluids (Bennett and Skosey, 1977). Thus, Lf might modulate the cellular immune events associated with inflammatory responses.

Previous studies have demonstrated a regulatory role for Lf. It has been shown that human Lf binds to monocytes and macrophages (van Snick and Masson, 1976; Bennett an d Davis, 1981; Goavec *et al.*, 1985; Oria *et al.*, 1988), and that the protein was internalized and degraded by the cells which incorporated iron into Ft (van Snick *et al.*, 1977). Furthermore, it has been claimed that a specific receptor exists on monocytes (Birgens *et al.*, 1983; Bartal *et al.*, 1987). Lf has also been found to play a regulatory role in the immune responses involving lymphocytes. The primary *in vitro* antibody response of murine spleen cells to T-dependent and independent antigens is suppressed in the presence of Lf (Duncan and McArthur, 1981). Recently, Mazurier *et al* (1989) have showed that Lf binds to a specific receptor on lymphocytes, and that these receptors are only expressed on the surface of actively proliferating cells, and unlike Tf receptors they are not expressed intracellularly in resting cells. These claims of the existence of specific receptors for Lf by these studies are open to doubt due to the fact that Lf is highly cationic and binds to a wide range of proteins and other molecules (Hekman, 1971).

There are conflicting reports concerning whether Lf can support proliferation, not only in studies dealing with different types of cells but also among those which dealt exclusively with lymphocytes. In this work neither iron-free nor iron-loaded Lf had any effect on PHA-stimulated human lymphocytes as compared to control cultures with no addition, and was thus found to be incapable of exhibiting Tf-like activity when Previous investigations have shown that colostrum added alone. suppresses mitogen and alloantigen-stimulated lymphocyte proliferation (Crago et al., 1981; Richie et al., 1981). It has been reported that Lf released from phagocytosing neutrophils inhibits transcriptional activity in mixed lymphocyte culture and PHA-stimulated human lymphocytes (Slater and Fletcher, 1987). On the other hand, Mazurier et al (1989) have described Lf as a growth stimulating factor for PHA-stimulated lymphocytes. These latter findings must be taken very cautiously since up till now no evidence whatsoever has been advanced to demonstrate an ability of Lf to donate iron to lymphocytes, despite their claim that lymphocytes exhibit a receptor for Lf. However as mentioned above the nature of the binding sites for Lf is still controversial and many authors believed that they encompass a variety of molecular species (non-specific adsorptive endocytosis) as was reported in the case of macrophages (Regoeczi et al., 1985; Moguilevsky et al., 1985).

The effect of Lf on the proliferation of lymphocytes cultured in the presence of increasing saturation of Tf with iron was examined to see whether Lf, which has a much higher affinity for iron than Tf (Bezkorovainy and Zschocke, 1974), could help in restoring the ability of lymphocytes to proliferate in the presence of excess non-Tf bound iron.

First of all it is worth noting that apoLf significantly inhibited proliferation of lymphocytes as compared to control cultures when added When added to cultures containing lower iron alone(P < 0.005).saturations of Tf (15-75%) the iron-free form of the protein was slightly inhibitory to proliferation of cells. These results are in accordance with the finding of Richie et al (1987), who showed that while iron-free Lf inhibited proliferation of mitogen-stimulated human lymphocytes cultured in serum containing medium, and of mixed lymphocyte cultures, ironsaturated Lf failed to inhibit mitogen-induced proliferation. This suggests that the mechanism of suppression involves the chelating property of Lf. What are the possible mechanisms involved in this inhibitory activity? Is it by mere retention of the metal at the extracellular level or, knowing that some reports have claimed that lymphocyte exhibit specific receptors for the protein (Mazurier et al., 1989) might this involve interference with the intracellular iron pool?

Apo-Lf could compete with Tf in the medium for iron, resulting in the deprivation of cells of iron and thus preventing them from proliferating. By removal of iron from Tf in the extracellular environment, Lf makes iron unavailable, and it is not likely to deliver iron to these cells as discussed above. One could also speculate that Lf competes for the intracellular iron of these cells. Lf might be able to enter the cell and compete for iron, and having bound this iron, there are two possibilities. If the protein has the ability to cross the cell membrane to the extracellular environment, it could shift iron from the intra- to the extracellular milieu, or alternatively it could accumulate within the cell as an inactive form of iron. In both cases, apo-Lf would make iron unavailable to the cells rather than preventing its uptake. It is also possible that the inhibitory effect of Lf is a result of the combination of both mechanisms.

To confirm this view it was of interest to examine the effect of Lf on the intracellular iron reserves of the cells i.e its effect on cellular Ft levels. Apo-Lf was found to decrease intracellular Ft levels compared with the control, while 90%-saturated FeLf had no effect. The effect of Lf on intracellular Ft levels is analogous to the effect of iron chelates on Ft synthesis. Summers *et al* (1975) observed that chelation of iron with DFO markedly reduced lymphocyte Ft synthesis *in vitro*, and suggested that mechanisms maintaining normal basal levels of cellular Ft synthesis are iron-dependent. However, this seems to be in opposition to the earlier findings (see previous section 3.3.4) that excess iron failed to stimulate more Ft synthesis. It could be that under normal iron conditions the response is maximal, and that the presence of a chelator causes a decrease in Ft synthesis while addition of iron increases it.

On the other hand, in cultures containing Tf at a higher degree of saturation (200% saturation) the inhibitory effect of non-Tf bound iron was reversed by the presence of apo-Lf in the culture medium, where it was markedly stimulatory. In sharp contrast, the iron-loaded form of the protein had no effect whatsoever on lymphocyte proliferation either when added alone or when added to lymphocytes cultured in the presence of varying saturations of Tf with iron. In the presence of non-Tf bound iron, apoLf, probably binds this excess and neutralizes its toxic effect on cells, resulting in restoration of cell proliferation. Indeed, Ambruso and Johnston (1981) showed that iron-poor Lf (apparently not completely depleted of iron) inhibited hydroxyl radical production. Similarly, using lipid peroxidation as an indicator of hydroxyl radical activity, Gutteridge *et al* (1981) showed that partially saturated Lf (20%) has antioxidant properties in that it can inhibit the catalytic action of iron salts, but iron saturated Lf had no effect. All this suggests that one of the functions of unsaturated Lf which never occurs completely saturated under normal physiological conditions is to dampen these potentially toxic reactions.

The above results are comparable to the well established function of Lf in human milk as an inhibitor of bacterial growth by chelating iron and preventing its availability to microorganisms (Bullen et al., 1972; Arnold et al., 1977; Bullen, 1987). However another possibility is that the inhibitory effect of apoLf on lymphocyte proliferation could involve more than one mechanism. Lf may possess other non-iron related intrinsic properties which in turn cause other biological effects that could affect lymphocyte proliferation indirectly. Among these effects are inhibition of colony stimulating activity (CSA) production, which results in decreased release of cytokine(s) from monocytes which are responsible for triggering other cells such as T-lymphocytes to release other stimulating factors (Broxmeyer et al., 1976, 1978; Bagby et al., 1981, 1983). This suggests that Lf might mediate its effect through suppression of IL-1 production (Smith et al., 1980; Zucali et al., 1987; 1989), which has numerous immunological activities, including triggering the release of the T-cell growth factor IL-2 from T-lymphocytes

(Dinarello, 1984). Slater and Fletcher (1987) have also reported that Lf inhibits uridine uptake in allogenic mixed lymphocyte reactions, by suppressing the production and release of a soluble factor subsequently identified as IL-2 (Slater and Fletcher, 1987).

In summary at very high iron saturation of Tf, the presence of apo-Lf in the medium contributed to avoiding the deleterious effect of non-Tf bound iron on proliferation probably by sequestering the metal, resulting in prevention of the decline of proliferation seen when high levels of iron were present. At the same time the excess iron loaded Lf and thus prevented there being any inhibitory effect of iron-free Lf, either on lymphocytes directly or on other accessory cells.

3.4.4 The effect of iron saturation of transferrin, FeNTA and FePIH on the expression of Tcell surface markers

In the previous experiments it was shown that excess iron in the presence or the absence of iron-loaded Tf affects T-lymphocyte transformation *in vitro*, It might be possible that iron also affects the immune response by altering the relative populations of T-lymphocyte populations and subpopulations.

To examine this aspect of immunoregulatory properties of iron, evaluation of its effect on the expression of lymphoid surface markers on actively dividing lymphocytes was carried out. Human lymphocytes were chosen to perform this study because of the availability of the monoclonal antibodies to the well known different subset markers of these cells. The phenotypic study reported in the present study, revealed that iron

interferes with expression of two cell surface antigens required for cell activation i.e CD4 and CD8. The presence of iron not bound to Tf caused a decrease in CD4/CD8 ratio, due mainly to depression of the proportion of CD4⁺ cells and a slight increase in the proportion of CD8⁺ cells. The same findings were seen when cells were exposed to iron in the form of FeNTA and in the absence of Tf. However, unlike FeNTA, iron in the form of FePIH did not lower the CD4/CD8 ratio. These findings are relevant to the results of the effect of iron on transformation. A likely explanation of the overall decrease in T-lymphocyte proliferation following polyclonal stimulation when cells were exposed to non-Tf bound iron in the form of FeNTA (see section 3.3.3) is that this iron affects the clonal expansion of CD4⁺ cells which may be more vulnerable to the oxidative stress of iron resulting in oxidative injury to these cells. Allan et al. (1986) have demonstrated that T-lymphocytes are more susceptible to oxidant induced killing than non-T-cells. Therefore the number of the daughter cells generated in these cultures at the end of the incubation period would be less than in cultures where cells were exposed to Tf-bound iron or FePIH. These results are in agreement with findings of Good et al (1986) who reported that Fe^{3+} reduced both the cloning efficiency of human CD4⁺ precursor lymphocytes and the rate of clone growth of the T-cells that did proliferate. Among T-subsets, Bryan et al (1986) have shown a selective effect of high concentrations of chelatable iron on CD4⁺ cells, consisting of a decrease in the proportion of this subpopulation. It has also been reported that fewer helper precursors were present in the spleen of iron-loaded mice (Dialynas et al., 1983). On the other hand, Carvalho and de Sousa (1988) were not able to show any alteration of CD3, CD4, CD8, CD1, CD22, CD10, and HLA-DR surface markers by iron, and the only molecule that was found to be

affected was the CD2. The discrepancy with this study could lie in the difference of the state of cells studied since they assessed non-proliferating cells.

One explanation of the differential effect of high iron concentrations on T-cell subsets is that it may be that in these conditions $CD4^+$ cells accumulate iron intracellularly more rapidly than the $CD8^+$ cells with toxic consequences. Alternatively, the cells may take up the iron at an equal rate, but the CD8 cells may be able to process this iron more efficiently, thus preventing accumulation of excess iron. It has been reported that among T-subsets, helper cells may be relatively sensitive to oxidant injury and that in the presence of DFO injury did not occur and proliferation was enhanced (Sagone *et al.*, 1984).

3.4.5 Proliferation of CCRF-CEM cells in the presence of transferrin, FeNTA, FePIH, and lactoferrin

Among the many differences between normal and transformed cells as studied in cell culture are their requirement for serum in promoting growth and their response to serum growth factors. Many transformed cells have less need for serum for optimal growth than untransformed cells (Sporn and Todoro, 1980). An autocrine growth regulation hypothesis which proposes that malignant cells produce and secrete hormone-like substances that induce further autonomous proliferation, has been postulated as a possible mechanism leading to malignant transformation and was studied in several tumour cell systems (Todaro *et al.*, 1976). Some transformed cells which proliferate in defined serumfree media do not require Tf (Fernandez-Pol, 1978; Zwiller *et al.*, 1982; Basset *et al.*, 1986; Taetle *et al.*, 1985). It is therefore possible that some transformed cells must be able to regulate iron acquisition by a mechanism different from Tf-mediated uptake. Therefore it was of interest to look at the differences in the capacity of CCRF-CEM cells, a T-leukaemic cell line to utilize iron in different forms and to investigate whether these cells behave differently from their untransformed counterparts.

Unlike mouse and normal human lymphocytes in which proliferation in the presence of Tf was found to be a function of the bound iron up to 100% saturation of Tf, CCRF-CEM cells did not show any difference in their proliferative rate at different saturations of Tf up to 100% saturation. Moreover, at higher levels of iron (100-600% saturation), when non-Tf bound iron was present in the culture medium, CCRF-CEM cells were able to achieve good proliferation in contrast to normal mouse and human lymphocytes. The same was true with the chelate FeNTA which unlike normal lymphocytes had the same stimulatory effect on proliferation as Tf. As with normal lymphocytes FePIH was efficient in supporting proliferation of CCRF-CEM cells. These results are in agreement with the findings mentioned above that some iron-containing compounds were able to stimulate the growth of several transformed cell lines other than by the TfR cycle by using haem or haemin (Ward et al., 1984), soluble iron salts (Rudolph et al., 1985), or chelators (Taetle et al., 1985). Iron-binding proteins have been identified in the membrane of neoplastic and neoplasma-derived cell lines. The p97 molecule, a glycoprotein found in melanoma cells which has a high degree of homology with Tf could be an example of a

membrane-located iron binding protein that could participate in such a mechanism (Brown et al., 1982).

The finding that FeNTA was able to allow proliferation of CCRF-CEM cells might also be explained by the possibility that these cells synthesise Tf, in line with the idea that iron from synthetic chelates is taken by the cells after exchange with cell associated Tf (Hemmapardh and Morgan, 1974). FeNTA has been shown to deliver iron into hepatoma cells, and it is known that FeNTA loads apoTf quite efficiently (Bates and Schlabach, 1973) and that contaminating exogenous Tf can affect any experiment studying iron uptake (Brock, 1989). Perhaps the growth stimulation of hepatoma cells seen with FeNTA which was mentioned above is due to this loading of apoTf, a protein which hepatocytes synthesise and secrete. In the present study, this suggestion is supported by the high proliferative background of CCRF-CEM cells cultured in the medium alone, suggesting that these cells are very efficient in taking advantage of the unavoidable iron contamination of the media and using it, perhaps by secreting their own Tf. This hypothesis supports the finding that blockade of TfR of these very same cells (CCRF-CEM) by the monoclonal antibody 42/6 inhibited cell growth and that the addition of ferric complexes or ferrous sulphate to cultures did not overcome the inhibitory effects of the antibody (Trowbridge and Lopez, 1982). However a further report by the same group provided evidence that the addition of soluble iron in the form of FeNTA results in partial reversal of inhibition of cell growth induced by 42/6 antibody in PHA-stimulated lymphocytes (Mendelsohn et al., 1983).

To check this hypothesis Tf synthesis by these cells was examined.

The present study clearly demonstrated the synthesis of Tf by the CCRF-CEM cell line. The Tf produced may act as an autocrine promoter of cellular proliferation in a similar fashion to a number of other growth factors synthesised by malignant cells (Carney et al., 1985). However one may wonder about the necessity of Tf synthesis as a requirement for proliferation of these cells, which had been cultured routinely in serumcontaining medium for several months before performing the study. It could be that unlike unstimulated lymphocytes (see section 2.5.4), the Tf gene in these cells is constantly switched on as they are constantly dividing. The gene could be switched on as a preventive measure in case of shortages in the iron supply since withholding iron is a characteristic of host defence mechanisms against neoplasia (Weinberg, 1984; Letendre, 1985). The same thing applies to cells reaching high densities in cultures where availability of iron becomes a limiting factor. Synthesis of Tf would enable them to continue to grow under these conditions. Kitada and Hays (1985) have described two cloned murine malignant Tlymphoma cell lines which grew in serum-free medium without added Tf. These cells produced a Tf-like activity essential for cell proliferation. Another Tf-like growth factor produced by lymphoma cells has been reported by Morrone et al (1988). This factor supported autocrine growth of these cells, but inhibited normal T-lymphocyte proliferation, in effect serving to provide a substantial growth advantage for the malignant cells. In all the studies mentioned above, Tf synthesised by the cells has marked functional and biochemical similarities to native serum Tf.

If the amount of Tf synthesised by these cells was sufficient to

bind the large amount of FeNTA added to the culture medium and permit growth under those conditions, then it must also be sufficient to bind the trace amounts of endogenous contaminating iron present in the medium. In this case addition of apoTf should not make any difference. However addition of apoTf caused considerable increase in the rate of proliferation. It seems therefore that loading of iron from FeNTA is possible only for cells with an abundant endogenous Tf production sufficient to chelate most of the iron bound to NTA. Hence the existence of an alternative pathway for FeNTA acquisition by these cells, as well as that of the TfR pathway seems to be quite likely. Mouse leukaemic cells exhibit a Tfindependent iron transport system, which appears to be involved in the ability of these cells to take up chelate iron and use it to promote DNA synthesis and cell growth (Taetle et al., 1985). This suggests that this uptake must occur through a regulatory transport system and not by mere diffusion of the chelator. The finding of a siderophore-like growth factor secreted by SV-40-transformed 3T3 mouse cells provides evidence for the existence of a specific low molecular weight transport system for iron and other trace metals (Fernandez-Pol, 1978). This factor was shown to induce DNA synthesis and proliferation of the cells.

The finding that these cells managed to grow at high levels of iron compared to their normal counterpart could suggest the hypothesis that unlike normal lymphocytes, these cells have the ability to synthesise increasing amounts of Ft in response to non-Tf bound iron. However this hypothesis could not be substantiated since CCRF-CEM cells were found to have extremely low intracellular Ft levels compared to their normal counterpart. The reason of this difference could lie in the differences in the degree of proliferation. It is well known that iron in these proliferating cells is mainly needed for the functioning of the enzyme ribonucleotide reductase. Therefore, the amount of iron used by the cell will be proportional to the uptake of thymidine which directly measures DNA synthesis, and since the proliferative activity of CCRF-CEM cells was much higher than that of PHA-stimulated normal lymphocytes (~12X at optimum proliferation on a cell to cell basis; see sections 3.3.2 and 3.3.9), it is likely then that the former were able to use larger amounts of iron to sustain this high blastogenic activity. The intracellular low molecular weight iron pool which is supposed to be the precursor of incorporation into Ft (Jacobs, 1977) and which could mediate cytotoxicity is probably very limited. This could explain why these cells were not synthesising as much Ft as their normal counterpart and could also account for the resistance of these cells to the oxidative injury in the presence of high iron concentration unlike normal lymphocytes. It could also be that these cells synthesise mainly H-rich Ft which could not be detected by the assay used in this study. It has been shown that transformed cells synthesise H-rich Ft predominantly (Dörner et al., 1983a). This type of Ft has been shown to uptake iron more readily than L-rich Ft at pH 7.0 (Levi et al., 1989), probably because of the presence of a ferroxidase site on the H-chain.

As with normal proliferating T-lymphocytes, Lf was found not to have any effect on the growth of CCRF-CEM cells when added alone whether it was "iron-free" or iron-loaded. However, Hashizume *et al.*(1983) have reported that several human, but not mouse B and T-cell lines proliferated in serum-free medium supplemented with Lf. However in their study, B-cell lines generally grew better than T-cell lines, and they found that growth of CCRF-CEM cells in Lf medium was only 20% over that found in serum-containing medium. This figure represents nearly the proliferative background of these cells cultured in the medium alone, found in the present study.

On the other hand, CCRF-CEM cells unlike their normal counterparts were found to be able to proliferate in the presence of FeNTA. This could also be explained by the possibility that these cells may synthesis abundant amounts of their own Tf (see below) or more likely could be due to a low-affinity iron-uptake system not involving Tf which might operate in these cells as has been reported previously (Basset *et al.*, 1986).

GENERAL DISCUSSION

The work reported in this thesis has attempted to define more closely the role of iron in the proliferation of lymphocytes in vitro with particular reference to the intracellular biochemical events occurring Few have looked at iron metabolism in following its uptake. lymphocytes, despite its importance in understanding how lymphocytes react to different degrees of extracellular iron availability. This study has demonstrated that the availability of Tf-bound iron in the extracellular medium and in particular the degree of Tf saturation has a critical effect on lymphocyte proliferation in response to mitogens. Iron donated in this form was used by the cell and permitted an increase in cellular metabolism as the largest proportion of iron donated to the cell in this form was found in a subcellular fraction probably consisting of enzymes and haem-containing proteins. However, when cells were cultured in the presence of high concentrations of iron (beyond the level that saturates all Tf present in the medium) proliferation was inhibited probably by affecting preferably the helper subsets, suggesting that some mechanism may exist for the uptake of the unbound iron into these cells, resulting in toxic consequences. Since cells cultured in the presence of 100% saturated Tf exhibit optimum transformation, this rules out the possibility that the inhibitory effect of iron seen at those high concentrations is a result of excess accumulation of iron taken up from rapidly endocytosed diferric Tf as previously suggested (Brock 1981). On the other hand, the effect of non-Tf bound iron in the form of FeNTA was clearly neutralized by the presence of apoLf in the medium.

It is well known that iron enhances the formation of toxic oxygen free radicals (Biemond et al., 1988). The occurrence of damage due to oxygen free radicals depends on the balance between the amount of radicals which are formed and the amount of protection at the same location. The generation of these toxic oxygen intermediates, in particular hydroxyl radicals responsible for the peroxidation of membrane lipids might decrease in the presence of iron binding proteins because catalytically active coordination sites of the metal diminish. Plasma Tf has considerable reserves for coping with increasing amounts of incoming iron, but these may be exceeded in certain pathological conditions. Cultures containing the FeNTA complex in the present study perhaps provide a system comparable to the non-specific iron found in thalassaemic sera. In severe iron-overloaded haemochromatotic patients, Tf is completely saturated and as a result an abnormal serum iron fraction appears as low molecular weight iron, which has been reported to be present, loosely bound to a variety of serum protein such as albumin or non-specifically bound to Tf itself (reviewed by Hershko and Peto, 1987). Iron in the circulation in excess of that required to saturate available Tf is associated with increased susceptibility to infection. This is not only due to the fact that such iron complexes are more readily available to microbial-iron scavenging mechanisms than Tf-bound iron (Weinberg, 1978), but also to the possibility that this iron might interact with immune cells and affect their function. Because lymphocytes appear to lack the buffering effect of increased Ft synthesis against the excessive entrance of iron into the cell as discussed in section 3.4.2, they are particularly vulnerable to iron-dependent oxidative injury. As mentioned above the presence of non-Tf-bound iron, but not iron-saturated Tf,

tends to decrease the response of mouse and human lymphocytes to mitogens. Similarly a decreased response of lymphocytes to PHA and Con A was reported in thalassaemia patients with very high iron levels (Munn *et al.*, 1981; Dwyer *et al.*, 1987). Lymphocytes in liver, a tissue recognised for its high iron content, have been reported to respond poorly in the mixed lymphocyte reaction (Mukhopadhyay *et al.*, 1978). This observation raises the possibility that *in vivo* iron can also cause inhibition of lymphocyte responses under normal conditions. All this is in agreement with the findings of Kraut and Sagone (1981) who demonstrated that oxidant injury to lymphocytes impairs both membrane and cellular function such as their ability to form E-rosettes and impaired cap formation after binding of Con A as well as impaired transformation and cytotoxic capacity.

Iron loading can place a limit on the degree of activation of lymphocytes apparently by reduction of functional helper T cells. The CD4⁺ subset plays a crucial role in the initial stages of development of immune responses. They are particularly important in regulating either directly or indirectly the functional activity of various other immune cells, including B lymphocytes, monocytes/macrophages, cytotoxic T lymphocytes, and natural killer cells. Therefore, the quantitative defect of these vitally important cells and the subsequent functional defect that this might cause in specific immune surveillance is of great physiological importance and may lead to immunosuppression. Thus, the decrease in the ratio of CD4/CD8 seen when high levels of iron were present in the culture medium suggests that this could be a handicap to the infected host as the decrease in proliferating cells, especially helper cells, could not help to combat the effect of the increased free iron available to

microorganisms. This is particularly relevant to iron overloaded patients in whom increased susceptibility to infection has been reported (idiopathic haemochromatosis or transfusional iron-overload thalassaemia) (Weinberg, 1978). As many as 12% of idiopathic haemochromatotic patients were reported to die with pneumonia, and severe bacterial infections are an important cause of morbidity and mortality in thalassaemia major (Hershko et al., 1988). Cell mediated immunity appears to be particularly impaired in uraemic subjects who have received previous blood transfusions (Watson et al., 1979). Allograft survival following renal transplantation is significantly prolonged in such subjects (Vincenti et al., 1978; Watson et al., 1979). The degree of improvement of kidney survival has been directly related to number of transfusions received (Opelz and Persijin, 1981), which illustrates the suppressed status of their immune system and supports de Sousa's view that this is due to an iron-related immunosuppression (de Sousa, 1983). However, other authors have reported that this is not just due to iron overload (Woodruff and van Rood, 1983).

The findings reported in this study are in line with many clinical investigations carried out on iron-overload related disease. Kapadia *et al* (1980) reported a decrease in number of T-helper cells and concomitant increase in B-cells associated with iron overload. Abnormally low numbers and functionally defective CD4⁺ cells have been reported in thalassaemia intermedia (Guglielmo *et al.*, 1984), and in thalassaemia major (Grady *et al.*, 1985; Dwyer *et al.*, 1987; Pardalos *et al.*, 1987). In beta-thalassaemic children with very high transferrin saturation, the

ratio of CD4/CD8 was low compared to control, this decrease being due primarily to a reduction of CD4⁺ cells and partly to an increase of the CD8⁺ cells (Pardalos et al., 1987). Similar findings as far as T-cell subpopulations are concerned have also been reported by others in betathalassemia patients (Neri et al., 1984, Kessler et al., 1983). In thalassaemia major the impaired production of immunoglobulins by Bcells was attributed to a defective T-helper cell population and not to the non-T-cell population (Nualart et al., 1987). Iron overload has also been reported to influence the expansion of the CD8+ cell population in vivo. In patients with thalassaemia major, whether splenectomized or not, an increase in the numbers of CD8+ cells has been reported (Grady et al., 1987). Other indirect evidence in favour of iron overload influencing expansion of the CD8+ cell population in vivo comes from the observation of a negative correlation between the amount of DFO received and the numbers of CD8+ cells present in thalassaemic patients (Dwyer et al., 1987).

The reduction in the number of cells that are proliferating may be due to either cell death (cytotoxicity) as discussed above or to an impairment of proliferative capacity (suppression), or a combination of both. The catalytic effect of ionic iron upon lipid peroxidation could lead to alteration of the cell membrane and subsequently to cell death. It could also result in the generation of suppressor factors, which inhibit lymphocyte proliferation and may trigger the generation of antigen-nonspecific suppressor cells (Fischer *et al.*, 1980). Conversely, the free oxygen radical species generated by iron might interact with suppressor cells to produce inhibitory factors. An example would be the interaction of these highly reactive species with arachidonic acid metabolism
resulting in prostaglandin synthesis which is known to modify T-cell functions and in particular to activate suppressor mechanisms (Hemler *et al.*, 1979; Dinarello *et al.*, 1983). This could explain the slight increase in the proportion of the CD8⁺ subset when cells where exposed to non-Tf bound iron in the form of FeNTA. All this points to specific responses of T-cell subsets to changes in the extracellular iron concentration. However the mechanism(s) whereby iron seems to have a selective influence on T-populations remain unclear.

In inflammatory diseases local accumulation of iron occurs and may cause damaging free radical reactions (Blake *et al.*, 1983) which may similarly affect the immune system. Indeed uptake of ionic iron as the chelate FeNTA by rabbit synovial fibroblasts was accompanied by increased production of latent collagenase and prostaglandin G_2 (Okazaki *et al.*, 1981). Addition of DFO prevents the whole process. On the other hand, the combination of the decrease in T helper/inducer cells and increase in the T suppressor/cytotoxic cells may be a protective mechanism as part of the normal accelerated acute phase response may be avoided. However, prolonged depression of T cell immune response could be a handicap to the host against microbial invaders and may also be a major factor in the development of malignancies as discussed above.

Lf can prevent potentially toxic effects of iron as a result of its ability to dampen tissue-damaging free radical reactions. When neutrophils phagocytose bacteria, particles, or immune complexes, they show a rapid burst of oxygen uptake associated with the production of oxygen derived species such as superoxide O_2^{--} and hydrogen peroxide H_2O_2 (Babior, 1978). In the presence of trace amounts of iron, O_2^{--} and

 H_2O_2 react together to form the hydroxyl radical OH' which participates in the bacterial killing mechanism but can also attack and damage almost all biological molecules and induce lipid peroxidation (reviewed by Halliwell, 1989; Hershko, 1989). In contrast to Tf, Lf can bind iron tightly at acid pH values and thus can function well in tissue sites in which the pH has been lowered by microbial and leucocytic metabolism. When released into the phagocytic vacuole, Lf can be expected to sequester any free iron and thus limit the damage caused to surrounding It might therefore be possible that secretion of Lf from tissues. neutrophils in large amounts during phagocytosis might be a mechanism by which surrounding tissues are protected against oxidative injury. This view is consistent with the report that iron-free or partially saturated Lf inhibits lipid peroxidation (Gutteridge et al., 1981; Sibille et al., 1987). This could be particularly important during inflammation, since synovial fluid from patients with inflammatory disease contains free iron and large numbers of phagocytic cells infiltrating at the site of inflammation (Wong et al., 1981; Gutteridge et al., 1981), at which radicals have been implicated in the pathogenesis of inflammation (Sacks et al., 1978). Lf is found extracellularly at sites of inflammation and its levels have been reported to correlate directly with the degree of inflammation of rheumatoid joints (Bennett and Skosey, 1977). There is little data about iron saturation of Lf in neutrophil granules, but it is usually considered to be iron free (van Snick et al., 1974; Bullen. 1981).

During an inflammatory response there must be a balance between inducing factors such as endotoxins and other infective agents or complement breakdown products, which provide a positive stimulus to mononuclear cells, and negative regulation factors. Lf could be one of the latter. A possible extracellular role could be attributed to Lf when neutrophils are activated, as it could provide a negative feedback control of lymphocyte growth possibly by depriving them of iron in the medium, which is in keeping with other evidence that Lf inhibits T-cell function (Bagby et al., 1981) and antibody production (Duncan and McArthur, 1981). Lf could be particularly important in neutralizing the harmful effect of local iron excess in inflammation as discussed above. On the other hand, iron overloaded patients could be facing a double problem; iron availability to invading microorganisms from one side, and the inefficiency of the protective effect of Lf which under these conditions became saturated with iron. In these cases, Lf might obtain iron from the iron rich milieu, which contains fully saturated Tf and low molecular weigh iron compounds. In addition pinocytosed Tf which in contrast to Lf might also release bound iron readily as the pH drops to levels easily achieved in the phagocytic vacuole (Lestas, 1976) could also contribute to loading Lf with iron within the phagocytic vacuole of neutrophils.

Many areas of research in this field still need to be examined, including whether Lf is internalized and cycled through cells, and if so whether it cycles with Tf or by a different route. It would also be of interest to examine the effect of Lf on iron uptake by cells. The recent finding of Furmanski *et al* (1989) of the existence of several isoLfs exerting different functions might open an exciting chapter in understanding the various and complicated functions of this protein and could lead the way to many interesting investigations. The work presented here has tried to establish the effect of different forms and levels of iron on mouse and human cells *in vitro*. It would be of interest to extend this work to an *in vivo* system using an experimental animal model which could probably provide a more complete picture of the effect of different levels of iron on lymphocytes and the immune system as a whole, and also a more adequate explanation of these effects.

The present study showed that the lymph node macrophages, which increased in number upon immunological activation have the capacity to synthesise Tf. This Tf could augment the iron supply of rapidly proliferating lymphocytes. For locally-synthesised Tf to be of benefit in immunological activation it would probably need to contain iron. Although hepatic Tf is generally thought to be synthesised and released in the iron-free form (Hemmapardh and Morgan, 1974) it has been hypothesized that iron released from macrophages is transported by Tf of endogenous origin (Haurani and Ballas, 1984), and it is therefore of particular interest that iron released by macrophages is increased by treatment in vivo with Corynebacterium parvum (Alvarez-Hernandez et al., 1986) or in vitro with γ -IFN (Taetle and Honeysett, 1988). Therefore in the future it would be of prime importance to investigate whether Tf synthesised by macrophages contains iron and to establish whether Tf synthesised by these cells can enhance lymphocyte proliferation which would open an area of possible immunoregulatory activity. This would in turn fundamentally affect wider issues such as the controversial question of whether iron deficiency affects immune function, and whether iron depletion is beneficial to immune function and resistance to infection. Further studies are also necessary in order to

examine the effect of various agent that would affect Tf synthesis by these cells. Conversely, CCRF-CEM, a T-leukaemic cell line has been found to produce Tf. These findings support the hypothesis (Fernandez-Pol, 1983) that growth modulating factors like siderophore-like growth factors and Tf-like activity synthesised and secreted by some transformed cells may enable tumour cells to take advantage of local conditions, and decrease their dependence on serum factors (Tf in this case), and compete with normal cells for specific trace metal ions. Moreover the findings of this study suggests that these cells, unlike their normal counterparts exhibit Tf-independent iron transport which appears to be involved in the ability of these cells to take up iron in a variety of forms and to use it efficiently for cell growth. Recently, Sturrock, et al., 1990) have demonstrated that uptake of non-Tf bound iron by HeLa cells from various chelates was Tf-independent. This iron was used by the cells as most of it was recovered in haem and Ft. These findings lend some indirect support to the frequent reports that patients with iron overload-related diseases are vulnerable to development of neoplasia (reviewed by Weinberg, 1985), as non Tf-bound iron probably exists in these patients.

196

REFERENCES

- Aisen, P. and Leibman, A. (1972). Lactoferrin and transferrin: A comparative study. *Biochim. Biophys Acta.*, 257:314.
- Aisen, P. and Listowsky, I. (1980). Iron transport and storage proteins. Ann. Rev. Biochem., 49:357.
- Aisen, P., Leibman, A., and Zweier, J. (1978). Stoichiometric and site characteristics of the binding of iron to human transferrin. J. Biol. Chem., 253:1930.
- Aldred, A.R., Dickson, P.W., Marley, P.D., and Schreiber, G. (1987). Distribution of transferrin synthesis in brain and other tissues in the rat. J. Biol. Chem., 262:5293.
- Allan, I.M., Lunec, J., and Bacon, P.A. (1986). Selective lymphocyte killing by reactive oxygen species (ROS). Agents and Actions, 19:5.
- Alvarez-Hernandez, X. and Loria, A. (1980). Antiferritin labelled with the Bolton-Hunter reagent. *Clin. Chem.*, **26**:1916.
- Alvarez-Hernandez, X., Felstein, M.V., and Brock, J.H. (1986). The relationship between iron release, ferritin synthesis and intracellular iron distribution in mouse peripheral macrophages. Evidence for a reduced level of metabolically-available iron in elicited macrophages. *Biochim. Biophys. Acta*, **886**:214.
- Alvarez-Hernandez, X., Piedra, J., Cordoba, M.S., Lopez-Karpovitch, X., and Cano, R. (1981). Ferritina serica en mujeres y varones. Valores de referencia. *Rev. Invest. Clin.*, 33:13.
- Ambruso, D.R. and Johnston, R.B.(1981). Lactoferrin enhances hydroxyl radical production by human neutrophils, neutrophil particulate fractions, and an enzymatic generating system. J. Clin. Invest., 67:352.
- Amouric, M., Marvaldi, J., Pichon, J., Bellot, F., and Figarella,C. (1984). Effect of lactoferrin on the growth of human colon adenocarcinoma cell line-comparison with transferrin. *In Vitro*,

20:543.

- Anderson, B.F., Baker, H.M., Dodson, E.J., Norris, G.E., Rumball, S.V., Waters, J.M., and Baker, E.N. (1987). Structure of human lactoferrin at 3.2 Å resolution. *Proc. Natl. Acad. Sci. USA*, 84:1768.
- Anderson, B.F., Baker, H.M., Norris, G.E., Rice, D.W., and Baker, E.N. (1989). Structure of human lactoferrin: crystallographic structure analysis refinement at 2.8 Å resolution. J. Mol. Biol., 209:711.
- Anderson, W.L., Chase, C.G., and Tomasi, T.B., Jr. (1982). Transferrin support of stimulated lymphocytes. In Vitro, 18:766.
- Aparicio, P., Alonso, J.M., Toribo, M.L. Gutierrez, L., Pezzi, L., and Martinez-A, C. (1989). Differential growth requirements and effector functions of α/β and γ/δ human T cells. *Immunol. Rev.*, 111:5.
- Apte, S.V. and Brown, E.B. (1969). Effects of plasma from pregnant women on iron absorption by the rat. *Gastroenterology*, **57**:126.
- Archibald, (1983). Lactobacillus plantarum, an organism not requiring iron. Fed. Eur. Microbiol. Soc. Lett., 19:29.
- Arnold, R.R., Cole, M.F., and McGhee, J.R. (1977). A bactericidal effect for human lactoferrin. *Science*, **197**:263.
- Arosio, P. (1989). The molecular biology of iron-binding proteins. In *Iron in Immunity, Cancer and Inflammation* (Eds M. de Sousa and J.H. Brock), pp. 55-79. John Wiley & Sons, Chichester.
- Arosio, P., Levi, S., Gabri, E., Ruggeri, E., Iacobello, R., Albertini, A., and Cavanna, F. (1984). Heterogeneity of ferritin.
 II. Immunological aspects. In *Ferritins and Isoferritins as Biochemical Markers* (Eds A. Albertini, P. Arosio, E. Chiancone, and J.W. Drysdale), pp. 33-46. Elsevier Science Publishers, Amsterdam.
- Aruoma, O.I., and Halliwell, B. (1987). Superoxide-dependent and

ascorbate-dependent formation of hydroxyl radicals from hydrogen peroxide in the presence of iron. Are lactoferrin and transferrin promoters of hydroxyl-radical generation? *Biochem. J.*, 241:273.

- Atkin, C.L., Thelander, L., Reichard, P. (1973). Iron and free radical in ribonucleotide reductase. Exchange of iron and Mossbauer spectroscopy on the protein B_2 subunit of the *Escherichia coli* enzyme. *J. Biol. chem.*, **248**:7464.
- Aziz, N. and Munro, H.N. (1986). Both subunits of rat liver ferritin are regulated at a translational level by iron induction. *Nucleic Acids Research* 14:915.
- Aziz, N. and Munro, H.N. (1987). Iron regulates ferritin mRNA translation through a segment of its 5' untranslated region. *Proc. Natl. Acad. Sci. USA*, 84:8478.
- Babior, B.M., Kipnes, R.S., and Curnutte, J.T. (1978). The production by leucocytes of superoxide, a potential bactericidal agent. J. Clin. Invest., 52:741.
- Bacon, R.B., Park, C.H., Brittenham, G.M., O'Neil, R., and Tavill, A.S. (1985). Hepatic mitochondrial oxidative metabolism in rats with chronic dietary iron overload. *Hepatology*, 5:789.
- Bagby, G.C., McCall, E., and Layman, D.L. (1983). Regulation of colony-stimulating activity production of fibroblasts, mononuclear phagocytes and lactoferrin. J. Clin. Invest., 71:340.
- Bagby, G.C., Rigas, V.D., Bennett, R.M., Vanderbark, A.A., and Garewal, H.S. (1981). Interaction of lactoferrin, monocytes and Tlymphocyte subsets in the regulation of steady-state granulopoiesis in vitro. J. Clin. Invest., 68:56.
- Baggiolini, M., DeDuve, C., Masson, P.L., and Heremans, J.F. (1970). Association of lactoferrin with specific granules in rabbit heterophil leukocytes. J. Exp. Med., 131:559.
- Bailey, S., Evans, R.W., Garrat, R.C., Gorinsky, B., Hasnain,
 S., Horsburgh, C., Jhoti, H., Lindley, P.F., Mydin, A., Sarra,
 R., and Watson, J.L. (1988). Molecular structure of serum

transferrin at 3.3 Å resolution. Biochemistry, 27:5804.

- Bailey-Wood, R., White, G.P., and Jacobs, A. (1975). The use of Chang cells cultured *in vitro* for the investigation of cellular iron metabolism. Br. J. Exp. Pathol., 56:358.
- Bain, J.A. and Deutsch, H.F. (1948). Separation and characterization of conalbumin. J. Biol. Chem., 172:547.
- Baker, E. and Morgan, E.H. (1969). The kinetics of the interaction between rabbit transferrin and reticulocytes. *Biochemistry*, 8:1133.
- Baker, E., Shaw, D.C., and Morgan, E.H. (1968). Isolation and characterization of rabbit serum and milk transferrins. Evidence for a difference in sialic acid content only. *Biochemistry*, 7:1371.
- Baker, E.N., Rumball, S.V., and Anderson, B.F. (1987). Transferrins: insights into structure and function from studies on lactoferrin. *Trends Biochem. Sci.*, **12**:350.
- Bakker, G.R. and Boyer, R.F. (1986). Iron incorporation into apoferritin. The role of apoferritin as ferroxidase. J. Biol. Chem., 261:13182.
- Bakkeren, D.L., de Jeu-Jaspars, C.M.H., van der Heul, C., and van Eijk, H.G. (1985). Analysis of iron-binding components in the low molecular weight fraction of rat reticulocyte cytosol. *Int. J. Biochem.*, 17:925.
- Baliga, B.S., Kuvibidila, S., and Suskind., R.M. (1981). The cytotoxic capacity of spleen and peritoneal cells from iron-deficient animals. *Fed. Proc*, **40**:918.
- Ballantyne, G.H. (1984). Rapid drop in serum iron concentration as a host defense mechanism: a review of experimental and clinical evidence. Am. Surg., 50:405.
- Bartlett, G.R. (1976). Iron nucleotides in human and rat red cells. Biochem. Biophys. Res. Comm., 70:1063.
- Bartal, L., Padeh, S., and Passwell, J.H. (1987). Lactoferrin

inhibits prostaglandin E_2 secretion by breast milk macrophages. *Pediatric Res.*, **21**:54.

- Basset, P., Quesneau, Y., and Zwiller, J. (1986). Iron-induced L1210 cell growth: evidence of a transferrin-independent iron transport. *Cancer Res.*, **46**:1644.
- Bates, G.W. and Schlabach, M.R. (1973). The reaction of ferric salts with transferrin. J. Biol. Chem., 248:3228.
- Bates, G.W., and Schlabach, M.R. (1975). The nonspecific binding of Fe^{3+} to transferrin in the absence of synergistic anions. *J.Biol. Chem.*, **250**:2177.
- Batey, R.G., Lai Chung Fong, P., Shamir, S., and Sherlock, S. (1980). A non-transferrin bound serum iron in idiopathic hemochromatosis. *Dig. Dis. Sci.*, **25**:340.
- Beach, R.L., Popiela, H., Festoff, B.W. (1983). The identification of neutrophilic factor as a transferrin. *FEBS Lett.*, **156**:151.
- Beaumont, C., Jain, S.K., Bogard, M., Nordman, Y., and Drysdale, J. (1987). Ferritin synthesis in differentiating Friend erythroleukemic cells. J. Biol. Chem., 262:10619.
- Beinert, H. (1973). Development of the field and nomenclature. In Iron Sulphur Proteins (Eds. W. Lovenberg), pp. 1-36. Academic Press, New York.
- Bennett, R.M. and Davis, J. (1981). Lactoferrin binding to human peripheral blood cells: an interaction with B-enriched population of lymphocytes and subpopulation of adherent mononuclear cells. J. Immunol, 127:1211.
- Bennett, R.M. and Mohla, C. (1976). A solid-phase radioimmunoassay for the measurement of lactoferrin in human plasma: variations with age, sex and disease. J. Lab. Clin. Med., 88:156.
- Bennett, R.M. and Skosey, J.L. (1977). Lactoferrin and lysozyme levels in synovial fluid. Differential induces of articular inflammation

and degradation. Arthri. Rheum., 19:84.

- Berridge, M.J. and Irvine, R.F. (1984). Inositol triphosphate: a novel messenger in cellular signal transduction. *Nature*, **312**:315.
- Besancon, F., Bourgeade, M-F, and Testa, U. (1985). Inhibition of transferrin receptor expression by interferon- α in human lymphoblastoid cells and mitogen-induced lymphocytes. J. Biol. Chem., **260**:13074.
- Bezkorovainy, A. and Zschocke, R.H. (1974). Structure and function of transferrins. I. Physical, chemical and iron binding properties. *Arzneimittel Forschung*, 24:476.
- Biemond, P., Swaak, A.J.G., van Eijk, H.G., and Koster, J.F. (1988). Superoxide dependent iron release from ferritin in inflammatory diseases. *Free Radical Biol. Med.*, 4:185.
- Birgens, H.S., Hansen, N.E., Karle, H., and Kristensen, L.Ø. (1983). Receptor binding of lactoferrin by human monocytes. Br. J. Haematol., 54:383.
- Bjorkman, P.J., Saper, M.A., Samraoui, B., Bennett, W.S., Strominger, J.L., and Wiley, D.C. (1987). The foreign antigen binding site and T cell recognition regions of class I histocompatibility antigens. *Nature*, **329**:512.
- Blake, D. R., Hall, N.D., Bacon, P.A., Dieppe, P.A., Halliwell,
 B., Gutteridge, J.M.C. (1983). Effect of a specific iron chelating agent on animal models of inflammation. Ann. Rheum. Dis., 42:89.
- Bloch, B., Popovici, T., Levin, M.J., Tuil, D., and Kahn, A. (1985). Transferrin gene expression visualized in oligoendrocytes of the rat brain by using *in situ* hybridization and immunohistochemisty. *Proc. Natl. Acad. Sci. USA*, 82:6706.
- Bogdanikowa, B. and Grabowski, R. (1972). Serum proteins present in edema fluids. *Clin. Chim. Acta*, **36**:351.
- Bomford, A., Conlan-Hollingshead, C., and Munro, H.N. (1981). Adaptive responses of rat liver isoferritins to iron administration.

Changes in subunit synthesis, isoferritin abundance, and capacity for iron storage. J. Biol. Chem., 256:948.

- Bomford, A., Young, S., and Williams, R. (1986). Intracellular forms of iron during transferrin iron uptake by mitogen-stimulated lymphocytes. *Br J. Haematol.*, **62**:487.
- Bomford, A., Young, S.P., Nouri-Aria, K., and Williams, R. (1983). Uptake and release of transferrin and iron by mitogenstimulated human lymphocytes. *Br J. Haematol.*, **55**:93.
- Bowman, B.H., and Yang, F.(1987). DNA sequencing and chromosomal locations of human plasma protein genes. In *The Plasma Proteins* (Ed F.W. Putnam), Vol. V, pp. 1-48. Academic Press. New York.
- Bowman, B.H., Yang, F., and Adrian, G.S. (1988). Transferrin: evolution and genetic regulation of expression. *Adv. Genet.*, **25**:1.
- Boyd, D., Vecoli, C., Belcher, D.M., Jain, S.K., and Drysdale, J.W. (1985). Structural and functional relationships of human ferritin H and L chains deduced from cDNA clones. J. Biol. Chem., 260:11761.
- Braughler, M.J., Chase, R.L., and Pregenzer, J.F. (1988). Stimulation and inhibition of iron-dependent lipid peroxidation by desferrioxamine. *Biochem. Biophys. Res. Commun.*, **153**:933.
- Brenner, M.B., and Trowbridge, I.S., and Strominger, J.L. (1985). Cross-linking of human T cell receptor proteins: association between the T cell idiotype β subunit and the T3 glycoprotein heavy subunit. *Cell*, **40**:183.
- Brock, J.H. (1980). Lactoferrin in human milk: its role in iron absorption and protection against enteric infection in the newborn infant. Arch. Dis. Child., 55:417.
- Brock, J.H. (1981). The effect of iron and transferrin on the response of serum-free cultures of mouse lymphocytes to concanavalin A and lipopolysaccharide. *Immunology*, **43**:387.

- Brock, J.H. (1985). The transferrins. In *Metalloproteins* (Ed P.M. Harrison), Part 2, pp. 183-262, MacMillan. London.
- Brock, J.H. (1989). Problems with iron and iron-binding proteins in tissue culture. In *Iron in Immunity, Cancer and Inflammation* (Eds. M. de Sousa and J.H. Brock), pp. 399-408. John Wiley and Sons, Chichester.
- Brock, J.H. and Arzabe, F.R. (1976). Cleavage of diferric bovine transferrin into two monoferric fragments. *FEBS Lett.*, **69**:63.
- Brock, J.H. and Mainou-Fowler, T. (1983). The role of iron and transferrin in lymphocyte transformation. *Immunol. Today*, **4**:347.
- Brock, J.H. and Rankin, M.C. (1981). Transferrin binding and iron uptake by mouse lymph node cells during transformation in response to concanavalin A. *Immunology*, **43**:393.
- Brock, J.H. and Stevenson, J. (1987). Replacement of transferrin in serum-free cultures of mitogen-stimulated mouse lymphocytes by a lipophilic iron chelator. *Immunol. Lett.*, **15**:23.
- Brock, J.H., Mainou-Fowler, T., and Webster, L.M. (1986). Evidence that transferrin functions only as an iron donor in promoting lymphocyte proliferation. *Immunology*, **57**:105.
- Brockman, R.W., Shaddix, S., and Stringer, V. (1971). Effects of Ara-C and ribonucleotide reductase inhibitors on DNA synthesis in sensitive and resistant lines of leukaemia L1210. Proc. Am. Assoc. Cancer Res., 12:67.
- Brottier, P., Boumsell, L., Gelin, C., and Bernard, A. (1985). T cell activation via CD2 (Tgp50) molecules: accessory cells are required to trigger T cell activation via CD2-D66 plus CD2-9.6/T111 epitopes. *J.Immunol.*, 135:1624.
- Brown, E.B., Okada, S., Awai, M.T., and Chipman, B. (1975). In vivo evidence for the functional heterogeneity of transferrin-bound iron III. Studies of transferrin at high and low iron saturation. J. Lab. Clin. Med., 85:567.
- Brown, J.P., Hewick, R.M., Hellstrom, I., Hellstrom, K.E.,

Doolittle, R.F., and Dreyer, W.J. (1982). Human melanomaassociated antigen p97 is structurally and functionally related to transferrin. *Nature*, **296**:171.

- Brown, N.C., Eliasson, R., Reichard, P., and Thelander, L. (1969). Spectrum and iron content of protein B₂ from ribonucleoside disulphate reductase. *Eur. J. Biochem.*, **9**:512.
- Broxmeyer, H.E. (1989). Iron-binding proteins and the regulation of haematopoietic cell proliferation/differentiation. In *Iron in Immunity, Cancer and Inflammation* (Eds. M. de Sousa and J.H. Brock), pp. 199-221. John Wiley and Sons, Chichester.
- Broxmeyer, H.E., Baker, F., and Galbraith, P.R. (1976). *In vitro* regulation of granulopoiesis in human leukemia. Application of an assay for colony inhibiting cells. *Blood*, 47:389.
- Broxmeyer, H.E., Lu, L., and Bognacki, J. (1983). Transferrin, derived from an OKT8-positive subpopulation of T lymphocytes, suppresses the production of granulocyte-macrophage colonystimulatory factors from mitogen-activated T lymphocytes. *Blood*, 62:37.
- Broxmeyer, H.E., Smithyman, A., Eger, R.R., Meyers, P.A., and de Sousa, M. (1978). Identification of lactoferrin as the granulocytederived inhibitor of colony stimulating activity (CSA)-production. *J.Exp. Med.*, 148:1052.
- Bryan, C.F., Leech, S.H., and Bozelka, B. (1986). The immunoregulatory nature of iron. II. Lymphocyte surface marker expression. J. Leukoc. Biol., 40:589.
- Bryan, C.F., Nishiya, K., Pollack, M.S., Du pont, B., and de Sousa (1981) Differential inhibition of the MLR by iron: association with *HLA* phenotype. *Immunogenetics*, **12**:129.
- Bullen, J.J. (1981). The significance of iron in infection. *Rev. Infec. Dis.*, **3**:1127.
- Bullen, J.J. (1987). Iron and the antibacterial function of

polymorphonuclear leucocytes. In *Iron and Infection*.(Eds. J.J. Bullen and G. Griffiths). pp. 211-242. John Wiley and Sons, Chichester.

- Bullen, J.J., Rogers, H.J., and Leigh, L. (1972). Iron-binding proteins in milk and resistance to *Escherichia coli* infection in infants. *Br. Med. J.*, 1:69.
- Cairo, G., Bardella, L., Schiaffonati, L., Arosio, P., Levi, S., and Bernelli-Zazzera, A. (1985). Multiple mechanisms of ironinduced ferritin synthesis in HeLa cells. *Biochem. Biophys. Res. Commun.*, 33:314.
- Cairo, G., Vezzoni, P., Bardella, L., Aschiaffonati, L., Rappocciolo, E., Levi, S., Arosio, P., and Berenelli-Zazzera, A. (1986). Regulation of ferritin synthesis in malignant and nonmalignant lymphoid cells. *Biochem. Biophys. Res. Commun.*, 139:652.
- Campbell, C.H., Solgonick, R.M., Linder, M.C. (1989).
 Translational regulation of ferritin synthesis in rat spleen: Effects of iron and inflammation. *Biochem. Biophys. Res. Comm.*, 160:453.
- Carney, D.N., Gazdar, A.F., Bepler, G., Guccion, J.G., Marangos, P.J., Moody, T.W., Zweig, M.H., and Minna, J.D. (1985). Establishment and identification of small-cell lung cancer cell lines having classic and variant features. *Cancer Res.*, 45:2913.
- Casey, J.L., di Jeso, B., Rao, K., Klausner, R.D., and Harford, J.B. (1988). Two genetic loci participate in the regulation by iron of the gene for the human transferrin receptor. *Proc. Natal. Acad. Sci.* USA, 85:1787.
- Carvalho G.S. and de Sousa, M. (1988). Iron exerts a specific inhibitory effect on CD2 expression of human PBL. *Immunol. Lett.*, 19:163
- Chang, M.p., Makinodan, T., Peterson, W.J., and Strehler, B.I. (1982). Role of T cells and adherent cells in age related decline in murine interleukin 2 production. J. Immunol., 129:2426.

- Chasteen, N.D., (1983). The identification of the probable locus of iron and anion binding in the transferrins. *Trends. Biochem. Sci.*, 8:272.
- Cheung, W.Y.(1980). Calmodulin plays a pivotal role in cellular recognition. *Science*, 207:19.
- Chien, Y., Iwashima, M., Kaplan, K.B., Elliott, J.F., and Davis, M.M. (1987). A new T cell receptor gene located within the α locus and expressed early in T cell differentiation. *Nature*, 327:677.
- Costanzo, F., Santoro, Colantuoni, V., Bensi, G., Raugei, G., Romano, V. and Cortese, R. (1984). Cloning and sequencing of a full length cDNA coding for a human apoferritin H chain: evidence for a multigene family. *EMBO J.*, 3:23.
- Cox, T.M., Mazurier, J., Spik, G., Montreuil, J., and Peters, T.J. (1979). Iron binding proteins and influx of iron cross the duodenal brush border. Evidence for specific lactotransferrin receptors in the human intestine. *Biochim. Biophys. Acta*, 588:120.
- Cragg, S.J., Drysdale, J., and Worwood, M. (1985). Genes for the 'H' subunit of human ferritin are present on a number of human chromosomes. *Hum. Genet.*, **71**:108.
- Cragg, S.J., Wagstaff, M., and Worwood, M. (1981). Detection of glycosylated subunit in human serum ferritin. *Biochem. J.*, 199:565.
- Crago, S.S., Kulhavey, R., Prince, S.J., and Mestecky, J. (1981). Inhibition of the pokeweed mitogen-induced response of normal peripheral blood lymphocytes by humoral components of colostrum. *Clin. Exp. Immunol.*, 45:386.
- Cramer, E., Pryzwansky, K.B., Villeval, J-L., Testa, U., and Breton-Gorius, J. (1985) Ultrastructural localization of lactoferrin and myeloperoxidase in human neutrophils by immunogold. *Blood*, 65:423.
- Crane, F.L., Sun, I.L., Clark, M.G., Grebing, C., and Low, H. (1985). Transplasma-membrane redox systems in growth and development. *Biochim. Biophys. Acta*, 811:233.

- Crichton, R.R. (1985). Intracellular iron metabolism. In Proteins of Iron Storage and Transport (Eds. G. Spik, J. Montreuil, R.R. Crichton, and J. Mazurier), pp. 99-110. Elsevier, Amsterdam.
- Dallman, P.R. (1987). Iron deficiency and the immune response. Am. J. Clin. Nutrition, 46:179.
- Dautry-Varsat, A., Ciechanover, A., and Lodish, H.F. (1983). pH and the recycling of transferrin during receptor-mediated endocytosis. *Proc. Natl. Acad. Sci. USA*, 80:2258.
- Davis, R.J. and Meisner, H. (1987). Regulation of transferrin receptor recycling by protein kinase C is independent of receptor phosphorylation at serine 24 in swiss 3T3 fibroblasts. J. Biol. Chem., 262:10641.
- de Sousa, M. (1983). Blood transfusion and allograft survival. Ironrelated immunosuppression. *Lancet*, 2:681.
- de Sousa, M. (1989). The immunology of iron overload. In *Iron in Immunity, Cancer and Inflammation* (Eds M. de Sousa and J.H. Brock), pp. 247-258. John Wiley and Sons, Chichester.
- de Vos, R., Sciot, R., Van Eyken, P., and Desmet, V.J. (1988). Immunoelectron microscopic localization of hepatic transferrin receptors in human liver with and without iron overload. Virchow's Arch. (Cell. Pathol.), 55:11.
- Dembic, Z., Werner, H., Weiss, S., McCubey, J., Kiefer, H., von Boehmer, H., and Steinmetz, M. (1986). Transfer of specificity by murine α and β T cell receptor genes. *Nature*, **320**:232.
- Depper, J.M., Leonard, W.J., Drogula, C., Krönke, M., Waldmann, T.A., and Greene, W.C. (1985). Interleukin-2 (IL-2) augments transcription of IL-2 receptor gene. *Proc. Natl. Acad. Sci.* USA, 82:4230.
- Dialynas, D.P., Wilde, D.B., Marrack, P., Wall, K.A., Otten, G., Loken, R.L., Pierras, M., and Kappler, J. (1983).
 Characterization of the murine antigenic determinant, designated L3T4a, recognized by monoclonal antibody GK1.5: expression of

L3T4a by functional T cell clones appears to correlate primarily with class II MHC antigen-reactivity. *Immunol. Rev.*, **74**:29.

- Dickson, D.P.E., Reid, N.M.K., Mann, S., Wade, V.J., Ward, R.J., Peters, T.J. (1988). Mössbauer spectroscopy, electron microscopy and electron diffraction studies of the iron core in various human and animal haemosiderins. *Biochim. Biophys. Acta*, 957:81.
- Dickson, P.W., Aldred, A.R., Marley, P.D., Tu, G.F., Howlett, G.J., and Schreiber, G. (1985). High prealbumin and transferrin mRNA levels in the choroid plexus of rat brain. *Biochem. Biophys. Res. Commun.*, 127:890.
- Dillner-Centrelind, M.L., Hammerstrom, S., and Perlmann, P. (1979). Transferrin can replace serum for *in vitro* growth of mitogenstimulated T-lymphocytes. *Eur. J. Immunol.*, **9**:942.
- Dinarello, C.A. (1984). Interleukin 1. Rev. Infec. Dis., 6:51.
- Dinarello, C.A., Marnoy, S.O., and Rosenwasser, L.J. (1983). Role of arachidonate metabolism in the immunoregulatory function of human leukocytic pyrogen/lymphocyte-activating factor/interleukin 1. J. Immunol., 130:890..
- Dorland, L., Haverkamp, J., Vliegenthart, J.F.G., Spik, G., Fournet, B., and Montreuil, J. (1979). Investigation by 360 MHz ¹H-nuclear magnetic resonance spectroscopy and methylation analysis of the single glycan chain of chicken ovotransferrin. *Eur. J.Biochem.*, 100:569.
- Dörner, M.H., Broxmeyer, H.E., Silverstone, A., and Andreef, M. (1983a). Biosynthesis of ferritin subunits from different cell lines of H1-60 human promyelocytic leukemia cells and the release of acidic isoferritin-inhibitory activity against normal granulocyte-macrophage progenitor cells. Br. J. Haematol., 55:47.
- Dörner, M.H., Silverstone, A., Nishiya, K., de Sostoa, A., Munn, G., and de Sousa, M. (1980). Ferritin synthesis by human Tlymphocytes. Science, 209:1019.

- Dörner, M.H., Silverstone, A.E, Sostoa, A., Munn, G., and de Sousa M. (1983b). Relative subunit composition of the ferritin synthesized by selected human lymphomyeloid cell population. *Exp. Hematol.*, 11:866.
- Draser, B.S. and Irving, D. (1973). Environmental factors and cancer of the colon and breast. *Br. J. Cancer*, **27**:167.
- Drysdale, J.W. (1970). Microheterogeneity of ferritin molecules. Biochem. Biophys. Acta, 207:256.
- Drysdale, J.W. (1988). Human ferritin gene expression. *Prog. Nucleic* Acid Res., 35:127.
- Drysdale, J.W. and Munro, H.N. (1966). Regulation of synthesis and turnover of ferritin in rat liver. J. Biol. Chem., 241:3630.
- Duncan, R.L. and McArthur, W.P. (1981). Lactoferrin-mediated modulation of mononuclear cell activities. I. Suppression of the murine *in vitro* primary antibody response. *Cell. Immunol.*, 63:308.
- Dwyer, J., Wood, C., McNamara, J., Williams, A., Andiman, W., Rink, L., O'Connor, T., and Pearson, H. (1987). Abnormalities in the immune system of children with beta-thalassemia. *Clin.Exp. Immunol.*, 68:621.
- Ekblom, P., Thesleff, I., Saxen, L., Miettinen, A., and Timpl, R. (1983). Transferrin as a fetal growth factor: acquisition of responsiveness related to embryonic induction. *Proc. Natl. Acad. Sci. USA*, 80:2651.
- Enns, C.A. and Sussman, H.H. (1981a). Physical characterization of the transferrin receptor in human placentae. J. Biol. Chem., **256**:9820.
- Enns, C.A. and Sussman, H.H. (1981b). Similarities between the transferrin receptor proteins on human reticulocytes and human placentae. J. Biol. Chem., 256:12620.
- Esaguy, N., Aguas, A.P., and Silva, M. (1989). High-resolution localization of lactoferrin in human neutrophils: Labelling of secondary

granules and cell heterogeneity. J. Leukocyte Biol., 46:51.

- Esparza, I. and Brock, J.H. (1980). The effect of trypsin digestion on the structure and iron donating properties of transferrins form several species. *Biochim. Biophys. Acta*, **622**:297.
- Evans, R.W. and Williams, J. (1978). Studies of the binding of different iron donors to human serum transferrin and isolation of iron binding fragments from the N- and C-terminal regions of the protein. *Biochem. J.*, 173:543.
- Fan, P.T., Yu, D.T.W., Opelz, C.G., Goldberg, L., and Bluestone,
 R. (1977). Daily variations in circulating lymphocyte counts, T. B.
 proportions and responsiveness to PHA. *Life Sci.*, 21:793.
- Feeney, R.E. and Komatsu, S.K. (1966). The transferrins. *Structure* and Bonding, 1:149.
- Fernandez-Pol, J.A. (1978). Isolation and characterization of a siderophore-like growth factor from mutants of SV40-transformed cells adapted to picolinic acid. *Cell*, **14**:489.
- Fernandez-Pol, J.A. (1983). Siderophore-like growth factor synthesis by transformed cells. *Microbiology*, **313**:317.
- Fibach, E., Konijn, A.M., and Rachmilewitz, E.A. (1985). Changes in cellular ferritin content during myeloid differentiation of human leukemic cell lines. *Am. J. Hematol.*, **18**:143.
- Figarella, C. and Sarles, H. (1975). Lactoferrin, a protein of human pancreatic external secretion. *Scand. J. Gastroenterol.*, **10**:449.
- Finch, C.A. and Huebers, H. (1982). Perspectives in iron metabolism. N. Eng. J. Med., 306:1520.
- Finkelstein, P.A., Sciortino, C.V., and McIntoch, M.A. (1983). Role of iron in microbe-host interactions. Rev. Infect. Dis., 5(suppl. 4):759
- Fischer, E., Lenhard, V., Seittert, P., Kluge, A., and Johansen, R. (1980). Blood transfusion-induced suppression of cellular immunity in man. *Human Immunol.*, 3:189.

- Fletcher, J. (1989). Iron, the iron-binding proteins and bone marrow cell differentiation. In *Iron in Immunity, Cancer and Inflammation* (Eds. M. de Sousa and J.H. Brock), pp. 223-244. John Wiley and Sons, Chichester.
- Fodor, I. and Marx, J.J.M. (1988). Lipid peroxidation of rabbit small intestinal microvillus membrane vesicles by iron complex. *Biochem. Biophys. Acta*, 961:69.
- Foley, G.E., Lazarus, H., Farber, S., Uzman, B.G., Boone, B.A., and McCarthy, R.E. (1965). Continuous culture of human lymphoblasts from peripheral blood of a child with acute leukemia. *Cancer*, 18:522.
- Ford, G.C., Harrison, P.M., Rice, D.W., Smith, J.M.A., Treffry,
 A., White, J.C., and Yariv, J. (1984). Ferritin: design and formation of an iron-storage molecule. *Philos. Trans. R. Soc. Lond.*B. Biol. Sci., 304:551.
- Funk, F., Lenders, J.P., Crichton, R.R., and Scheider, W. (1985). Reductive mobilization of ferritin iron. *Eur. J. Biochem.*, **152**:167.
- Furmanski, P., Li, Z-P., Fortuna, M.B., Swamy, C.V.B., and Das, R. (1989). Multiple molecular forms of human lactoferrin. J. Exp. Med., 170:415.
- Galbraith, R.M., Werner, P., Arnaud, P., and Galbraith, G.M.P. (1980). Transferrin binding to peripheral blood lymphocytes activated by phytohemagglutinin involves a specific receptor: Ligand interaction. J. Clin. Invest., 66:1135.
- Gelfand, E.W., Cheung, R.K., Mills, G.B., and Grinstein, S. (1985). Mitogens trigger a calcium-independent signal for proliferation in phorbol ester-treated lymphocytes. *Nature*, 315:419.
- Gitlin, D., and Biasucci, A. (1969). Development of γ -G, γ -A, γ -M, β 1C/ β 1A, C'1 esterase inhibitor, ceruloplasmin, transferrin, hemopexin, haptoglobin, fibrinogen, plasminogen, α_1 -antitripsin, orosomucoid, β -lipoprotein, α_2 -macroglobulin, and prealbumin in

the human conceptus. J. Clin. Invest., 48:1433.

- Goavec, M., Mazurier, J., Montreuil, J., and Spik, G. (1985). Roles des glyncanes dans la fixation de la sérotransferrine humaines sur les macrophages alvéolaires humains. C. R. Acad. Sci. Paris, serie 3, 301:689.
- Good, M.F., Chapman, D.E., Powell, L.W., and Halliday, J.W. (1986). The effect of iron (Fe³⁺) on the cloning efficiency of human memory T4⁺ lymphocytes. *Clin. Exp. Immunol.*, 66:340.
- Good, M.F., Chapman, D.E., Powell, L.W., and Halliday, J.W. (1987). The effect of experimental iron overload on splenic T cell function: analysis using clonic techniques. *Clin. Exp. Immunol.*, 68:375..
- Good, M.F., Powell, L.W., and Halliday, J.W. (1988). Iron status and cellular immune competence. *Blood Rev.*, **2**:43.
- Goto, Y., Paterson, M., and Listowsky, I. (1983). Iron uptake and regulation of ferritin synthesis by hepatoma cells in hormone-supplemented serum-free media. J. Biol. Chem., 258:5248.
- Grady, R.W., Akbar, A., Giardina, P.J., Hilgartner, M.W., and de Sousa, M. (1985). Disproportionate lymphoid cell subsets in thalassemia major: the relative contributions of transfusion and splenectomy. Br. J. Haematol., 72:361.
- Griffiths, E. (1987). Iron in biological systems. In *Iron and Infection* (Eds. J.J. Bullen and E. Griffiths), pp. 1-19. John Wiley and Sons, Chichester.
- Guglielmo, P., Cunsolo, F., Lombrado, T., Sortino, G., Giustolisi, R., Cacciola, E., and Cacciola, E. (1984). T subset abnormalities in thalassemia intermedia: possible evidence of a thymus functional deficiency. Acta Haematol., 72:361.
- Gutteridge, J.M.C., Paterson, S.K., Segal, A.W., and Halliwell, B. (1981). Inhibition of lipid peroxidation by the iron-binding protein lactoferrin. *Biochem. J.*, **199**:259.
- Gutteridge, J.M.C., Richmond, R., and Halliwell, B. (1979).

Inhibition of the iron-catalysed formation of hydroxyl radicals from superoxide and of lipid peroxidation by desferrioxamine. *Biochem. J.*, 184:469.

- Hahn, D., Baviera, B., and Ganzoni, A.M. (1975). Functional heterogeneity of transport iron compartment. I. In vivo radioiron clearance from high and low saturated transferrin. Acta Haematol., 53:285.
- Halliwell, B. (1978). Superoxide-dependent formation of hydroxyl radicals in the presence of iron chelates. Is it a mechanism for hydroxyl radical production in biological systems. FEBS Lett., 92:321.
- Halliwell, B. (1985). Use of desferrioxamine as a probe for irondependent formation of hydroxyl radicals. *Biochem. Pharmacol.*, 34:229.
- Halliwell, B. (1989). Current status review: free radicals, reactive oxygen species and human disease: a critical evaluation with special reference to atherosclerosis. *Br. J. Exp. Pathol.*, **70**:737.
- Hamilton, T.A., Wada, H.G., and Sussman, H.H. (1979).
 Identification of transferrin receptors on the surface of human cultured cells. *Proc. Natl. Acad. Sci. USA*, **76**:6406.
- Harrison, P.M., Fischbach, F.A., Hoy, T.G., and Haggis, G.H. (1967). Ferric oxyhydroxide core of ferritin. *Nature*, **216**:1188.
- Harrison, P.M., Clegg, G.A., and May, A. (1980). Ferritin structure and function. In *Iron in Biochemistry and Medicine* (Eds A. Jacobs and M. Worwood), Vol. II. pp. 131-166. Academic Press, London.
- Hashizume, S., Kuroda, K., and Murakami, H. (1983). Identification of lactoferrin as an essential growth factor for human lymphocytic cell lines in serum-free medium. *Biochim. Biophys. Acta*, **763**:377.
- Haurani, F.I. and Ballas, S.K. (1984). Iron metabolism. In *The* reticuloendothelial System (Eds. S.M. Reicherd and J.P. Filkins), Vol. 7A pp. 353-377. Plenum, New York.

¥

- Haurani, F.I., Meyer, A., O'Brien, R. (1973). Production of transferrin by the macrophage. J. Reticuloendothelial Soc., 14:309.
- Heaphy, S. and Williams, J. (1982). Removal of iron from diferric rabbit serum transferrin by rabbit reticulocytes. *Biochem.J.*, 205:619.
- Hekman, A. (1971). Association of lactoferrin with other proteins, as demonstrated by changes in electrophoretic mobility. *Biochim. Biophys. Acta*, 251:380.
- Helmer, M.E., Cook, H.W., and Lands, W.E.M. (1979).Prostaglandin biosynthesis can be triggered by lipid peroxidates.Arch. Biochem. Biophys., 193:340.
- Hemmapardh, D. and Morgen, E.H. (1974). The mechanism of iron exchange between synthetic iron chelators and rabbit reticulocytes. *Biochim. Biophys. Acta*, 373:84.
- Hentze, M.W., Caughman, S.W., Rouault, T.A., Barriocanal, J.G., Dancis, A., Harford, J.B., and Klausner, R.D. (1987).
 Identification of the iron-responsive element for the translational regulation of human ferritin mRNA. *Science*, 238:1570.
- Hershko, C. and Peto, T.E.A. (1987). Non-transferrin plasma iron. Br. J. Haematol., 66:149.
- Hershko, C. (1989). Mechanism of iron toxicity and its possible role in red cell membrane damage. *Sem. Hematol.*, **26**:277.
- Hershko, C., Graham, G., Bates, G.W., and Rachmilewitz, E.A. (1978). Non-specific serum iron in thalassemia: an abnormal serum iron fraction of potential toxicity. *Br. J. Haematol.*, **40**:255.
- Hershko, C. and Weatherall, D. (1988). Iron-chelating therapy. Crit. Rev. Clin. Lab. Sci., 26:303.
- Hider, R.C. (1984). Siderophore mediated absorption of iron. *Structure and Bonding*, 38:25.
- Hoffbrand, A.V., Ganeshaguru, K., Hooton, J.W.L., and Tattersall,

M.H.N. (1976). Effect of iron deficiency and desferrioxamine on DNA synthesis in human cells. *Br. J. Haematol.*, **33**:517.

- Hoffenberg, R., Gordon, A.H., Black, E.G., and Louis, L.N. (1970). Plasma protein catabolism by the perfused rat liver. The effect of alteration of albumin concentration and dietary protein depletion. *Biochem. J.*, **118**:401.
- Hopkins, C.R. (1985). The appearance and internalization of transferrin receptors at the margins of spreading human tumor cells. *Cell*, **40**:199.
- Hovanessian, A.G. and Awdeh, Z.L. (1976). Gel isoelectric focusing of human serum transferrin. *Eur. J. Biochem.*, **68**:333.
- Hoy, T.G. and Jacobs, A. (1981). Ferritin polymers and the formation of haemosiderin. In *The Biochemistry and Physiology of Iron* (Eds. P. Saltman and J.Hegenauer), pp. 211-214. Elsevier North Holland, New York.
- Hu, W-L., Mazurier, J., Sawatzki, G., Montreuil, J., and Spik, G. (1988). Binding characteristics of membrane-bound and Triton X-100-solubilised forms. *Biochem. J.*, 249:435.
- Huebers, H., Josephson, B., Huebers, E., Csiba, E., and Finch, C.A. (1984). The occupancy of the binding sites of human transferrin. *Proc. Acad. Sci.*, 81:4326.
- Huebers, H.A. and Finch, C.A. (1987). The physiology of transferrin and transferrin receptors. *Physiol. Rev.*, 67:520.
- Huggenvik, J.I., Idzerda, R.L., Haywood, L., Lee, D.C., McKnight, G.C., and Griswold, M.D. (1987). Transferrin messenger ribonucleic acid: Molecular cloning and hormonal regulation in rat Sertoli cells. *Endocrinology*, **120**:332.
- Hunter, R.L., Bennett, B., Towns, M., and Vogler, W.R. (1984).
 Transferrin in disease. II: Defects in regulation of transferrin saturation with iron contributes to susceptibility to infection. Am. J. Clin. Pathol., 81:748.

- Iacopetta, B.J., Morgan, E.H., and Yeoh, G.C.T. (1982). Transferrin receptor and iron uptake during erythroid cell development. *Biochim. Biophys. Acta*, 687:204.
- Iancu, T.C., Landing, B.H., and Neusrein, H.B. (1977). Pathogenetic mechanism in hepatic cirrhosis of thalassemia major: light and electron microscopic studies. *Path. Annal.*, 12:171.
- Iancu, T.C., Shiloh, H., Link, G., Bauminger, E.R., Pinson, A., and Hershko, C. (1987). Ultrastructural pathology of iron-loaded rat myocardial cells in culture. *Br. J. Exp. Pathol.*, 68:53.
- Idzerda, R.L., Huebers, H., Finch, C.A., and McKnight, G.S. (1986). Rat transferrin gene expression: tissue-specific regulation by iron deficiency. *Proc. Natl. Acad. Sci.USA*, 83:3723.
- Imboden, J. B., Weiss, A., and Stobo, J.D. (1985). The antigen receptor on a human T cell line initiates activation by increasing cytoplasmic free calcium. *J. Immunol.*, **134**:663.
- Imboden, J.B. and Stobo, J.D. (1985). Transmembrane signaling by the T cell antigen receptor. J. Exp. Med., 161:446.
- Isakov, N., Scholz, W., and Altman, A. (1986). Signal transduction and intracellular events in T lymphocyte activation. *Immunol. Today*, 7:271.
- Jacobs, A. (1977). Low molecular weight intracellular iron transport compounds. *Blood*, **50**:433.
- Jacobs, A., Hoy, T., Humphreys, J., and Perera, P. (1978). Iron overload in Chang cell cultures: biochemical and morphological studies. *Br. J. Exp. Pathol.*, **59**:489.
- Jain, S.K., Barrett, K.J., Boyd, D., Faureau, M.F., Crampton, J., and Drysdale, J.W. (1985). Ferritin H and L chains are derived from different multigene families. J. Biol. Chem., 260:11762.
- Jandl, J.H., Inman, J.K., Simmons, R.L., and Allen, D.W. (1959). Transfer of iron from serum iron-binding protein to reticulocytes. J. Clin. Invest., 38:161.

- Jeltsch, J.M. and Chambon, P. (1982). The complete nucleotide sequence of the chicken ovotransferrin mRNA. *Eur. J. Biochem.*, **122**:291.
- Johansson, B.G. (1969). Isolation of crystalline lactoferrin from human milk. Acta Chem. Scand., 23:683.
- Jones, B.M., Worwood, M., and Jacobs, A. (1983). Isoferritin in normal leucocytes. Br. J. Haematol., 55:73.
- June, C.H., Ledbetter, J.A., Rabinovitch, P.S., Martin, P.J., Beatty, P.G., and Hansen, J.A. (1986). Distinct patters of transmembrane calcium flux and intracellular calcium mobilization after differentiation antigen cluster 2 (E rosette receptor) or 3 (T3) stimulation of human lymphocytes. J. Clin. Invest., 77:1224.
- Kanellopoulos, J.M., de Petris, S., Leca, G., and Crumpton, M.J. (1985). The antigenic lectin from *Phaseolus vulgaris* does not recognize the T3 antigen of human T lymphocytes. *Eur. J. Immunol.*, 15:479.
- Kapadia, A., de Sousa, M., Markenson, A., Miller, D., Good, R.A., and Gupta, S. (1980). Lymphoid cell sets and serum immunoglobulins in patients with thalassemia intermedia: relationship to serum iron and splenectomy. *Br. J. Haematol.*, 45:405.
- Karin, M. and Mintz, B. (1981). Receptor-mediated endocytosis of transferrin in developmentally totipotent mouse tetracarcinoma stem cells. J. Biol. Chem., 256:3245.
- Katz, J.H. (1961). Iron and protein kinetics studies by means of doubly labelled human crystalline transferrin. J. Clin. Invest., 40:2143.
- Kay, J.E. and Benzie, C.R. (1986). The role of the transferrin receptor in lymphocyte activation. *Immunol. Lett.*, **12**:55.
- Kelly, K., Cochran, B.H., Stiles, C.D., and Leder, P. (1983). Cell-specific regulation of the *c-myc* gene by lymphocyte mitogens and platelet-derived growth factor. *Cell*, 35:603.

- Kemp, J.D., Thorson, J.A., McAlmont, T.H., Horowitz, M., Cowdery, J.S., and Ballas, Z.K. (1987). Role of the transferrin receptor in lymphocyte growth: a rat IgG monoclonal antibody against the murine transferrin receptor produces highly selective inhibition of T and B cell activation protocols. J. Immunol., 138:2422.
- Keown, P. and Descamps-Latscha, B. (1983). In vitro suppression of cell-mediated immunity by ferroproteins and ferric salts. Cell. Immunol., 80:257.
- Kernof, L.M. and Baker, G. (1980). Direct measurement of transferrin synthesis rates in man using the ¹⁴C carbonate method. *Anal. Biochem.*, 106:529.
- Kessler, C.M., Schulof, R.S., Goldstein, A.L., Naylor, P.H., Luban, N.L.C., Kelleher, J.F., and Reaman, G.H. (1983).
 Abnormal T-lymphocyte sub-population associated with transfusions of blood-derived products. *Lancet*, 1:991.
- Khalfoun, R., Degenne, D., Crouzat-Reynes, G., and Bardos, P. (1986). Effect of human syncytiotrophoblast plasma membrane-soluble extracts on *in vitro* mitogen-induced lymphocyte proliferation. A possible inhibition mechanism involving the transferrin receptor. J. Immunol., 137:1187.
- Kinkade, J.M., Miller, W.W.K., and Segars, F.M. (1976). Isolation and characterization of murine lactoferrin. *Biochim. Biophys. Acta*, 446:407.
- Kitada, S. and Hays E. (1985). Transferrin-like activity produced by murine malignant T-lymphoma cell lines. *Cancer Res.*, **45**:3540.
- Klausner, R. D., Ashwell, G, Van Renswoude, J, Harford, B.G., and Bridges, K.R. (1983). Binding of apotransferrin to K562 cells : explanation of the transferrin cycle. *Proc. Natl. Acad. Sci. USA*, 80:2263.
- Klausner, R. D., Harford, J.B., and Van Renswoude, J. (1984a). Rapid internalization of the transferrin receptor in K562 cells is triggered by ligand binding or treatment with a phorbol ester. *Proc. Natl. Acad. Sci. USA*, 81:3005.

- Klausner, R. D., Van Renswoude, J, Kempf, C., Rao, K., Bateman, J.L., and Robbins, A.R. (1984b). Failure to release iron from transferrin in a chinese hamster ovary cell mutant pleiotropically defective in endocytosis. J. Cell Biol., 98:1098.
- Kohgo, Y., Yacota, M., and Drysdale, J.W. (1980). Differential turnover of rat liver isoferritins. J. Biol. Chem., 255:5195.
- Kohno, H., Tahetani, S., and Takunoga, R. (1985). Tumor promoting, phorbol ester-induced phosphorylation of cell surface transferrin receptors in human erythroleukemic cells. *Cell Structure Function*, 10:95.
- Konijn, A.M., Carmel, N., Levy, R., and Hershko, C. (1981). Ferritin synthesis in inflammation. II. Mechanism of increased ferritin synthesis. Br. J. Haematol., 49:361.
- Kraut, E.H. and Sagone, A.L. (1981). The effect of oxidant on lymphocyte membrane and functions. J. Lab. Clin. Med., 98:697.
- Krönke, M., Leonard, W.J., Depper, J.M., and Green, W.C (1985). Sequential expression of genes involved in human T-lymphocyte growth and differentiation. J. Exp. Med., 161:1593.
- Krysteva, M.A., Mazurier, J., Spik, G., and Montreuil, J. (1975). Comparative study on histidine modification by diethylpyrocarbonate in human serotransferrin and lactotransferrin. *FEBS Lett.*, **56**:337.
- Kühn, L.C., Müllner, E.W., Teixeira, S., and Neupert, B. (1989).
 Regulation of transferrin receptor expression. *IXth International Conference on Proteins of Iron Transport and Storage*, Brisbane.
 Abstract p 24.
- Kumagai, N., Benedict, S.H., Mills, G.B., and Gelfand, E.W. (1987). Requirements for the simultaneous presence of phorbol esters and calcium ionophores in the expression of human T lymphocyte proliferation-related genes. J. Immunol., 139:1393.
- Kumagai, N., Benedict, S.H., Mills, G.B., and Gelfand, E.W. (1988). Comparison of phorbol ester/calcium ionophore and

phytohemagglutinin-induced signaling in human T lymphocytes. Demonstration of Interleukin 2-independent transferrin receptor gene expression. J. Immunol., 140:37.

- Kumar, R., Singh, U, and Mehrotra, G.C. (1978). Mechanisms of hypoferraemia in acute and chronic infection. *Indian J. Med. Res.*, 68:508.
- Kuvibidila, S., Nauss, K.M., Baliga, B.S., and Suskind, R.M. (1983). Impairment of blastogenesis response of splenic lymphocytes from iron-deficient mice: *in vivo* repletion. Am. J. Clin. Nutr., 37:15.
- Landschulz, W. and Ekblom, P. (1985). Iron delivery during proliferation and differentiation of kidney tubules. J. Biol. Chem., **98**:596.
- Landschulz, W., Thesleff, I., and Ekblom, P. (1984). A lipophilic iron chelator can replace transferrin as a stimulator of cell proliferation and differentiation. *J. Cell. Biol.*, **98**:596.
- Lane, R.S. (1973). Iron uptake by rabbit reticulocytes. Br. J. Haematol., 24:343.
- Larrick, J.W. and Cresswell, P. (1979). Modulation of cell surface iron transferrin receptors by cellular density and state of activation. J. Supramol. Struct., 11:579.
- Laurell, C. and Ingelman, B. (1947). The iron binding protein of swine serum. Acta Chem. Scand., 1:770.
- Le Magueresse, B., Pineau, C., Guillo, F., and Jegou, B. (1988). Influence of germ cells upon transferrin secretion by rat Sertoli cells *in vitro*. J. Endocrinol., **118**:R13.
- Leberman, H.M., Cohen, A., Lee, J.W.W., Freedman, M.H., and Gelfand, E.W. (1984). Desferrioxamine: a reversible S-phase inhibitor of human lymphocyte proliferation. *Blood*, **64**:748.
- Ledbetter, J.A., June, C.H., Martin, P.J., Spooner, C.E., Hansen, J.A., and Meier, K.E.(1986). Valency of CD3 binding and

internalization of the CD3 cell-surface complex control T cell responses to second signals: distinction between effects on protein kinase C, cytoplasmic free calcium, and proliferation. J. Immunol., **136**:3945.

- Lee, D.C., McKnight, G.S., and Palmiter, R.D. (1978). The action of estrogen and progesterone on the expression of the transferrin gene. A comparison of the response in chick liver and oviduct. J. Biol. Chem., 253:3494.
- Lee, E.Y-H., Barcellos-Hoff, M.H. Chen, L-H., Parry, G., and Bissell, M. J. (1987). Transferrin is a major mouse milk protein and is synthesised by mammary epithelial cells. *In Vitro*, **23**:221.
- Lee, N.T., Chae, C-B., and Kierszenbaum, A.L. (1986). Contrasting levels of transferrin gene activity in cultures rat Sertoli cells and indirect seminiferous tubules. *Proc. Natl. Acad. Sci. USA*, 83:8177.
- Leibman, A. and Aisen, P. (1979). Distribution of iron between the binding sites of transferrin in serum: methods and results in normal human subjects. *Blood*, **53**:1058.
- Leibold, E.A. and Munro, H.N. (1988). Cytoplasmic protein binds in vitro to a highly conserved sequence in the 5' untranslated region of ferritin heavy- and light-subunit mRNA. Proc. Natl. Acad. Sci. USA, 85:2171.
- Leibold, E.A., Aziz, N.N., Brown, A.J., and Munro, H.N. (1984). Conservation in rat liver light and heavy subunit sequences of mammalian ferritin. Presence of unique octapeptide. J. Biol. Chem., 259:4327.
- Lema, M.J. and Sarcione, E.J. (1981). A comparison of iron induction of rat lymphocyte ferritin synthesis *in vivo* and *in vitro*. Comp. Biochem. Physiol., 69:287.
- Lescoat, G., Jouanolle, H., Desvergne, B., Pasdeloup, Kneip, B., Deugnier, Y., Guillouzo, A., and Brissot, P. (1989). Effect of iron overload on transferrin secretion by cultured fetal rat hepatocytes. *Biology of the Cell*, 65:221.

Lestas, A.N. (1976). The effect of pH upon human transferrin: Selective

labelling of two iron-binding sites. Br. J. Haematol., 32:341.

- Letendre, E.D. (1985). The importance of iron in the pathogenesis of infection and neoplasia. *Trends Biochem. Sci.*, **9**:166.
- Levi, S., Salfeld, J., Franceschinelli, F., Cozzi, A., Dörner, M.H., and Arosio, P. (1989). Expression and structural and functional properties of human ferritin L-chain from *Escherichia coli*. *Biochemistry*, 28:5179.
- Levin, M.J., Tuil, D., Uzan, G., Dreyfus, J.C., and Kahn, A. (1984). Expression of the transferrin gene during development of nonhepatic tissues: high levels of transferrin mRNA in fetal muscle and adult brain. *Biochem. Biophys. Res. Commun.*, 122:212.
- Linder, M. and Munro, H.N. (1972). Assay of tissue ferritin. Anal. Biochem., 48:266.
- Linder, M.C, Mulligan, M, and Henley, D. (1983). Low molecular weight iron pools in tissue of the rat and pig. In Structure and Function of Iron Storage and Transport Proteins (Eds I. Urashizaki, P. Aisen, I. Listowsky, and J.W. Drysdale), pp. 465-468. Elsevier, Amsterdam.
- Lineback-Zins, J. and Brew, K. (1980). Preparation and characterization of an NH₂-terminal fragment of human serum transferrin containing a single iron-binding site. J. Biol. Chem., 255:708.
- Link, G., Pinson, A., and Hershko, C. (1985). Heart cells in culture: a model of myocardial iron overload and chelation. J. Lab. Clin. Med., 106:147.
- Lombardi, A., Tramonto, D., Braverman, L.E., and Ingbar, S.H. (1989). Transferrin in FTRL5 cells: regulation of its receptor by mitogenic agents and its role in cell growth. *Endocrinology*, 125:652.
- Lonnerdal, B. (1985). Biochemistry and physiological function of human milk proteins. Am. J. Clin. Nutr., 42:1299.

- Louache, F., Pelosi, E., Titeux, M., Peschle, C., and Testa, U. (1985). Molecular mechanisms regulating the synthesis of transferrin receptors and ferritin in human erythroleukemic cell lines. *FEBS Lett.*, 183:223.
- Lowry, O., Rosebrough, N., Farr, A., and Randall, R. (1951). Protein measurement with the folin phenol reagent. J. Biol. Chem., 193:265.
- Lum, J.B., Infante, A.J., Makker, D.M., Yang, F., and Bowman,
 B.H., (1986). Transferrin synthesis by inducer T lymphocytes. J. Clin. Invest., 77:841.
- Lum, J.B., Yang, F., Bowman, B.H. (1985). Molecular biology of the human transferrin gene. In *Proteins of Iron Storage and Transport* (Eds. G. Spik, J. Montreuil, R.R. Crichton, and J. Mazurier), pp. 357-360. Elsevier, Amsterdam.
- Lynch, S.R., Lipshchitz, D.A., Bothwell, T.H., and Charlton, R.W. (1974). Iron in the reticuloendothelial system. In *Iron in Biochemistry and Medicine* (Eds A. Jacobs and M. worwood), pp. 564-599. Academic Press, London.
- MacGillivray, R.T.A., Mendez, E., Sinha, S.K., Sutton, M.R., Lineback-Zins, J., and Brew, K. (1983). The primary structure of human serum transferrin. *J.Biol. Chem.*, **258**:3543.
- Mainou-Fowler, T. and Brock, J.H. (1985). Effect of iron deficiency on the response of mouse lymphocytes to concanavalin A: the importance of transferrin-bound iron. *Immunology*, 54:325.
- Makey, D.G., and Seal, U.S. (1976). The detection of four molecular forms of transferrin during the iron binding process. *Biochem. Biophys. Acta*, **453**:703.
- Manger B., A. Weiss, K. J. Hardy, and J. D. Stobo (1986). A transferrin receptor antibody represents one signal for the induction of IL-2 production by a human T-cell line. *J. Immunol.*, **136**:532-538.
- Mann, S., Bannister, J.V., Williams, R.J.P. (1986). Structure and composition of ferritin cores isolated from human spleen, limpet

(*Patella valgata*) hemolymph and bacterial (*Pseudomonas aeruginosa*) cells. J. Mol Biol., **188**:225.

- Martin, P.J., Longton, J.G., Ledbetter, J.A., Newman, W., Braun, M.P., Beatty, P.A., and Hansen, J.A. (1983). Identification and functional characterisation of two distinct epitopes on the human T cell surface protein Tp50. J. Immunol., 131:180.
- Marx, J.J.M. (1989). Behavior of iron (III) in aqueous solution: computed polynuclear complex formation compared with that during interaction with biological membrane. *IXth International Conference* on Proteins of Iron Transport and Storage. Brisbane. Abstract p. 36.
- Masson, P.L. and Heremans, J.F. (1968). Metal-combining properties of human lactoferrin (red milk protein). 1. The involvement of bicarbonate in the reaction. *Eur. J. Biochem.*, **6**:579.
- Mattia, E., Josic, D., Ashwell, G., Klausner, R., and van Renswoude, J. (1986). Regulation of intracellular iron distribution in K562 human erythroleukemia cells. J. Biol. Chem., 261:4587.
- May, P.M. and Williams, D.R. (1980). The inorganic chemistry of iron metabolism. In *Iron in Biochemistry and Medicine* (Eds A. Jacobs and M. worwood), pp. 1-26. Academic Press, London.
- May, W.S. and Tyler, G. (1987). Phosphorylation of the surface transferrin receptor stimulates receptor internalization in HL60 leukemia cells. J. Biol. Chem., 262:16710.
- May, W.S., Lapetina, E.G., and Cuatrecasas, P. (1986). Intracellular activation of protein kinase C and regulation of the surface transferrin receptor by diacyglycerol is a spontaneously reversible process that is associated with rapid formation of phosphatidic acid. *Proc. Natl. Acad. Sci. USA*, 83:1281.
- Mazurier, J., Legrand, D., Hu, W.L., Montreuil, J., Spik, G. (1989). Expression of human lactotransferrin receptors in phytohemagglutinin-stimulated human blood lymphocytes. Isolation of the receptors by antiligand-affinity chromatography. *Eur. J.*

Biochem., 179:481.

- Mazurier, J., Montreuil, J., and Spik, G. (1985). Visualization of lactotransferrin brush-border receptors by ligand-blotting. *Biochim. Biophys. Acta*, 821:453.
- McClelland, A. Kühn, L.C., and Ruddle, F.H. (1984). The human transferrin receptor gene: Genomic organisation, and the complete primary structure of the receptor deduced from a cDNA sequence. *Cell*, **39**:267.
- McCord, J.M. and Day, E.D. (1978). Superoxide-dependent production of hydroxyl radical catalysed by iron-EDTA complex. *FEBS Lett.*, 86:139.
- McGill, J.R., Naylor, S.L., Sakaguchi, A.Y., Moore, C.M., Boyd,
 D., Barrett, K.j., Shows, T.B., and Drysdale, J.W. (1987).
 Human ferritin H and L sequences lie on ten different chromosomes.
 Hum. Genet., 76:66.
- McGraw, T.E., Greenfield, L., and Maxfield, F.R. (1987). Functional Expression of the human transferrin receptor cDNA in chinese hamster ovary cells deficient in endogenous transferrin receptor. J. Cell. Biol., 105:207.
- McKnight, G.C., Lee, D.C., and Palmiter, R.D. (1980). Transferrin gene expression. Regulation of mRNA transcription in chick liver by steroid hormones and iron deficiency. *J. Biol. Chem.*, **255**:148.
- Means, A.R. and Dedman, J.R. (1980). Calmodulin-an intracellular calcium receptor. *Nature*, **285**:73.
- Mello Filho, A.C., Hoffmann, M.E., and Meneghini, R. (1984). Cell killing and DNA damage by hydrogen peroxide are mediated by intracellular iron. *Biochem. J.*, **218**:273.
- Mendelsohn, J., Trowbridge, I., and Castagnola, J. (1983). Inhibition of human lymphocyte proliferation by monoclonal antibody to transferrin receptor. *Blood*, **62**:821.

Metz-Boutigue, M-H., Jolles, J., Schoentgen, F., Legrand, D.,
Spik, G., Montreuil, J., and Jolles, P. (1984). Human lactotransferrin: amino acid sequence and structural comparisons with other transferrins. *Eur. J. Biochem.*, 145:659.

- Meuer, S.C., Hussey, R.E., Cantrell, D.A., Hodgdon, J.C., Schlossman, S.F., Smith, K.A., and Reinherz, E.L.(1984a).
 Triggering of the T3-Ti antigen-receptor complex results in clonal T cell proliferation through an interleukin-2-dependent autocrine pathway. *Proc. Natl. Acad. Sci. USA*, 81:1509.
- Meuer, S.C., Hussey, R.E., Fabbi, M., Fox, D., Acuto, O., Fitzgerald, K.A., Hodgdon, J.C., Protentis, J.P., Schlossman, S.F., and Reinherz, E.L.(1984b). An alternative pathway of T cell activation: a functional role for the 50 Kd T11 sheep erythrocyte receptor protein. *Cell*, 36:897.
- Moguilevsky, N., Courtoy, P.J., Masson, P.L. (1985). Study of lactoferrin-binding sites at the surface of blood monocytes. In *Proteins of Storage and Transport* (Eds G. Spik, J. Montreuil, R.R. Crichton, and J. Mazurier) pp. 199-202. Elsevier, Amsterdam.
- Montreuil, J., Mazurier, J., Legrand, D., and Spik, G. (1985). Human lactotransferrin: structure and function. In *Proteins of Iron Storage and Transport* (Eds. G. Spik, J. Montreuil, R.R. Crichton, and J. Mazurier), pp. 25-38. Elsevier, Amsterdam.
- Montreuil, J., Tonnelat, J., and Mullet, S. (1960). Préparation de la lactosidérophiline (lactotransferrine) du lait de femme. *Biochim. Biophys. Acta*, **45**:413.
- Morgan, E.H. (1964). Interaction between rabbit, human and rat transferrin and reticulocytes. Br. J. Haematol., 10:442.
- Morgan, E.H. (1980). The role of plasma transferrin in iron absorption in the rat. *Quart. J. Exp. Physiol.*, **65**:239.
- Morgan, E.H. (1981). Transferrin: biochemistry, physiology, and clinical significance. *Mol. Aspects Med.*, **4**:1.
- Morgan, E.H. (1983). Synthesis and secretion of transferrin. In *Plasma Protein Secretion by the Liver* (Eds. H. Glaumann, T. Peters,

Jr., and C. Redman), Vol. 4, pp. 96-101. Academic Press, New York.

- Morgan, E.H. and Peters, T. (1971). The biosynthesis of rat serum albumin V. Effect of protein deletion refeeding on albumin and transferrin synthesis. J. Biol. Chem., 146:3500.
- Moroz, C., Giler, S., Kupfer, B., and Urca, I. (1977). Ferritinbearing lymphocytes and T-cell levels in peripheral blood of patients with breast cancer. *Cancer Immunol. Immunother.*, **3**:101.
- Moroz, C., Kan, M., Chaimof, C., Marcus, H., Kupfer, B., and Cuckle, H.S. (1984). Ferritin-bearing lymphocytes in the diagnosis of breast cancer. *Cancer*, 54:84.
- Morrone, G., Corbo, L., Turco, M.C., Pizzano, R., de Felice, M., Bridges, S., and Venuta, S. (1988). Transferrin-like autocrine growth factor, derived from T-lymphoma cells, that inhibits normal T-cell proliferation. *Cancer Res.*, 48:3425.
- Morton, A.G. and Tavill, A.S. (1977). The role of iron in the regulation of hepatic transferrin synthesis. Br. J. Haematol., 36:383.
- Müllner, E.W. and Kühn, L.C. (1988). A stem-loop in the 3' untranslated region mediates iron-dependent regulation of transferrin receptor mRNA stability in the cytoplasm. *Cell*, **53**:815.
- Müllner, E.W., Neupert, B., and Kühn, L.C. (1989). A specific mRNA binding factor regulates the iron-dependent stability of cytoplasmic transferrin receptor mRNA. *Cell*, **58**:373.
- Mukhopadhyay, N., Fernbach, D.J., Mumford, D.M., and South, M.A. (1978). T and B cells and immune competence in human fetal liver cells. *Clin. Immunol. Immunopathol.*, **10**:59.
- Munn, C.G., Markeson, A.L., Kapadia, A., and de Sousa, M. (1981). Impaired T-cell mitogen responses in some patients with thalassemia major. *Thymus*, **3**:119.
- Møllgård, K., Stagaard, and Saunders, N.R. (1987). Cellular

distribution of transferrin immunoreactivity in the developing rat brain. *Neurosci. Lett.*, **78**:35.

- Nakamoto, S., Yamanoi, Y., Kawataba, T., Sahira, Y., Mori, M., and Awai, M. (1986). Lipid peroxidation and cytotoxicity of Ehrlich ascites tumor cells by ferric nitrilotriacetate. *Biochim. Biophys. Acta*, 889:15.
- Navas, P., Sun, I.L., Morre, D.J., and Crane, F.L. (1986). Decrease of NADH in HeLa cells in the presence of transferrin or ferricyanide. *Biochem. Biophys. Res. Commun.*, 135:110.
- Neckers L. M. (1984). Transferrin receptors regulate proliferation of normal and malignant B cells. Curr. Topics Microbiol. Immunol., 62:68.
- Neckers L. M., Vidal, C., McGlennan, R., and Colamonici, O. (1986). Phorbol-ester-induced surface transferrin receptor modulation. *Exp. cell. Res.*, **166**:151.
- Neckers, L.M. and Cossman, J. (1983). Transferrin receptor induction in mitogen-stimulated human lymphocytes is required for DNA synthesis and cell division and is regulated by interleukin-2. *Proc. Natl. Acad. Sci., USA,* 80:3494.
- Neilands, J.B. (1981). Microbial iron compounds. Ann. Rev. Biochem., 50:715.
- Neri, A., Brugiatelli, M., Iacopino, P., and Callea, F. (1984). Natural killer cell activity and T cell subpopulations in thalassemia major. *Acta Haematol.*, **71**:263.
- Neupert, B., Thompson, N.A., Meyer, C., and Kühn, L. (1990). A high yield affinity purification method for specific RNA-binding protein: isolation of the iron regulatory factor from human placenta. *Nucleic Acids Res.*, **18**:51.
- Nichols, B.L., McKee, K.S., Henry, J.F., and Putnam, M. (1987). Human lactoferrin stimulates thymidine incorporation into DNA of rat crypt cells. *Pediat. Res.*, **21**:563.

- Niemelä, A., Kulomaa, M., Vilja, P., Tuohimaa, P., and Saarikoski, S. (1989). Lactoferrin in amniotic fluid. *Human Reproduction*, 4:99.
- Nishiya, K., and Horwitz, D.A. (1982). Contrasting effect of lactoferrin on human lymphocyte and monocyte natural killer cell activity and antibody-dependent cell-mediated cytotoxicity. J. *Immunol.*, **129**:2519.
- Nishiya, K., de Sousa, M., Tsoi, E., Bognacki, J.J., and de Harven, E. (1980). Regulation of expression of human lymphoid cell surface marker by iron. *Cell. Immunol.*, 53:71.
- Niitsu, Y. and Listowsky, I. (1973). Mechanisms for the formation of ferritin oligomers. *Biochemistry*, **12**:4690.
- Nualart, P., Estevez, M.E., Ballart, I.J., de Miani, S.A., Penalver, J., and Sen, L. (1987). Effect of α interferon on the altered B-T cell mediated immunity in patients with thalassemia major. Am. J. Haematol., 24:151.
- O'Connell, M., Halliwell, B., Moorehouse, C.P., Aruoma, O.I., Baum, H., and Peters, T.J. (1986a). Formation of hydroxyl radicals in the presence of ferritin and haemosiderin. J. Biochem., 234:727.
- O'Connell, M.J., Ward, R.J., Baum, H., and Peters, T.J. (1985). The role of iron in ferritin- and haemosiderin-mediated lipid peroxidation in liposomes. *Biochem. J.*, 229:135.
- O'Connell, M.J., Ward, R.J., Baum, H., and Peters, T.J. (1986b). In vitro and in vivo studies on the availability of iron from storage proteins to stimulate membrane peroxidation. In Free Radicals, Cell Damage and Disease (Ed C. Rice-Evans), pp. 29-27. Richelieu Press, London.
- O'Flynn, K., Zanders, E.D., Lamb, J.R., Beverley, P., Wallace, D.L., Tatham, P.E.R., Tax, W.J.M., and Linch, D.C. (1985). Investigation of early T-cell activation: analysis of the effect of specific antigen, IL-2 & monoclonal antibodies on intracellular free calcium concentration. *Eur. J. Immunol.*, 15:7.

- Octave, J., Schneider, Y., Trouet, A., and Crichton, R.R. (1981). Iron uptake and utilization by mammalian cells. I. Cellular uptake of transferrin and iron. *Trend Biochem. Sci.*, **8**:217.
- Oettegen, H.C., Terhorst, C., Cantley, L.C., and Rosoff, P.M. (1985). Stimulation of the T3-T cell receptor complex induces a membrane-potential sensitive calcium influx. *Cell*, **40**:583.
- Okazaki, I., Brinckehoff, C.E., Sinclair, J.F., Sinclair, P.R., Bonkowsky, H.L., and Harris, Jr, E.D. (1981). Iron increases collagenase production by rabbit synovial fibroblasts. J. Lab. Clin. Med., 97:396.
- Omary, M.B. and Trowbridge, I.S. (1981). Biosynthesis of the human transferrin receptor in cultured cells. J. Biol. Chem., 256:12888.
- Omary, M.B., Trowbridge, I.S., and Minowada, J. (1980). Human cell surface glycoprotein with unusual properties. *Nature*, **286**:9.
- Oram, J.D. and Reiter, B. (1968). Inhibition of bacteria by lactoferrin and other iron-chelating agents. *Biochim. Biophys. Acta*, **170**:351.
- Opelz, G. and Persijn, G.G. (1981). Blood transfusion in renal transplantation. *Transplant. Proc.*, 13:1658.
- Oria, R., Alvarez-Hernandez, X., Licéaga, J., and Brock, J.H. (1988). Uptake and handling of iron from transferrin, lactoferrin, and immune complexes by a macrophage cell line. *Biochem. J.*, 251:221.
- Oseas, R., Yang, H-H., Baehner, R.L., and Boxer, L.A. (1981). Lactoferrin: a promoter of polymorphonuclear leukocyte adhesiveness. *Blood*, 57:939.
- Owen, D. and Kühn, L.C. (1987). Noncoding 3' sequences of the transferrin receptor gene are required for mRNA regulation by iron. *EMBO J.*, 6:1287.
- Pardalos, G., Kannakoudi-Tsakalidis, F., Malaka-Zafirin, M., Tsantali, H., and Papaevangelou, G. (1987). Iron-related disturbances of cell mediated immunity in multitransfused children with

thalassemia major. Clin. Exp. Immunol., 68:138.

- Park, I., Schaeffer, E., Sidoli, A., Baralle, F.E., Cohen, G.N., and Zakin, M.M. (1985). Organization of the human transferrin gene: direct evidence that it originated by gene duplication. *Proc. Natl. Acad. Sci. USA*, 82:3149.
- Parker, W.C., Hagstrom, J.W.C., and Bearn, A.G. (1963). Additional studies on the transferrin of cord serum and cerebrospinal fluid. Variation in carbohydrate prosthetic groups. J. Exp. Med., 118:975.
- Pattanapanyasat, K., Hoy, T.G., and Jacobs, A. (1988). Effect of phytohaemagglutinin on the synthesis and secretion of ferritin in peripheral blood lymphocytes. Br. J. Haematol., **69**:565.
- Pattanapanyasat, K. (1989a). Expression of isoferritins in peripheral blood lymphocytes: Effect of phytohaemagglutinin and iron. Eur. J. Haematol., 43:143.
- Pattanapanyasat, K. (1989b). Ferritin synthesis in peripheral blood lymphocytes following in vitro stimulation. Eur. J. Haematol., 42:454.
- Pattanapanyasat, K., Hoy, T.G., and Jacobs, A. (1987). The response of intracellular and surface ferritin after T-cell stimulation *in vitro*. *Clin. Sci.*, **73**:605.
- Pecoraro, V.L., Harris, W.R., Carrano, C.J., and Raymond, K.N. (1981). Siderophilin metal coordination. Different ultraviolet spectroscopy of di, tri, and tetravalent metal ions with ethylenebis (O-hydroxyphenyl) glycine. *Biochemistry*, 20:7033.
- Pelosi, E., Testa, U., Louache, F., Thomopoulos, P., Samoggia, P., and Peschle, C. (1986). Expression of transferrin receptors in phytohaemagglutinin-stimulated human T-lymphocytes. J. Biol. Chem., 261:3036.
- Pelosi-Testa, E., Samoggia, P., Giannella, G., Montesoro, E., Caravita, T., Salvo, G., Camagna, A., Isacchi, G., Testa, U., and Peschele, C. (1988). Mechanisms underlying activation: mitogen

initiates and IL-2 amplifies the expression of transferrin receptors via intracellular iron level. *Immunology*, **64**:273.

- Perez-Infanté, V. and Mather, J.P. (1982). The role of transferrin in the growth of testicular cell lines in serum-free medium. *Exp. Cell. Res.*, 142:325.
- Perez-Infante, V., Bardin, C.W., Gunsalus, G.L., Musto., N.A., Rich, K.A., and Mather, J.P. (1986). Differential regulation of testicular transferrin and endrogen-binding protein secretion in primary cultures of rat Sertoli cells. *Endocrinology*, **118**:383.
- Phillips, J.L. and Azari, P. (1975). Effect of iron transferrin on nucleic acid synthesis in phytohaemagglutinin-stimulated human lymphocytes. *Cell. Immunol.*, 15:94.
- Phillips, J.L., Boldt, D.H., and Harper, J. (1987). Iron-transferrininduced increase in protein kinase C activity in CCRF-CEM cells. J. Cell. Physiol., 132:3349.
- Phillips, J.L.and Rutledge, L.M. (1984). Effects of iron transferrin and zinc transferrin on ferritin synthesis by human lymphocytes. *Biol. Trace Element Res.*, 6:337.
- Phillips, M.E. and Thorbecke, G.J. (1966). Studies on the serum proteins of chimeras. I. Identification and study of the site of origin of donor type serum proteins in adult rat-into-mouse chimeras. *Int. Arch. Allergy*, 29:553.
- Pietrangelo, A., Cossariza, A., Monti, D., Ventura, E., and Franceschi, . (1988). DNA repair in lymphocytes from humans and rats with chronic iron overload. *Biochem. Biophys. Res. Commun.*, 154:698.
- Platz, P., Fog, T., Morling, N., Suejgaard, A., Ryder, L.P., Thomsen, M., and Jerild, C. (1976). Immunological *in vitro* parameters in patients with multiple sclerosis and normal individuals. *Acta Pathol. Microbiol. Scand.*, 84:501.
- Pollack, S. and Campana, T. (1981). Early events in the guinea pig reticulocyte iron uptake. *Biochim. Biophys. Acta*, 673:366.

- Ponka, P., Borova, J., Neuwirt, J., Fuchs, O., and Necas, E. (1979). A study of intracellular iron metabolism using pyridoxal isonicotinoyl hydrazone and other synthetic chelating agents. *Biochim. Biophys. Acta*, 586:278.
- Ponka, P., Schulman, H.M., and Wilczynska, A. (1982). Ferric pyridoxal isonicotinoyl hydrazone can provide iron for heme synthesis in reticulocytes. *Biochim. Biophys. Acta*, **718**:151.
- Porter, J.B., Huehns, E.R., and Hider, R.C. (1989). The development of iron chelating drugs. *Bailliere's Clin. Haematol.*, 2:257.
- Powell, L.W., Mortimer, R., and Harris, O.D. (1971). Cirrhosis of the liver. A comparative study of the four major aetiological groups. *Med. J. Austral.*, 1:941.
- Prince, H.E. and John, J.K. (1986). Cyclosporine inhibits the expression of receptors for interleukin 2 and transferrin on mitogenactivated human T lymphocytes. *Immunol. Invest.*, **15**:463.
- Prunier, J.H., Bearn, A.G., and Cleve, H. (1964). Site of formation of the group-specific component and certain other serum protein. *Proc. Soc. Exp. Biol. Med.*, 115:1005.
- Puschmann, M. and Ganzoni, A.M. (1977). Infected resistance of irondeficient mice to Salmonella infection. *Infect. Immunity*, **17**:663.
- Rabin, M., McClelland, A., Kühn, L., and Ruddle, F.H. (1985). Regional localization of human transferrin receptor gene to 3q26.2→ter. Am. J. Hum. Genet., 37:1112.
- Rao, K., Harford, J.B., Rouault, T., McClelland, A., Ruddle,
 F.H., and Klausner, R.D. (1986). Transcriptional regulation by iron of the gene for the transferrin receptor. *Mol. Cell. Biol.*, 6:236.
- Rao, K.K., Shapiro, D., Matia, E., Bridges, K., and Klausner,
 R.D. (1985). Effect of alterations in cellular iron on biosynthesis of the transferrin receptor in K562 cells. *Mol. Cell. Biol.*, 5:595.

- Rao, K.M.K. (1982). Capping and mitogenesis: a model implicating microfilaments in lymphocyte activation. J. Theor. Biol., 98:61.
- Regoeczi, E., Chindemi, P.A., Debanne, M.T., and Prieels, J-P. (1985). Lactoferrin catabolism in rat liver. Am. J. Physiol., 248:G8.
- Reichard, P. and Ehrenberg, A. (1983). Ribonucleotide reductase a radical enzyme. *Science*, **221**:514.
- Richardson, D., Vitolo, L.W., Baker, E., and Webb, J. (1989). Pyridoxal isonicotinoyl hydrazone and analogues. Study of their stability in acidic, neutral and basic aqueous solution by ultravioletvisible spectrophotometry. *Biol. Metals*, 2:69.
- Richie, E., Gillespie, D., and Kohl, S. (1981). Human colostrum and milk differ in capacity to inhibit lymphocyte activity. *Fed. Proc.*, 40:1810.
- Richie, E.R., Hilliard, J.K., Gilmore, R., and Gillespie, D. (1987).
 Human milk-derived lactoferrin inhibits mitogen and alloantigen induced human lymphocyte proliferation. J. Reproductive Immunol., 12:137.
- Rittling, S.R., and Woodworth, R.C. (1984). The synthesis and turnover of ferritin in rat L-6 cells. J. Biol. Chem., 259:5561.
- Robb, R.J., Greene, W.C., and Rusk, C.M. (1984). Low and high affinity cellular receptors for interleukin-2. Implications for the level of Tac antigen. *J. Exp. Med.*, **160**:1126.
- Roberts, S. and Bomford, A. (1988). Ferritin iron kinetics and protein turnover in K562 cells. J. Biol. Chem., 263:19181.
- Rogers, H.J. and Synge, C. (1978). Bacteriostatic effect of human milk on *Escherichia coli*: the role of IgA. *Immunology*, **34**:19.
- Rogers, J. and Munro, H. (1987). Translation of ferritin light and heavy subunit mRNAs is regulated by intracellular chelatable iron levels in rat hepatoma cells. *Proc Natl. Acad. Sci. USA*, **84**:2277.
- Rogers, T.B., Feeney, R.E., and Meares, C.F. (1977). Interaction of

anions with iron-transferrin-chelate complexes. J. Biol. Chem., 252:8108.

- Romslo, I. (1980). Intracellular transport of iron. In Iron in Biochemistry and Medicine (Eds A. Jacobs and M. Worwood), pp. 325-362. Academic Press, London.
- Rose, I.M., Plowman, G.D., Teplow, D.B., Dreyer, W.J., Hellstrom, K.E., and Brown, J.P. (1986). Primary structure of the human melanoma-associated antigen p97 (melanotransferrin) deduced from the mRNA sequence. *Proc. Natl. Acad. Sci. USA*, 83:1261.
- Rothenberger, S., Iacopetta, B.J., and Kühn, C.L. (1987). Endocytosis of the transferrin receptor requires the cytoplasmic domain but not its phosphorylation site. *Cell*, **49**:423.
- Rouault, T.A., Hentze, M.W., Caughman, S.W., Harford, J.B., and Klausner, R.D. (1988). Binding of a cytosolic protein to the iron-responsive element of human ferritin messenger RNA. Science, 241:1207.
- Rudland P. S., H. Durbin, D. Clingan, and L. Jimenez de Asua (1977). Iron salts and transferrin are specifically required for division of cultured 3T6 cells. *Biochem. Biophys. Res. Comm.*, 75:556.
- Rudolph, N.S., Ohlsson-Wichelm, B.M., Leary, J.F., and Rowley, P.T. (1985). Single cell analysis of the relationship among transferrin receptors, proliferation and cell cycle phase in K562 cells. *Cytometry.*, 6:151.
- Sacks, T., Moldow, C.F., Craddock, P.R., Bowers, T.K., and Jacob, H.S. (1978). Oxygen radicals mediate endothelial cell damage by complement-stimulated granulocytes. An *in vitro* model of immune vascular damage. J. Clin. Invest., 61:1161.
- Sagone, A.L., Jr., Husney, R., Guter, H., and Clark, L. (1984). Effect of catalase on the proliferation of human lymphocyte to phorbol myristate acetate. *J. Immunol.*, **133**:1488.
- Saito, K., Nishisato, T., Grasso, J.A., and Aisen, P. (1986).

Interaction of transferrin with iron-loaded rat peritoneal macrophages. Br. J. Haematol., 62:275.

- Sanchez, L., Aranda, P., Perez, M.D., and Calvo, M. (1988). Concentration of lactoferrin and transferrin throughout lactation in cow's colostrum and milk. *Biol. Chem. Hoppe-Seyler*, **369**:1005.
- Sayre, P.H., Chang, H.C., Hussey, R.E., Brown, N.R., Richardson, N.E., Spagnoli, G., Clayton, L.K., and Reinherz, E.L. (1987). Molecular cloning and expression of T11 cDNAs reveal a receptor-like structure on human T lymphocytes. *Proc. Natl. Acad. Sci. USA*, 84:2941.
- Schade, A.L.and Caroline, L. (1946). An iron binding component in human blood plasma. *Science*, **104**:340.
- Schlabach, M.R. and Bates, G.W. (1975). The synergistic binding of anions and Fe³⁺ by transferrin. *J. Biol. Chem.*, **250**:2182.
- Schneider, W. and Erni, I. (1981). Chemistry of transit iron pools. In The Biochemistry and Physiology of Iron (Eds. P. Saltman and J.Hegenauer), pp. 121-126. Elsevier North Holland, New York.
- Schreiber, G., Dryburgh, H., Millership, A., Matsuda, Y., Inglis, A., Phillips, J., Edwards, K., and Maggs, J. (1979). The synthesis and secretion of rat transferrin. J. Biol. Chem., 254:12013.
- Schwab, R., Crow, M.K., Russo, C., and Webster, M.E. (1985). Requirements for T-cell activation by OKT3 monoclonal antibody: role of modulation of T3 molecules and interleukin 1. J. Immunol., 135:1714.
- Sciot, R., Paterson, A.C., van der Oord, J.J., and Desmet, V.J. (1987). Lack of hepatic transferrin receptor expression in hemochromatosis. *Hepatology*, 7:831.
- Selden, C., Owen, M., Hopkins, J.M.P., and Peters, T.J. (1980). Studies on the concentration and intracellular localization of iron proteins in liver biopsy specimens from patients with iron overload with special reference to their role in lysosomal disruption. *Br. J*

Haematol., 44:593.

- Seligman, P.A. (1983). Structure and function of the transferrin receptor, *Prog. Hematol.*, 13:131.
- Seligman, P.A., Schleicher, R.B., and Allen, R.H. (1979). Isolation and characterization of the transferrin receptor form human placenta. J. Biol. Chem., 254:9943.
- Sibille, J.C., Doi, K., and Aisen, P. (1987). Hydroxyl radical formation and iron binding proteins. Stimulation by the purple acid phosphatase. J. Biol. Chem., 262:59.
- Sitkovsky, M.V., Pasternack, M.S., Lugo, J.P., Klein, J.P., and Eisen, H.N. (1984). Isolation and partial characterization of concanavalin A receptors on cloned cytotoxic T lymphocytes. Proc. Natl. Acad. Sci. USA, 81:1519.
- Skinner, M.K. and Griswold, M.D. (1980). Sertoli cells synthesize and secrete transferrin-like protein. J. Biol. Chem., 255:9523.
- Skinner, M.K. and Griswold, M.D. (1982). Secretion of testicular transferrin by cultured Sertoli cells is regulated by hormones and retinoids. *Biol. Reprod.*, 27:211.
- Skinner, M.K., Cosand, W.L., and Griswold, M.D. (1984). Purification and characterization of testicular transferrin secreted by rat Sertoli cells. *Biochem. J.*, 218:313.
- Slater, K. and Flecher, J. (1987). Lactoferrin derived from neutrophils inhibits the mixed lymphocyte reaction. *Blood*, **69**:1328.
- Smith, K.A., Lackman, L.B., Oppenheim, J.J., and Favata, M.F. (1980). The functional relationship of the interleukins. J. Exp. Med., 151:1551.
- Solomon, L.R. (1982). Consideration in the use of B_6 vitamers in hematologic disorders: I. Red cell transport and metabolism of pyridoxal. *Blood*, **59**:495.
- Soltys, H.D. and Brody, J.I. (1970). Synthesis of transferrin by human peripheral blood lymphocytes. J. Lab. Clin. Med., 75:250.

- Soyano, A., Fernandez, E., and Romano, E. (1985). Suppressive effect of iron on *in vitro* lymphocyte function. Formation of iron polymers as a possible explanation. *Int. Arch. Allergy Appl. Immunol.*, **76**:376
- Spik, G. (1982). Structure, conformation, and functions of the glycans of serum transferrin. In *The Biochemistry and Physiology of Iron* (Eds. P. Saltman and J. Hegenauer), pp. 49-56. Elsevier Biomedical Press, Amsterdam.
- Spik, G. and Montreuil, J. (1966). Études comparatives de la structure de la transferrine et de la lactotransferrine humaines. "Finger printing" des hydrolysats protéasique des deux glycoproteides. Comp. Rend. Soc. Biol.., 160:94.
- Spik, G., Brunet, B., Mazurier-Dohaine, C., Fontaine, G., and Montreuil, J. (1982). Characterization and properties of the human and bovine lactotransferrins extracted from the faeces of newborn infants. Acta Paediatr. Scand., 71:979.
- Spik, G., Debruyne, V., Montreuil, J., van Halbeek, H., Vliegenthart, J.F.G. (1985). Primary structure of two sialylated triantennary glycans from human serotransferrin. FEBS Lett., 183:65.
- Spik, G., Strecker, G., Fournet, B., Bouquelet, S., Montreuil, J., Dorland, L., Van Halbeek, H., and Vliegenthart, J.F.G. (1982).
 Primary structure of the glycans from human lactotransferrin. *Eur. J. Biochem.*, 121:413.
- Spiro, T.G., Bates, G., and Saltman, P. (1967). The hydrolytic polymerization of ferric citrate. II. The influence of excess citrate. J. Am. Soc., 89:5559.
- Sporn, M.D. and Todoro, G.J. (1980). Autocrine secretion and malignant transformation of cells. N. Eng. J. Med., 303:878.
- St Pierre, T.G., Pollard, R.K., Dickson, D.P.E., Ward, R.J., and Peters, T.J. (1988). Mössbauer spectroscopic studies of

deproteinised, subfractionated and reconstituted ferritins: the relationship between haemosiderin and ferritin. *Biochim. Biophys.* Acta, 952:158.

- Stecher, V.J. and Thorbecke, G.J. (1967a) Sites of synthesis of serum proteins. I. Serum proteins produced by macrophages in vitro. J. Immunol., 99:643.
- Stecher, V.J. and Thorbecke, G.J. (1967b) Sites of synthesis of serum proteins. III. Production of $\beta_1 C$, $\beta_1 E$ and transferrin by primate and rodent cell lines. *J. Immunol.* **99**:660.
- Stern, J.B. and Smith, K.A. (1986). Interleukin-2 induction of T-cell G₁ progression and *c-myb* expression. *Science*, **233**:203.
- Streb, H., Irvine, R.F., Berridge, M.J., and Schulz, I. (1983).
 Release of Ca²⁺ from a nonmitochondrial intracellular store in pancreatic acinar by inositol-1,4,5-triphosphate. *Nature*, 306:67.
- Sturrock, A., Alexander, J., Lamb, J., Craven, C.M., and Kaplan, J. (1990). Characterization of a Tf-independent uptake system for iron in HeLa cell. J. Biol. Chem., 265:3139.
- Suard, Y.L.M., Haeuptle, M-T., Farinon, E., Kraehenbuhl, J-P. (1983). Cell proliferation and milk protein gene expression in rabbit mammary cell cultures. J. Cell Biol., 96:1435.
- Summers, M., White, G., and Jacobs, A. (1975). Ferritin synthesis in lymphocytes, polymorphs, and monocytes, *Br. J. Haematol.*, 30:425.
- Summers, M.R. and Jacobs, A. (1976). Iron uptake and ferritin synthesis by peripheral blood leucocytes from normal subjects and patients with iron deficiency and the anaemia of chronic disease. *Br*. *J. Haematol.*, 34:221.
- Sun, I.L., Navas, P., Crane, F.L., Morre, D.J., and Low, H. (1987). NADH diferric transferrin reductase in liver plasma membrane. J. Biol. Chem., 262:15915.

Sutherland, R., Delia, D., Schneider, C., Newman, R., Kemshead,

J., and Greaves, M. (1981). Ubiquitous cell-surface glycoproteins on tumor cells is proliferation-associated receptor for transferrin. *Proc. Natl. Acad. Sci. USA*, **78**:4515.

- Sykes, A.J.C., Thomas, M.J., Goldie, D.J., and Turner, G.M. (1982). Plasma lactoferrin levels in pregnancy and cystic fibrosis. *Clin. Chim. Acta*, **122**:385.
- Taetle, R. and Honeysett, M. (1988). γ-interferon modulates human monocyte/macrophage transferrin receptor expression. *Blood*, 71:1550.
- Taetle, R., Castagnola, J., and Mendelsohn, J. (1986). Mechanisms of growth inhibition by anti-transferrin receptor monoclonal antibodies. *Cancer Res.*, **46**:1759.
- Taetle, R., Rhyner, K., Castagnola, J., To, D., and Mendelsohn, J. (1985). Role of transferrin, Fe, and transferrin receptors in myeloid leukemia cell growth. Studies with an antitransferrin receptor monoclonal antibody., J. Clin. Invest., 75:1061.
- Tanno Y., Arai, S., and Takishima, T. (1982). Iron containing proteins augment responses of human lymphocytes to phytohaemagglutinin and pokeweed mitogen in serum-free medium. *Tohoku J. Exp. Med.*, 137. 335.
- Tanno, Y. and Takishima, T. (1982). Enhancing effect of saccharated oxide on human lymphocyte transformation in serum-free medium. *Tohoku J. Exp. Med.*, 136:463.
- Taylor, P.G., Soyano, A., Romano, E., and Layrisse, M. (1987). Physiological and non-physiological forms of iron affect differently proliferation and ferritin synthesis in human mononuclear cells *in vitro*. Tohoku J. Exp. Med., 153:285.
- Taylor, P.G., Soyano, A., Romano, E., and Layrisse, M. (1988). Iron and transferrin uptake by phytohemagglutinin-stimulated human peripheral blood lymphocytes. *Microbiol. Immunol.*, 32:949.
- Teng, C.T., Pentecost, B.T., Marshall, A., Solomon, A., Bowman, B.H., Lalley, P.A., and Naylor, S.L. (1987). Assignment of the

lactotransferrin gene to human chromosome 3 and to mouse chromosome 9. Somat. Cell. Mol. Genet., 13:689.

- Tengerdy, C., Azari, P., and Tengerdy, R.P. (1966). Immunochemical reaction of conalbumin and its metal complexes. *Nature*, **211**:203.
- Theil, E.C. (1987). Ferritin: structure, gene regulation, and cellular function in animals, plants, and microorganisms. *Ann. Rev. Biochem.*, 56:289.
- Thibodeau, S.N., Lee, D.C., and Palmiter, R.D. (1978). Identical precursors for serum transferrin and egg white conalbumin. J. Biol. Chem., 253:3771.
- Thomas, C.E. and Aust, S.D. (1986). Reductive release of iron from ferritin by cation free radicals of paraquat and other bipyridyls. J. Biol. Chem., 261:13064.
- Thorbecke, G.J., Liem, H.H., Knight, S., Cox, K., and Muller-Eberhard, U. (1973). Sites of formation of the serum proteins transferrin and hemopexin. J. Clin. Invest., 52:725.
- Titeux, M., Testa, U., Louache, F., Thomopoulos, P., Rochant, H., and Breton-Gorius, J. (1984). The role of iron in the growth of human leukemic cell lines. J. Cell. Physiol., 121:251.
- Todaro, G.J., de Larco, J.E., and Cohen, S. (1976). Transformation of murine and feline sarcoma viruses specifically blocks binding of epidermal growth factor. *Nature*:264:26.
- Tormey, D.C., Imrie, R.C., and Mueller, G.C. (1972). Identification of transferrin as a lymphocyte growth promoter in human serum. *Exp. Cell. Res.*, **74**:163.
- Torti, S.V., Kwak, E.L., Miller, L.L., Ringold, G.M., Myambo, K.B., Young, A.P., and Torti, F.M. (1988). The molecular cloning and characterization of murine ferritin heavy chain, a tumor necrosis factor-inducible gene. J. Biol. Chem., 263:12636.

Trowbridge I. S., Lesley, J., and Schulte, R. (1982). Murine cell

surface transferrin receptor: studies with an anti-receptor monoclonal antibody. J. Cellular Physiol., **112**:403-410.

- Trowbridge, I.S. and Lopez, F. (1982). Monoclonal antibody to transferrin receptor blocks transferrin binding and inhibits human tumour cell growth *in vitro*. *Proc. Natl. Acad. Sci. USA*, **79**:1175.
- Trump, B.H., Valigorsky, J.M., Arstila, A.U., Mergner, W.Y., and Kinney, T.D. (1973). The relationship of intracellular pathways of iron metabolism to cellular overload and the iron storage diseases. *Am. J. Pathol.*, **72**:295.
- Tsein, R.Y., Pozzan, Y., and Rink, T.J. (1982). T-cell mitogens cause early changes in cytoplasmic free Ca⁺⁺ and membrane potential in lymphocytes. *Nature*, **295**:68.
- Tsoukas, C.D., Landgraf, B., Bentin, J., Valentine, M., Lotz, M., Vaughan, J.H., and Carson, D.A. (1985). Activation of resting T lymphocytes by anti-CD3 (T3) antibodies in the absence of monocytes. J. Immunol., 135:1719.
- Tuil, D., Vaulont, S., Levin, M.J., Munnich, A., Moguilewsky, M., Bouton, M.M., Brissot, P., Dreyfus, J.-C., and Kahn, A. (1985). Transient transcriptional inhibition of the transferrin gene by cyclic AMP. *FEBS Lett.*, 189:310.
- Uzan, G., Frain, M., Park, I., Besmond, C., Maessen, G., Sala-Trépat, J., Zakin, M.M., and Kahn, A. (1984). Molecular cloning and sequence analysis of cDNA for human transferrin. *Biochem. Biophys. Res. Commun.*, 119:273.
- van Eijk, H.G. and van Noort, W.L. (1976). Isolation of rat transferrin using CNBr-activated Sepharose-4B. J. Clin. Chem. Clin. Biochem., 14:475.
- van Renswoude, J., Bridges, K.R., Harford, J.B., and Klausner, R.D. (1982). Receptor-mediated endocytosis of transferrin and the uptake of Fe in K562 cells: identification of a non-lysosomal acidic compartment. *Proc. Natl. Acad. Sci. USA*, **79**:6186.
- van Snick, J.L., and Masson, P.L. (1976). The binding of human

lactoferrin to mouse peritoneal cells. J. Exp. Med., 144:1568.

- van Snick, J.L., Markowetz, B., and Masson, P.L. (1977). The ingestion and digestion of human lactoferrin by mouse peritoneal macrophages and the transfer of its iron to ferritin. *J.Exp. Med.*, **146**:817.
- van Snick, J.L., Masson, P.L., and Heremans, J.F. (1974). The involvement of lactoferrin in hyposideremia of acute inflammation. J. *Exp. Med.*, **141**:1068.
- van Wauwe, J.P., de Mey, J.R., and Goossens, J.G. (1980). OKT3: a monoclonal anti-human T lymphocyte antibody with potent mitogenic properties. *J. Immunol.*, **124**:2707.
- van Wauwe, J.P., Goossens, J.G., and Beverley, P.C.L. (1984). Human T lymphocyte activation by monoclonal antibodies: OKT3, but not UCHT1, triggers mitogenesis via an interleukin-2-dependent mechanism. J. Immunol., 133:129.
- Verhoef, N.J., Kottenhagen, M.J., Mulder, H.J.M., Noordeloos, P.J., and Leijnse, B. (1978). Functional heterogeneity of transferrin-bound iron. Iron uptake by cell suspensions from bone marrow and liver and by cell cultures of fibroblasts and lymphoblasts. Acta Haematol., 60:210.
- Vezzoni, P., Levi, S., Gabri, E., Pozzi, M.R., Spinazze, S., and Arosio, P. (1986). Ferritin in malignant and non-malignant lymphoid cells. Br. J. Haematol., 62:105.
- Vincenti, F., Duca, R.M., Amend, W., Perkins, H.A., Cochrum, K.C., Feduska, N.J., and Salvatierra, Jr, O. (1978). Immunologic factors determining survival of cadaver-kidney transplants. The effect of *HLA* serotyping, cytotoxic antibodies and blood transfusions on graft survival. N. Eng. J. Med., 299:793.
- von Ehrenstein, G. (1956). Iron transport through the lymph steam. *Acta Chem. Scand.*, **10**:703.
- Vostrejs, M., Moran, P.L., and Seligman, P.A. (1988). Transferrin synthesis by small cell lung cancer cells acts as an autocrine regulator

of cellular proliferation. J. Clin. Invest., 82:331.

- Ward, J.H., Kushner, J.P., and Kaplan, J. (1982). Regulation of HeLa cell transferrin receptors. J. Biol. Chem., 257:10317.
- Ward, J.H., Kushner, J.P., Ray, F.A., and Kaplan, J. (1984). Transferrin receptor function in hereditary hemochromatosis. J. Lab. Clin. Med., 103:246.
- Watanabe, T., Nagura, H., Watanabe, K., and Brown, W.R. (1984). The binding of human milk lactoferrin to immunoglobulin A. FEBS Lett., 168:203.
- Watson, M.A., Briggs, J.D., Dimandopoulos, A.A., Hamilton, D.N.H., and Dick, H.M. (1979). Endogenous cell-mediated immunity, blood transfusion, and outcome of renal transplantation. *Lancet*, 2²:1323.
- Watt, G.D., Frankel, R.B., and Papaefthymiou. (1985). Reduction of mammalian ferritin. *Proc Natl. Acad. Sci. USA*, 82:3640.
- Watts, C.A. (1985). Rapid endocytosis of the transferrin receptor in the absence of bound transferrin. J. Cell Biol., 100:633.
- Wedner, H.J. and Parker, C.W. (1976). Lymphocyte activation. Prog. Allergy, 20:195.
- Weinberg, E.D. (1978). Iron and infection. Microbiol. Rev., 42:45.
- Weinberg, E.D. (1984). Iron withholding: a defence against infection and neoplasia. *Physiol. Rev.*, 64:65.
- Weinberg, E.D. (1985). Iron, infection, and neoplasia. Clin Physiol. Biochem., 4:50.
- Weir, M.P., Gibson, J.F., and Peters, T.J. (1984). Haemosiderin and tissue damage. *Cell Biochem. Function*, 2:186.
- Weir, M.P., Sharp, G.A., and Peters, T.J. (1985). Electron microscopic studies of human haemosiderin and ferritin. J. Clin. Pathol., 38:915.

- Weiss, A., Imboden, J., Hardy, K., Manger, B., Terhorst, C., and Stobo, J. (1986). The role of the T3/antigen receptor complex in T-cell activation. Ann. Rev. Immunol., 4:543.
- Weiss, A., Wiskocil, R., and Stobo, J. (1984). The role of T3 surface molecules in the activation of human T cells: A two-stimulus requirement for IL-2 production reflects events occurring at a pre-translational level. J. Immunol., 133:1.
- Weiss, A. and Stobo, J. (1984). Requirement for the coexpression of T3 and the T cell antigen receptor on a malignant human T cell line. J. Exp. Med., 160:1284.
- White, G.P. and Jacobs, A. (1978). Iron uptake by Chang cells from transferrin, nitriloacetate and citrate complexes. The effects of ironloading and chelation with desferrioxamine. *Biochim. Biophys. Acta*, 543:217.
- White, G.P., Bailey-Wood, R., and Jacobs, A. (1976). The effect of chelating agents on cellular iron metabolism. *Clin. Sci. Mol. Med.*, 50:145.
- Williams, J. (1975). Iron-binding fragments from the carboxyl-terminal region of hen ovotransferrin. *Biochem. J.*, 149:237.
- Williams, J. (1982). The evolution of transferrin. Trends Biochem. Sci., 7:215.
- Williams, J., and Moreton, K. (1980). The distribution of iron between the metal-binding sites of transferrin in human serum. *Biochem. J.*, 185:483.
- Williams, J., Elleman, T.C., Kingston, I.B., Wilkins, A.G., and Kuhn, K.A. (1982). The primary structure of hen ovotransferrin. *Eur. J. Biochem.*, 122:297.
- Willingham, M.C. and Pastan, I. (1985). Ultrastructural immunocytochemical isolation of the transferrin receptor using a monoclonal antibody in human KB cells. J. Histochem. Cytochem., 33:59.

- Willingham, M.C., Hanover, J.A., Dickson, R.B., and Pastan, I. (1984). Morphological characterization of the pathway of transferrin endocytosis and recycling in human KB cells. *Proc. Natl. Acad. Sci.* USA, 81:175.
- Wochner, R.D., Weissman, S.M., Waldmann, T.A., and Berlin, N.I. (1968). Direct measurement of the rates of synthesis of plasma proteins in control subjects and patients with gastrointestinal protein loss. J. Clin. Invest., 47:3.
- Wong, C.T. and Morgan, E.H. (1973). Placental transfer of iron in the guinea pig. *Quart. J. Exp. Physiol.*, 58:47.
- Wong, S.F., Halliwell, B., Richmond, R., and Skowroneck, W.R. (1981). The role of superoxide and hydroxyl radicals in he generation of hyaluronic acid induced by metal ions and by ascorbic acid. J. Inorg. Biochem., 14:127.
- Woodruff, M.F.A. and van Rood, J.J. (1983). Possible implication of the effect of blood transfusion on allograft survival. *Lancet*, 1:1201.
- Worwood, M. (1986). Serum ferritin. Clin. Sci., 70:215.
- Worwood, M., Hourahane, D., and Jones, M.B. (1984). Accumulation and release of isoferritins during incubation *in vitro* of human peripheral blood mononuclear cells. *Br. J. Haematol.*, 56:31.
- Wrigglesworth, J.M. and Baum, H. (1980). The biochemical functions of iron. In *Iron in Biochemistry and Medicine* (Eds A. Jacobs and M. Worwood), Vol. II. pp. 29-86. Academic Press, London.
- Yamada, M., Okigaki, T., and Awai, M. (1987). Role of superoxide radicals in cytotoxic effects of Fe-NTA on cultured normal liver epithelial cells. *Cell Struc. Func.*, **12**:407.
- Yamanoi, Y., Matsura, R., and Awai, M. (1982). Mechanism of iron toxicity in the liver and pancreas after a single injection of ferric nitrilotriacetate. *Acta Haematol. Japonica*, **45**:47.
- Yang, F., Lum, J.B., McGill, JR., Moore, C.M., Naylor, S.L., van Bragt, P.H., Baldwin, W.D., and Bowman, B.H. (1984).

Human transferrin: cDNA characterization and chromosomal localization. *Proc. Natl. Acad. Sci. USA*, **81**:2752.

- Yang, F., Naylor, S.L., Lum, J.B., Cutshaw, S., McCombs, J.L., Naberhaus, K.H., McGill, J.R., Adaian, G.S., Moore, C.M., Barnett, D.R., and Bowman, B.H. (1986). Characterization, mapping, and expression of human ceruloplasmin gene. *Proc. Natl. Acad. Sci. USA*, 83:1525.
- Yang, S.Y., Chouaib, S., and Dupont, B. (1986). A common pathway for T lymphocyte activation involving both the CD3-Ti and CD2 sheep erythrocyte receptor determinants. J. Immunol., 137:1097.
- Yeoh, G.C.T. and Morgan, E.H. (1974). Albumin and transferrin synthesis during development in the rat. *Biochem. J.*, 144:215.
- Young, D.V., Cox III, F.W., Chipman, S., and Hartman, S.C. (1979). The growth stimulation of SV3T3 cells by transferrin and its dependence on biotin. *Exp. Cell. Res.*, **118**:410.
- Young, S. P., Roberts, S., and Bomford, A. (1985). Intracellular processing of transferrin and iron by isolated rat hepatocytes. *Biochem. J.*, 232:819.
- Young, S. and Bomford, A. (1984). Transferrin and cellular iron exchange. *Clin. Sci.*, **67**:273.
- Young, S.P. and Aisen, P. (1980). The interaction of transferrin with isolated hepatocytes. *Biochim. Biophys. Acta*, 633:145.
- Young, S.P., Bomford, A., and Williams, R. (1984). The effect of iron saturation of transferrin on its binding and uptake by rabbit reticulocytes. *Biochem. J.*, **219**:505.
- Zahringer, J., Baliga, B.S., and Munro, H.N. (1976). Novel mechanism for translational control in regulation of ferritin synthesis by iron. *Proc. Natl. Acad. Sci. USA*, **73**:857.
- Zak, O. and Aisen, P. (1986). Nonrandom distribution of iron in circulating human transferrin. *Blood*, **68**:157.

- Zapolski, E.J. and Princiotto, J.V. (1977). Transferrin iron binding: observation of nonrandom (heterogeneic) binding. In *Proteins of Iron Metabolism* (Eds. E.B. Brown, P. Aisen, J. Fielding, and R.R. Crichton), pp. 205-210. Grune and Stratton, New York.
- Zarrabi, M.H., Lysik, R., di Stefano, J., and Zucker, S. (1977). The anaemia of chronic disorders: studies of iron reutilization in the anaemia of experimental malignancy. *Br. J. Haematol.*, **35**:647.
- Zerial, M., Huylebroeck, D., and Garoff, H. (1987). Foreign transmembrane peptides replacing the internal signal sequence of transferrin receptor allow its translation and membrane binding. *Cell*, 48:147.
- Zucali, J.R., Broxmeyer, H.A., Levy, D., and Morse, C. (1989). Lactoferrin decreases monocyte-induced fibroblast production of myeloid colony-stimulating activity by suppressing monocyte release of interleukin-1. *Blood*, 74:1531.
- Zucali, J.R., Levy, D.A., Broxmeyer, H.E., and Morse, C.A. (1987). Lactoferrin acts on monocytes to inhibit production of interleukin 1 and the ability of monocyte-conditioned medium to induce fibroblasts to produce granulocyte-macrophage colony stimulating factor. *Blood*, **70**:191a (abstract).
- Zweier, J.L. and Aisen, P. (1977). Studies of transferrin with the use of Cu^{2+} as an electron paramagnetic resonance spectroscopic probe. *J. Biol. Chem.*, **252**:6090.
- Zwiller, J., Basset, P., Ulrich, G., and Mandel, P. (1982).
 Guanylate cyclase activators hemin and sodium nitroprusside stimulate cell growth in serum-free medium. *Exp. Cell Res.*, 141:445.

